ABSTRACT

WHAT DO YOU THINK ABOUT YOUR SCHOOLS:

A STUDY OF THE DETERMINATION OF
DISCRIMINATION AND RELIABILITY
INDICES AND THE ESTABLISHMENT
OF MICHIGAN NORMS

Ву

Gary F. Alkire

Purposes of the Study

The purposes of this study are three-fold: (1) to determine the ability of the What Do You Think About Your Schools to discriminate between the responses of elementary pupils, secondary students, parents, patrons, and faculty within and between cost quartiles; (2) to establish the instrument's reliability; and (3) to establish Michigan norms for the What Do You Think About Your Schools.

Two major hypotheses are formulated to test the discriminating ability of the instrument within the four financial support quartiles and between the five batteries by each financial support quartile. The two major hypotheses are:

1. The <u>What Do You Think About Your Schools</u> instrument will discriminate between the attitudes of elementary pupils, secondary students, parents, patrons, and faculty on the basis of financial support quartiles of Michigan

school districts (K-12). Financial support quartile is defined in terms of size, effort, ability, and expenditure.

2. The <u>What Do You Think About Your Schools</u> instrument will discriminate between the responses of elementary pupils, secondary students, parents, patrons, and faculty within fourth, third, second, and first financial support quartiles of Michigan school districts (K-12).

To improve the precision of the instrument two additional studies were done: (1) a study of item reliabilities, and (2) the establishment of norms for the total sample of school districts. These two studies established the stability of the instrument from test to test and provided a norm population against which users of the instrument could make comparisons.

Sample and Design

The sample was selected on the basis of a stratified random sample of Michigan school districts (K-12).

The sample included two districts in the fourth quartile,
two districts in the third quartile, four districts in the
second quartile, and six districts in the first quartile.

A one per cent sample of fourth quartile, a two per cent
sample of third quartile, a five per cent sample of
second quartile, and a ten per cent sample of first quartile were taken to determine the number of respondents for
the respective districts. Useable data were collected

from 3,057 respondents representing 882 elementary pupils, 869 secondary students, 739 parents, 480 patrons, and 87 faculty members. Ten per cent of each battery was identified with code numbers for re-testing within three weeks as a check on instrument reliability.

Instrumentation and Data Collection

Data for this study came from two sources: (1) the Michigan Education Association's Ranking of Michigan High School Districts by Selected Financial Data for 1966-67, and (2) the responses to the five batteries of the What Do You Think About Your Schools.

The financial data on size, effort, ability, and expenditure were organized by quartiles and a composite quartile was derived from the four financial characteristics. This procedure equated the districts within each quartile.

The measurement of attitudes was obtained through responses of elementary pupils, secondary students, parents, patrons, and faculty to the instrument, What Do You Think About Your Schools. This instrument is based on the assumption that attitudes of the school community can be measured with precision. Each battery varies in the number of items asked, but nineteen items are responded to by all groups. The total number of items for each battery is: (1) elementary pupils--37, (2) secondary students--

45, (3) parents--53, (4) patrons--31, and (5) faculty-60. Each battery consists of six categories, dealing
with some aspect of the school program. The six categories are: (1) satisfaction with schools, (2) school
program, (3) essential services desired, (4) school organization and size, (5) school plant, and (6) community
relations. The respondent indicates his choice to the
item by checking a four-point value scale. The scores
are dichotomized to indicate percentage of favorable and
unfavorable response for each item.

Each district in the sample was visited by the researcher. The instruments were distributed with the cooperation of administrators and teachers to sixth graders, twelfth graders, parents of sixth graders, patrons who were neighbors of sixth graders, and teachers. An analysis of the returns for each battery included:

(1) elementary pupils--96.7%, (2) secondary students--95.3%, (3) parents--81%, (4) patrons--52.6%, and faculty--100%.

Method of Treatment and Analysis

Testing the two hypotheses required the use of a statistical test for non-parametric data, since the responses were reported in percentage of favorable response. The following tests of significance were used:

1. The Chi Square (X^2) was used to test the significance of difference between the percentage of favorable

response by item for fourth, third, second, and first financial support quartile districts within each battery.

2. The Chi Square (X²) was used to test the significance of difference between the percentage of favorable response by item for elementary pupils, secondary students, parents, patrons, and faculty within fourth, third, second, and first financial support quartile districts.

Three other statistical procedures were undertaken to increase the precision of the instrument.

- 1. Item reliabilities were determined for the consistency of the first testing to the second testing for a ten per cent sample of the total sample using a Pearson product-moment correlation method.
- 2. Estimates of battery reliabilities were computed using a Spearman-Brown prophecy formula for estimating total battery reliability.
- 3. State norms were computed for the total sample of Michigan school districts by the application of the standard error of a percentage at the 95% confidence level.

Major Findings

The major findings of this study were:

1. The <u>What Do You Think About Your Schools</u> instrument does not discriminate between fourth, third, second, and first financial support quartile districts on the majority of items. The percentage of significant items

for each battery was: (1) elementary pupils--32%, (2) secondary students--27%, (3) parents--25%, (4) patrons--13%, and (5) faculty--10%.

- 2. The <u>What Do You Think About Your Schools</u> instrument did discriminate between elementary pupils, secondary students, parents, patrons, and faculty within each financial support quartile. An analyses of the four quartiles indicate the following percentage of significant items:

 (1) fourth quartile—61%, (2) third quartile—59%, (3) second quartile—72%, and (4) first quartile—77%.
- 3. The What Do You Think About Your Schools instrument produced item reliability coefficients ranging from a high of .88 for facultys' response to "Teacher Gives Help" to a low correlation of -.19 for facultys' response to "Parent-Teacher Relations." Of the total 229 items for the five batteries, only 27 were below .50.
- 4. The What Do You Think About Your Schools instrument produced the following estimates of battery reliabilities: (1) elementary pupils, .92; (2) secondary students, .95; (3) parents, .94; (4) patrons, .84; and (5) faculty, .94.
- 5. The <u>What Do You Think About Your Schools</u> instrument produced useable norms for the State of Michigan at a 95% confidence level. The standard error of a percentage was applied to each item providing a range of percentages to which users of the instrument could make comparisons.

WHAT DO YOU THINK ABOUT YOUR SCHOOLS:

A STUDY OF THE DETERMINATION OF
DISCRIMINATION AND RELIABILITY
INDICES AND THE ESTABLISHMENT
OF MICHIGAN NORMS

Ву

Gary F. Alkire

A THESIS

Submitted to

Michigan State University
in partial fulfillment of the requirements
for the degree of

DOCTOR OF EDUCATION

College of Education

G 53003 1/15/69

DEDICATION

To my loving wife and children Ingrid, David, and Heidi

ACKNOWLEDGEMENTS

The major influence and inspiration for this study was given by the chairman of the writer's dostoral committee, Dr. Herbert C. Rudman. Dr. Rudman has given unselfishly of his time to provide constant guidance to the writer and has instilled far higher aspirations in the writer than he thought possible.

Acknowledgement is also given to the members of the writer's doctoral guidance committee. Dr. Donald Leu, Dr. Carl H. Gross, Dr. James B. McKee, and Dr. William Durr. The opportunity of working with Dr. Rudman and Dr. Leu on many departmental and private projects has given the writer a deeper understanding of the role of the university professor.

A special acknowledgement is given to Dr. Carl Midjaas who was a constant companion and help for this study.

Acknowledgement is also given to the superintendents, principals, and thousands of elementary pupils, secondary students, parents, patrons, and faculty in the State of Michigan who participated in this study.

TABLE OF CONTENTS

														Page
DEDICAT:	ION	•	•	•	•	•	•	•	•	•	•	•	•	ii
ACKNOWL	EDGEMENT	TS.	•	•	•	•	•	•	•	•	•	•	•	111
LIST OF	TABLES	•	•	•	•	•	•	•	•	•	•	•	•	vi
LIST OF	APPENDI	CES	•	•	•	•	•	•	•	•	•	•	•	viii
CHAPTER I.	THE PRO	ים ום י	Л											٦
1.	Ine Inc	יוםנוכוי	1 •	•	•	•	•	•	•	•	•	•	•	1
	Purpo	ses	of	the	S1	tud	у.	•	•	•	•	•	•	1
	Signi	fica	ance	of	tl	ne '-	Pro	bler	n.	•	•	•	•	6
	Assum													6 8
	Delim Defin	iital	TOI	15 C	T I	une	200	uay	•	•	•	•	•	9
	Hypot	hase) II ()1 1	eri	115	•	•	•	•	•	•	•	12
	Resea										nt.	Wh	a t	12
	Do	You	Th.	ink	Abo	out.	You	יינו	Sch	ool	s.	****	<u> </u>	14
	Organ											•	•	15
II.	RELATED													17
		_	- .							_				
	Natio													٦ 0
	Theor	essn					• on 1				•	•	•	18
	111601	CCIC	:dl nant	- MOU -	ers	5 1	01, 1	1.05	gr.a	111				23
	Ass Relat	1166511 64	'Ost	- ∩11	• • 1 •	• 1 + 17	• 5 f i	1011	•	•	•	•	•	30
	Instr	umer	ntat	J-&u Ti∩n	.a	LUy		au I C		•	•	•	•	44
	Summa					•	•	•	•	•	•	•	•	62
		- 3	-	•	•	•	•	•	•	•	•	•	•	٠ <u>٠</u>
III.	INSTRUM	ENTA	TIC	ON A	ND	ME	THOI	OOLO)GY	OF				
	THE S	TUDY	. •		•		•	•		•	•	•	•	65
	Intro	duct	ior	1.	•	•	•	•	•	•	•	•	•	65
	Instr					•	•	•	•	•	•	•	•	65
	Finan					•	•	•	•	•	•	•	•	69
	Class					Di	str	icts	5 0	n				~ li
		t Fa					•	•	•	•	•	•	•	74
	Selec								•	٠.	•	•	•	77
	Distr					OT	тест	llor	1 0	1				7.0
		Ins				· ``	•	•	•	•	•	•	•	79 83
	Treat Stati							•	•	•	•	•	•	83
	JUAUL	\circ	<i>-</i> L L	1.1C C	1100		•	•	•	•	•	•	•	\circ

Chapte	r	Page
IV.	ANALYSIS OF THE HYPOTHESES	85
	Introduction	85 86 86 87
	Treatment	94 96 97 98
	Implications of the Statistical Treatment	103 104
V .	RELIABILITY COEFFICIENTS AND NORMS FOR THE WHAT DO YOU THINK ABOUT YOUR SCHOOLS BATTERIES	105
	Introduction	105 105 116 118
VI.	SUMMARY, CONCLUSIONS, IMPLICATIONS, AND RECOMMENDATIONS	123
	Summary	123 128 130 133 135
BIBLIC	GRAPHY	138
A DDENI	TOFS	146

LIST OF TABLES

Table]	Page
1.	A comparison of skills by battery for the Stanford Achievement Test	•	•	51
2.	Number of items within each battery by category	•	•	68
3.	Item numbers for each battery by category	•		70
4.	Classification of 507 Michigan school districts according to size, as of the school year, 1966-67	•		73
5.	Classification of 507 Michigan school districts according to state equalized valuation per pupil, as of the school yea 1966-67	r •		73
6.	Classification of 507 Michigan school districts according to expenditure per pupil, as of the school year 1966-67.	•	•	74
7.	Classification of 507 Michigan school districts according to mills levied for operation, as of the school year 1966-67	•	•	74
8.	School districts with the same quartile rank on all four cost factors			75
9.	Quartile range and number of school districts represented by the four cost-factors	•	•	76
10.	Quartile ranking and the number of school districts selected for this study			78
11.	Total school district population and number deemed necessary for this study.	•	•	80

Table		Page
12.	Total number of items significant at P > .05 by battery across quartile	88
13.	The items for each quartile across batteries significant at P > .05	99
14.	Number of respondents for the reliability study by battery and item	106
15.	The reliability coefficients of the instrument, What Do You Think About Your Schools by battery for each item	107
16.	Item reliabilities below .50 by battery	114
17.	Estimates of battery reliability	116
18.	Sample districts attitudes about schools expressed in a range of percentage of favorable response at the .95 probability level	119

LIST OF APPENDICES

Appendix		Page
Α.	What Do You Think About Your Schools	147
	Elementary Student Questionnaire	149
	Secondary Student Questionnaire	155
	Parent Questionnaire	162
	Patron Questionnaire	. 171
	Faculty Questionnaire	176
В.	Elementary Pupils Results by Quartile with Number of Respondents, Total Percentage of Favorable Response, "X", Degrees of Freedom, and Significance at P > .05	. 184
С.	Secondary Students Results by Quartile with Number of Respondents, Total Percentage of Favorable Response, "X", Degrees of Freedom, and Significance at P > .05	. 188
D.	Parents Results by Quartile with Number of Respondents, Total Percentage of Favorable Response, "X", Degrees of Freedom, and Significance at P > .05.	. 192
E.	Patrons Results by Quartile with Number of Respondents, Total Percentage of Favorable Response, "X2", Degrees of Freedom, and Significance at P > .05.	. 197
F.	Faculty Results by Quartile with Number of Respondents, Total Percentage of Favorable Response, "X2", Degrees of Freedom, and Significance at P > .05.	. 200
G.	Fourth Quartile Results by Battery with Number of Respondents, Total Percentage of Favorable Response, "X2", Degrees of Freedom, and Significance at P > .05.	. 205

Appendix		Page
н.	Third Quartile Results by Battery with Number of Respondents, Total Percentage of Favorable Response, "X2", Degrees of Freedom, and Significance at	
	P > .05	210
I.	Second Quartile Results by Battery with Number of Respondents, Total Percentage of Favorable Response, "X", Degrees of Freedom, and Significance at P > .05.	215
J.	First Quartile Results by Battery with Number of Respondents, Total Percentage of Favorable Response, "X2", Degrees of	220
	Freedom, and Significance at P > .05	220

CHAPTER I

THE PROBLEM

Purposes of the Study

The purposes of this study are: (1) the determinination of the discriminating ability of the What Do You Think About Your Schools, an instrument used to measure the attitudes of elementary pupils, secondary students, parents, patrons, and faculty; (2) to establish the instrument's reliability; and (3) to establish norms for the instrument.

Significance of the Problem

The need for evaluation of the school curriculum has been a continuing problem. Progress is evident, but much remains to be investigated before an objective measure of curriculum can be accomplished. It is a fundamental assumption of this study that curriculum can be delineated and measured. The American people have made significant contributions in financing public education and are demanding to know what they are getting for their money. This, coupled with the concern educationists and laymen have for the function of education

in society, requires the development of accurate instrumentation for assessment of educational program.

The difficulties in measuring curriculum are hindered by inadequate definitions of curriculum, changing educational patterns, variations in community characteristics, confusion over educational goals, and by social forces within our pluralistic society. The development of a theoretical model for the assessment of curriculum may establish a better basis for assessment and for deployment of limited resources.

Dr. Herbert C. Rudman¹ has conceptualized the school curriculum as consisting of four constituent elements: (1) educational program, (2) educational services, (3) organization of the school system, and (4) the values for education held by elementary pupils, secondary students, parents, faculty, and adults in the community who do not have children in the public school system (patrons).

The first three criteria lend themselves to systematic description. The child's <u>educational program</u> consists of course offerings and laboratory experiences (i.e., Fine Arts, Home Economics, Physical Education, field trips, foreign language laboratories, science

Herbert C. Rudman, "The Curriculum" (unpublished report, Michigan State University, East Lansing, Michigan, February 12, 1968), p. 1.

laboratories, and the like). The educational services consist of administration, supervision of teachers, lunch programs, transportation, libraries, testing programs, counseling and guidance, etc. The organization of the school district refers to the actual manner in which school functions are carried out at the classroom, building, and district levels. This has a profound influence on the effectiveness of educational programs and services within the school district. The remaining variable is the values held by the various members of These values reflect the expectations the community. and, therefore, the outcomes of the curriculum in many cases. It is this facet of curriculum assessment that has proved most troublesome for accurate measurement.

The Educational Characteristics Criterion (ECC) was developed and tested by Rudman, 2 Kraft, 3

²Herbert C. Rudman and Stanley E. Hecker, "The Determination and Measurement of Factors Which Directly or Indirectly Affect Quality of an Educational Program" (unpublished proposal, Michigan State University, East Lansing, 1961).

³Leonard E. Kraft, "The Perceptions Held by Professors of Education, Professors in Areas other than Education, and School Board Members on Ninety Factors Which May or May Not Affect the Quality of an Educational Program" (unpublished Ed.D. dissertation, Michigan State University, 1962).

Berg, Mueller, Springer, and Pelton. This instrument was used to measure educational quality. Educational quality was defined as those educational characteristics of school and community which are perceived effective in accomplishing the goals of public school education. The judgments of educational experts were found to agree on fifty-six characteristics which a good school program should contain. It was found that a significant correlation existed between quality education, as defined by experts, on the one hand and teachers and administrators on the other. When school districts were classified on high and low financial support quartiles, the ECC discriminated between districts on the basis of financial support with a high degree of reliability. A set of

Arthur D. Berg, "The Determination of the Discrimination and Reliability Indices of the Educational Characteristics Criterion with Implications Concerning Educational Cost-Quality Relationships" (unpublished Ph.D. dissertation, Michigan State University, 1962).

⁵Van Dyck Mueller, "A Study of the Relationship Between Teacher-Administrator Perceptions of Education Quality as Measured by the <u>Educational Characteristics Criterion (ECC)</u> and Selected Cost Factors" (unpublished Ed.D. dissertation, Michigan State University, 1964).

⁶ Owen Springer, "A Study of the Relationships Between the Educational Characteristics Criterion (ECC), the Stanford Achievement Test, and Selected Cost Factors" (unpublished Ed.D. dissertation, Michigan State University, 1964).

Maurice D. Pelton, "A National Analysis of Educational Quality as Measured by the Educational Characteristics Criterion (ECC), Achievement, and Selected Cost Factors" (unpublished Ed.D. dissertation, Michigan State University, 1966).

norms based upon <u>ECC</u> responses were established for school districts in high and low financial support quartiles. This information provided an objective measure of the community attitudes variable as it related to total curriculum assessment. Future curriculum studies would have an objective base for comparison of school district educational programs.

The major difference between the ECC and the What Do You Think About Your Schools is the factor of educational quality. The ECC provides a quality measure based on the judgments of educational experts in relationship to financial support quartiles. The purpose of this study is to base the attitudes of elementary pupils, secondary students, parents, patrons, and faculty on financial support quartiles to add another dimension to the assessment of the school program. What Do You Think About Your Schools makes no assumption that the attitudes of the respondents are directly related to quality. Rather, the attitudes of the respondents can give a more complete picture of the total values held by the community. If these attitudes reflect a consistent discriminating ability -- as measured in terms of financial support quartiles -- a better assessment of the curriculum can be derived. Norms can be established for each financial support quartile by the application of statistical limits of probability for each school district assessed, and future applications of the batteries can have real meaning in terms of curriculum development.

Assumptions of this Study

This study assumes that the curriculum assessment model developed by Rudman is workable. There are many theoretical models for curriculum assessment, but Rudman's exhibits a more pragmatic approach in terms of actual behavior observed in the public schools.

The school of perceptual psychology holds that individuals react to their environment as they perceive it. For purposes of this study, it is assumed that this theory is tenable, and that the attitudes of students, parents, patrons, and faculty will reflect the community's expectations of the educational program.

It is assumed that the attitudes of students, parents, patrons, and faculty can be accurately expressed regarding areas of the school program about which they have knowledge and opinions. It is also assumed that these attitudes can be classified into the following categories: (1) satisfaction with school, (2) school program, (3) essential services desired, (4) school organization and size, (5) school plant, and (6) community relations.

The basis for assuming the relationship between the factors of size, effort, ability, and expenditure and quality of the educational program is derived from the

results of research in the area of cost-quality relation-From this research it is assumed that there may be a similar relationship between the attitudes of the school community and size, effort, ability, and expenditure. Size of a school district is defined as the average daily membership (ADM) in grades kindergarten through twelve in the State of Michigan. Effort is defined as the total operational millage levied based upon the final appraisal of real and personal property valuation of school districts in the State of Michigan. Ability is defined as the total value of real and personal property of the school district divided by the average daily membership. Expenditure is defined as the amount of total dollars expended for elementary and secondary education in the school district divided by the average daily membership. This does not include expenditures for capital outlay and debt retirement. Financial Support Quartile is defined as a composite ranking by quartile of a school district according to size, ability, effort, and expenditure.

The classification of Michigan school districts according to size, effort, ability, and expenditure into quartiles gives a concrete base for developing item norms

For one example see, William S. Vincent, "Quality Control: A Rationale for Analysis of a School System," IAR Research Bulletin, Vol. I, No. 2 (January, 1961), pp. 1-7.

You Think About Your Schools can be compared to a base population of school districts with similar financial support levels and a statement of statistical probability can be derived. If there is a wide divergence in district score from the norms established for a similar financial support district, statements of a diagnostic nature for that specific item can be made. A specific item will have meaning only when compared to districts of like financial support. The attitudes of elementary pupils, secondary students, parents, patrons, and faculty will give the total curriculum assessment model better data for judgments regarding the quality of the educational program.

Delimitations of the Study

The parameters of this study are delimited by the following factors:

1. The major variables in this study are the individual's attitudes, as measured by the What Do You Think About Your Schools and the cost factors of size, effort, ability, and expenditure as derived from the Michigan Education Association's, Ranking of Michigan High School Districts by Selected Financial Data for 1966-67.

- 2. The study is limited to a sample of fourth, third, second, and first quartile school districts in the State of Michigan. No results are drawn from individual school districts.
- 3. The statistical analyses are limited to determining: (1) the reliability of the What Do You Think

 About Your Schools, (2) the discrimination of items
 between elementary pupils, secondary student's parents,
 patrons, and faculty within quartiles, and (3) the discrimination of items between fourth, third, second, and
 first quartiles and each individual respondent group.
- 4. This study uses only selected financial cost factors and does not include all possible permutations of cost analyses.
- 5. The findings of a relationship between school-community attitudes and financial factors are viewed as associational and not causal.
- 6. The study assumes that the individual sampled will respond to the instrument with his true perceptions in regard to the school-community situation.

Definition of Terms

Public schools. -- The term, "public schools" refers to Michigan elementary and secondary schools in school districts which maintain grades kindergarten through twelfth. Any school which receives full support of its program from state or federal sources will be excluded.

School district. -- A school district is a legal entity created by the Michigan State Legislature for the purpose of operating and maintaining public education within the boundaries established by law.

State equalized valuation. -- State equalized valuation is the final appraisal by the Michigan Tax Commission of the worth of real and personal property in the State of Michigan.

Mill.--A mill is the value of a tenth of a cent or thousandth of a dollar.

Size.--Size is the number of public school member-ship as computed on the fourth Friday following Labor Day of each year. All pupils, to be included must be at least five years old on December first.

Financial ability. -- Financial ability is an expression of the state equalized valuation (SEV) divided by the total number of resident pupils. This figure shows the dollar amount in local equalized valuation behind each resident child.

<u>Financial effort</u>.--Financial effort is the number of mills levied on the state equalized valuation (SEV) for the purpose of operating the school district.

<u>Financial expenditure</u>.--Financial expenditure is the cost per pupil expended in operation of the school district exclusive of school board salaries, tuition expense, capital outlay, and transportation.

Financial support quartile. -- A financial support quartile is a composite quartile ranking of a school district according to size, financial ability, financial effort, and financial expenditure.

<u>Curriculum assessment</u>.--Curriculum assessment or educational program assessment is defined as the measurement of a theoretical model of curriculum that encompasses educational program, educational services, school organization, and values held by the community toward the curriculum of the school district.

What Do You Think About Your Schools. -- An instrument measuring attitudes of elementary pupils, secondary students, parents, patrons, and faculty concerning the educational program of the school district.

Elementary pupil. -- An elementary pupil is a student enrolled in grades kindergarten through six in the public schools of Michigan. For the purposes of this study, an elementary pupil is defined as a sixth grader.

Secondary student. -- A secondary student is a student enrolled in grades seven through twelve in a public school of Michigan. For the purposes of this study, a secondary student is defined as a twelfth grader.

<u>Parents</u>.--Parents are defined as father, mother or legal guardian of pupils in the public schools of Michigan.

<u>Patrons</u>.--Patrons are defined as taxpayers in a school district of Michigan who do not have children in the public schools.

Faculty. -- Administrators and teachers of a public school in Michigan whose positions provide instruction or supervision of pupils in the public schools are defined as faculty.

Hypotheses

General Hypothesis I

The What Do You Think About Your Schools instrument will discriminate between the attitudes of elementary pupils, secondary students, parents, patrons, and faculty on the basis of financial support quartile of Michigan school districts (K-12). Support quartile is defined in terms of size, effort, ability, and expenditure.

Operational HIa. -- The instrument will show ability to discriminate between fourth, third, second, and first financial support quartile districts on each item score according to elementary pupil responses.

Operational HIb. -- The instrument will show ability to discriminate between fourth, third, second, and first financial support quartile districts on each item score according to secondary student responses.

Operational HIc. -- The instrument will show ability to discriminate between fourth, third, second, and first financial support quartile districts on each item score according to parent responses.

Operational HId. -- The instrument will show ability to discriminate between fourth, third, second, and first financial support quartile districts on each item score according to patron responses.

Operational HIe. -- The instrument will show ability to discriminate between fourth, third, second, and first financial support quartile districts on each item score according to faculty responses.

General Hypothesis II

The What Do You Think About Your Schools instrument will show ability to discriminate between responses of elementary pupils, secondary students, parents, patrons, and faculty within fourth, third, second, and first financial support quartiles of Michigan school districts.

Operational HIIa. -- The instrument will discriminate between responses of elementary pupils, secondary students, parents, patrons, and faculty on each item score for fourth financial support quartile districts.

Operational HIIb. -- The instrument will discriminate between responses of elementary pupils, secondary

students, parents, patrons, and faculty on each item score for third financial support quartile districts.

Operational HIIc. -- The instrument will discriminate between responses of elementary pupils, secondary students, parents, patrons, and faculty on each item score for second financial support quartile districts.

Operational HIId. -- The instrument will discriminate between responses of elementary pupils, secondary students, parents, patrons, and faculty on each item score for first financial support quartile districts.

Research to Improve the Instrument, What Do You Think About Your Schools

The need for improving the precision of the instrument required that two additional studies be done: (1) a study of reliability by item, and (2) the establishment of norms for the total sample of school districts. These two studies would establish the stability of the instrument from test to test and would provide norm populations against which users of the instrument could make comparisons.

The study of reliability provides a reliability coefficient for each item on the elementary pupils, secondary students, parents, patrons, and faculty test batteries. To accomplish this objective, scores on the first test of a sample of each battery were correlated

with scores for the same sample and test three weeks later.

The establishment of norms for elementary pupils, secondary students, parents, patrons, and faculty were determined by total sample. The level of probability for the "true score" of the universe of respondents was estimated by computing the standard error of a percentage at the .95 level. This provides users of the What Do You Think About Your Schools a range of percentages of favorable response for the sample population to which other scores can be compared with .95 confidence.

Organization of the Thesis

This chapter has stated the purposes of the study, importance of the problem, the assumptions upon which the study is based, the delimitations of the study, the definition of terms, the general and operational hypotheses, and the research to improve the precision of the instrument.

Chapter II presents a review of related literature. This review includes theoretical models of curriculum assessment, development of various instrumentation for program assessment, and a report of significant studies relating to cost-quality relationships.

Chapter III deals with the instrumentation and methodology of the study. A complete description of the instruments used to collect the data, method of sample

selection, classification of cost factors, research design, and proposed statistical treatment of the data are presented.

Chapter IV presents the statistical tests and results of the data in relation to the hypotheses.

Chapter V presents the statistical tests and results of the studies to improve the precision of the instrument.

Chapter VI presents the conclusions, implications, summary and areas recommended for further research.

CHAPTER II

RELATED LITERATURE

In the last fifty years many studies and surveys have been undertaken to assess school programs. The majority of these studies and surveys have come from two fields of endeavor: (1) school finance, and (2) curriculum development. The first tends to have a broader outlook in describing and evaluating the total school program. The second emphasizes a narrower definition of school program, primarily the outcomes of subject-matter taught in the schools. Finance and curriculum experts have not been able to agree on mutually acceptable definitions and philosophical viewpoints, and therefore their models for evaluation differ.

It is the purpose of this chapter to analyze and summarize the important models and research that has been done on program assessment. Three major areas will be discussed: (1) the major models for curriculum or educational program evaluation, (2) empirical studies of cost-quality relationships, and (3) instruments used to evaluate school programs.

National Interest in Educational Assessment

The past decade has seen a ferment in American education. Much has been said and written on the strengths and weaknesses of our educational system. Seminars, conferences and legislative committees have investigated various aspects of the American schools in seeking improvement through new programs and evaluation. In a 1962 conference held by the United States Office of Education several statements were issued relating to education's role in society: (1) a nation's strength lies in the strength of all its people; (2) it is tested in the aspirations of its youth and the quality of its schooling; (3) our democracy is no stronger than the moral and intellectual fiber of our people; (4) our country can be no richer than our teacher's minds and our children's opportunities; (5) since the quiet strength and latent power of education is less tangible than arms and missiles, it has been more difficult to realize; and (6) American education has become the testing ground for democracy. 1

A statement by the Committee of Economic Development echoed the above statements when it said:

United States Department of Health, Education, and Welfare, Education for Freedom and World Understanding, Bulletin OE10016 (Washington, D.C.: U. S. Government Printing Office, 1962), pp. 50-51.

A democracy lives or dies by the ability of its people to choose wisely. We need better schools to teach us how to understand the alternatives before us, and how to choose wisely among the real alternatives.²

Much of the move to improve education has come through discussion of assessment programs. Rickover³ in 1962 proposed to a House Committee that a national testing program be established to determine the educational levels of youth graduating from schools. He based his arguments on the policies of foreign countries, which have national testing programs, leading to valid diplomas. Rickover's concern for the allegedly falling standards of education and the wasted resources committed to vast numbers of youth who never finish their education prompted him to submit this suggestion.

As could be expected, the majority of American educators do not agree with this position. Stoddard exemplifies the majority opinion:

Do we prefer what the Bonn government has set up-a system of examinations through which the decision to go to the university, or not to go is firmly made when a child is ten years old?-- (The result: only one out of twenty pupils make the grade.) 4

²Ralph Lazarus, <u>We Can Have Better Schools</u> (New York: Committee for Economic Development, 1959), p. 4.

³Hyman G. Rickover, "Education for All Children" Hearing Before the Committee on Appropriations, House of Representatives, Eighty-Seventh Congress, 1962, p. 139.

⁴George Stoddard, "The Issues That Divide Us," School and Society, 86:237 (May 24, 1958).

McNeil suggested some of the reasons for opposition to program assessment by educators in general:

- 1. Educators are afraid the evidence will be collected only in respect to recall information and the competency of few skills.
- 2. Educators tend to be process-oriented. They believe teaching conditions are important and therefore make less effort to assess the learning that does occur from specific situations.
- 3. Educators distrust the way the results will be interpreted, knowing that a multitude of factors can be responsible for specific learnings.
- 4. There is the danger that favored practices and ideas may not prove to be valuable.⁵

Over the opposition, movement has been made toward a national assessment project. The Carnegie Corporation funded a project to construct instrumentation designed to assess education on a national basis. The major purposes of the project are: (1) to find the strengths and weaknesses of education on a regional basis for the whole nation, (2) to provide information to schools for research on educational problems, (3) to provide international

John D. McNeil, <u>Curriculum Administration</u> (New York: The MacMillian Company, 1965), pp. 115-116.

comparisons, and (4) to increase interest in education throughout the United States.

Typical of the criticism pointed at national assessment are the observations made by Hand:

. . . I am opposed to (1) a national testing program set up for purposes of comparing a school or schools in one district or region with those in other districts or regions, and (2) the way in which ECAPE is functioning. I am opposed to a national testing program set up for purposes of comparing schools chiefly because (a) it would set up new obstacles to realization of our goal of equality of educational opportunity, (b) it would be the nose under the tent which would be followed by a monstrous camel in the form of a centrally controlled curriculum, (c) it would stultify the curriculum, (d) it would stifle local innovation and experimentation in respect to the classroom, (e) it would result in unbearable pressures on classroom teachers and school administrators, and (f) it would encourage cheating on the part of students and teachers alike. I am opposed to the way ECAPE is functioning chiefly because it is violative of a cardinal principle of American democracy . . . namely, the principle_of government by the consent of the governed.7

There is no doubt that the schools need to be assessed, but it is the manner of this assessment that differs from individual to individual. Vincent and Mac Gregor have reviewed various approaches to the assessment of schools: (1) Bestor uses the criterion of stability. This is to be expected, as Bestor is a

⁶John W. Gardner, "A National Assessment of Educational Progress" (unpublished report, The Carnegie Corporation, April 23, 1965), p. 1.

⁷Harold C. Hand, "The Camel's Nose," <u>Phi Delta</u> Kappan, 47:9,12 (September, 1965).

historian and concerned with the culture and traditions of the past; (2) Zoll is concerned with economy—the minimum of funds necessary to carry on a basic program is all that is required; (3) Rickover wants more central control of education; (4) Conant is concerned with the equality of opportunity and therefore has promoted foundation—type programs for schools; and (5) education—ists tend to judge schools by their adaptability. This means the currency of procedures and program utilized in meeting the needs of a changing society.

The National School Boards Association and the American Association of School Administrators believe in evaluation and have established the following criteria:

(1) evaluation should be based on stated objectives,

(2) evaluation should be based on intimate and comprehensive knowledge of the community, (3) evaluation should be continuous, (4) evaluation should be comprehensive,

(5) evaluation should involve many people, (6) evaluation should be positive as well as negative, (7) evaluation should use many methods, (8) evaluation should be based on knowledges of students, (9) evaluation should require the administration and board to look at itself, (1) evaluation should appraise staff policies, (11) evaluation is

⁸William Vincent and Archie Mac Gregor, 1959 Review of Fiscal Policy for Public Education in New York State - Public Tests of School Quality (New York: New York State Educational Conference Board, 1960), pp. 1-2.

based on the belief that people make a difference, and (12) evaluation should bring forth improvement.⁹

America is concerned about the quality of its schools. It is the means that differ. A quote from Dewey is as relevant today as it was in 1900 concerning the quality of education:

What the best and wisest parent wants for his own child, that must the community want for all its children. Any other ideal for our schools is narrow and unlovely; acted upon it destroys our democracy. 10

Theoretical Models for Program Assessment

The Sequential Simplex Model of Mort and Furno

The <u>Sequential Simplex Model</u> is a statistical tool used to control gross differences among communities so that more subtle differences can be examined. The basic premise of this model is that the quality of a school can be viewed in concentric circles, with quality at the core. Around this core of quality is "The School" and its staff which have the most direct influence on quality. The next layer of the circle is the "School System Policy," which includes adequacy of salary, staffing

⁹American Association of School Administrators and National School Boards Association, <u>Judging Schools with Wisdom</u> (Washington, D. C.: National Education Association, 1959), pp. 1-11.

¹⁰ John Dewey, The School and Society (Chicago, Illinois: University of Chicago Press, 1900), p. 10.

ratios, and materials and supplies. The third layer is the "Educational Climate" which includes the socioeconomic and financial factors of the community. The final layer is "Community Characteristics" represented by physical conditions, wealth, and socio-economic characteristics. Running through all these circles is the common core of quality. 11

A school's overall quality can be determined by a "Quality Control Chart" which plots the school's position in relation to the four factors that influence total school quality. If the scores on various instruments depart significantly from the established norms, diagnostic measures can be applied. The instrumentation used to measure these four factors has been derived from studies and surveys conducted by Mort and his associates over the past 30 years. These instruments are correlated with the <u>Growing Edge</u> which is the established criteria for quality in the Sequential Simplex Model.

The major weaknesses of this model are: (1) the assumption that the <u>Growing Edge</u> is a good measure of quality. The <u>Growing Edge</u> is a measure of adaptability of innovative practices in schools, and therefore considerable doubt can be cast that this is a legitimate

¹¹ Paul R. Mort and Orlando F. Furno, Theory and Synthesis of a Sequential Simplex (New York: Institute of Administrative Research, Teachers College, Columbia University, 1960), p. 15.

criterion for quality. Innovative practices are continuously changing. This has been particularly true in the last ten years when new programs and materials have flooded the educational market. A continuous up-dating of the instrument would be necessary to find out what is truly innovative in American education today; (2) the assumption that expenditure is an independent variable is also questionable. The research over the past years seems to indicate that expenditure is a dependent variable of school quality; and (3) the complex computations of correlations and multiple correlations makes the use of the model limited to persons with a high degree of statistical competency.

Experimental Appraisals of Curriculum Patterns--Smith, Stanley, and Shores

Smith, Stanley, and Shores (1950) represents a group of American educators, mainly curriculum specialists, that view the evaluation of school program in a limited fashion. Their major work, <u>Fundamentals of Curriculum Development</u> defines what they mean by curriculum:

A sequence of potential experiences if set up in the school for the purposes of disciplining children and youth in group ways of thinking and acting. This set of experiences are referred to as the <u>curriculum</u>. 12

¹²B. Othanel Smith, William O. Stanley, and J. Harlan Shores, Fundamentals of Curriculum Development (New York: The World Book Company, 1950), p. 4.

The important words in this definition are "in the school." This implies that youth acquire most of their learning in a school situation.

Accepting this definition of curriculum, Smith, Stanley, and Shores outlined their model for curriculum evaluation:

- 1. The theory of the curriculum pattern must be stated clearly in order to be tested.
- 2. The social, psychological, educational, and physical conditions under which the curriculum pattern it to be tested must be clearly spelled out.
- 3. The anticipated results of the curriculum theory must be stated as hypotheses.
- 4. Data must be collected to ascertain whether or not the hypotheses derived from the curriculum theory were borne out by observed facts; for as these hypotheses are tested out, confirmed or invalidated by observation, the theory is affirmed or denied. 13

Attention must be given to the attitudes, beliefs, and knowledges of all members of the community, but the validity of these factors cannot be established; therefore the authors believe the evaluator must progress without this knowledge.

After a summary of six important curriculum studies, the authors gave their conclusions of curriculum evaluation:

¹³<u>Ibid</u>., p. 583.

falls considerably short of meeting desirable standards of research design. In general, neither the curriculum theories under test nor the hypotheses that should have been derived from these theories were made explicit. As a consequence, those who examined this research have real difficulty in knowing just what propositions were proved or disproved.

In reviewing this theory of curriculum evaluation several criticisms seem to be in order:

- 1. The definition of curriculum is much too limited. Children learn from all aspects of their environment. The school is one factor, but it is hardly all-inclusive.
- 2. Smith, Stanley, and Shores assert that community values cannot be measured, but other research indicates that an attempt can be made at evaluation of other factors influencing the schools, such as community values, finance, etc.
- 3. No measurable results are obtainable from this theory, therefore its value for comparisons or improvement is limited.
- 4. It leaves the reader feeling that nothing in the school program can be tested, because of the many variables, therefore evaluation of school program is a waste of time.

¹⁴<u>Ibid</u>., p. 612.

School Program Assessment--Rudman

The school program assessment model used in this study and developed by Rudman was discussed in Chapter I. To review briefly, the school program consists of four major elements which can be objectively assessed: (1) the educational program, (2) the educational services, (3) the organization of the school, and (4) community attitudes and values. 15

The use of objective measures to assess the elements of school program gives a pragmatic application to this theory. Schools can compare themselves to ideal models, state and national norms. As of this time instruments have been developed and tested to evaluate educational program, educational services, and teachers' and administrators' attitudes. This study hopes to add elementary pupils', secondary students', parents', patrons', and facultys' attitudes to the measurement of community values. The remaining factor of educational organization is still under study. Research is conflicting regarding the influence of school organization on total program quality. If measurable results for school organization can be obtained in the future, the total theory will be developed.

¹⁵ Rudman, "The Curriculum," p. 1.

Other Models for Curriculum Assessment

Most of the recent models for curriculum assessment define curriculum in a limited way, similar to Smith, Stanley, and Shores. The distribution of federal funds has dramatically increased the activity to construct models for the assessment of specific programs, but nothing has been done to give a broad view to total program assessment.

The major differences between models now developing seems to be the point of emphasis. Some of these models:

(1) point to the importance of the teacher as a developer of curricula, (2) rely on the developer's intuition rather than his rational skills, (3) emphasize subject-matter-content goals as opposed to intellectual-process-and-skill goals, (4) go beyond the stated goals, and (5) incorporate plans for reassessing goals during and after the development phase. 16

The movement is to define the problem in smaller units to which appropriate research techniques can be applied with success. Although this normally brings forth carefully drawn research studies there is some doubt whether the findings of these studies can be applied in the broad scope of school program assessment.

¹⁶ Robert S. Stake, "Testing in the Evaluation of Curriculum Development," Review of Educational Research, Vol. 38, No. 1 (February, 1968), p. 78.

Summary

- 1. There is extreme interest on all levels of American education to develop a systematic method for evaluating school systems.
- 2. No one model has been able to control all the variables that influence educational quality.
- 3. The various models stress different variables, whether they be finance, program, or subject-matter.
- 4. The trend in assessment is toward fragmentation of the school program to apply better controls for experimentation.
- 5. There is need for a comprehensive look at all segments of the school program to bring the various parts into a meaningful whole. Until this is accomplished little can be determined about the quality of schools at the local, state, or national levels.

Related Cost-Quality Studies

Early Cost-Quality Studies

There has been intense interest in the area of cost-quality studies since the early 1920's. Most of these studies follow the normative approach. They try to explain how schools differ at varying expenditure levels by placing a value statement on certain aspects of the school program.

One of the earliest studies with a normative framework was conducted by Ayres in 1920. He constructed a ten-item index to which he correlated state expenditures for the years 1896 through 1920. The <u>Index</u> had five expenditure items and five non-expenditure items. Ayres <u>Index</u> included:

- 1. Per cent of school population attending school daily.
- 2. Average days attended by each child of school age.
- 3. Average number of days the schools were kept open.
- 4. Per cent that high school attendance was of total attendance.
- 5. Per cent that boys were of girls in high school.
- 6. Average annual expenditure per child in school.
- 7. Average annual expenditure per child of school age.
- 8. Average annual expenditure per teacher employed.
- 9. Expenditure per pupil for purposes other than teacher's salaries.
- 10. Expenditure per teacher for salaries. 17

 Ayres found a correlation of .78 between the expenditure and non-expenditure items of his <u>Index</u>.

Norton in 1926 tried to ascertain the ability of the forth-eight states to support education. He found that high-expenditure states were: (1) spending more per

¹⁷ Leonard P. Ayres, An Index Number for State School Systems (New York: Russell Sage Foundation, 1920), p. 14.

pupils, (2) had better school plants, (3) had a longer school year, (4) had a higher preparation level for teachers, and (5) had a lower illiteracy ratio. ¹⁸ From these observations, Norton concluded that the level of education was higher in states spending more money on education.

Ferrell used an instrument to measure educational efficiency in 249 Kentucky county districts in 1937.

His Index included:

- 1. Per cent average daily attendance was of the census.
- 2. Holding power as measured by the average sum of;
 - (a) per cent eighth grade enrollment was of first grade enrollment.
 - (b) per cent high school enrollment was of total public school enrollment.
- 3. Per cent of teachers employed who have had at least three years or more of teaching experience.
- 4. Per cent of teachers employed who have had a given amount of preparation.
- 5. Per cent of teacher in relationship to pupils.
- 6. Per cent of days in the elementary school was of 200 days. 19

¹⁸ John K. Norton, The Ability of States to Support Education (Washington, D. C.: National Education Association, 1926).

¹⁹ Thomas Ferrell, Relation Between Current Expenditure and Certain Measures of Educational Efficiency in Kentucky County and Graded School Systems, Contributions to Education No. 216 (Richmond, Kentucky: Eastern State Teachers College, 1937).

He found a correlation of .92 between quality, as measured by his <u>Index</u>, and expenditure. Schools with high expenditures had better holding power, smaller classes, longer school years, and better prepared teachers.

The weakness of these early studies is the rather obvious relationship of the criteria to expenditure. It is no wonder that high correlations were derived from items that are so directly related to the expenditure of the sampled school districts.

Studies at Various Expenditure Levels

Most of the significant studies of cost-quality relationships were undertaken by Mort and his associates from 1930 to 1960. These studies were primarily concerned with showing the public what they could expect from a certain level of expenditure. This also gave impetus to the concept of a foundation program for school finances. Later a relationship was found between scores on Mort's various instruments and the adaptability of innovation by school districts. Norms were developed for these instruments at various expenditure levels to show exactly what could be expected from a given dollar investment.

Studies of High Expenditure Levels

In a study by Vincent (1945) of expenditure in the State of New York a significant correlation was found

between expenditure and quality as defined by the Mort-Burke-Fisk Guide. The New York schools represented high expenditure schools in relation to national norms. Three samples were used: (1) 52 districts were visited by field workers who collected data on the 1091 item Guide, (2) 71 districts were mailed forms which contained data on 101 items, and (3) 216 districts were analyzed through State Department reports. From these samples a correlation with expenditure was found for 73% of the items on the Guide, 99% of the items in the mailed report, and 80% of the items on the State Department report. Vincent concluded that five trends could be associated with increased expenditure:

(1) concern for the mastery of basic skills; (2) concern for the conditions of child growth; (3) attention to needs of the individual; (4) lack of dependence of teachers upon patent devices; and (5) increase proportion of teachers who are resourceful, imaginative, and intelligent.²⁰

Grace and Moe in the New York Regent's Inquiry of 1938 ranked 43 New York school districts on a five-point scale after visitation. Although no controls were applied to size, cost of living, or population sparsity, the following conclusions were reached:

High educational efficiency is not achieved without high expenditure, but many districts have high cost and distinctly inferior returns. The group of schools with superior educational results

²⁰William S. Vincent, Emerging Patterns of Public School Practices (New York: Bureau of Publications, Teachers College, Columbia University, 1945), p. 56.

spreads the greater expenditure over all the items of expense (except transportation) and also devotes a larger proportion of the entire budget to direct instruction. The best schools do not have an exceptionally small number of pupils per teacher, but pay a high average salary to the instructors. The best schools were all large, and permitted organization of fairly large classes and a rich curriculum.²¹

One of the most convincing studies with high expenditure districts was done by Woollatt (1949). The study used 33 New York and New Jersey suburban communities in the high expenditure levels. Corrections were made for population sparsity, transportation, tuition expense, and differentials between high school and elementary costs. It was found that the <u>Growing Edge</u> did differentiate among expenditure levels as well as within high expenditure schools. The correlation to total score on the <u>Growing Edge</u> and expenditure was .59. Other significant findings were:

- 1. High cost districts did a better job of teaching skills.
- 2. High cost districts did a better job of developing the child's problem solving ability.
- 3. High cost districts did a better job of build-ing good character.
- 4. There was no point of diminishing returns between expenditure level and quality scores as derived from the <u>Growing Edge</u>. 22

²¹A. G. Grace and G. A. Moe, State Aid and School Costs: Report of the Regent's Inquiry (New York: McGraw-Hill Book Company, 1938), pp. 324-329.

²²Lorne H. Woollatt, <u>The Cost-Quality Relationships</u> on the <u>Growing Edge</u> (New York: <u>Teachers College</u>, Columbia University, 1949).

Mort in analyzing all the cost-quality studies up to 1950 had this to say about Woollatt's study:

In all the data collected over thirty years of interest in this subject, there is none that speaks more convincingly than these.²³

Studies of Middle Expenditure Levels

The Pennsylvania study (1935) of 36 communities used an instrument designed by Mort and Cornell which measured adaptability. The schools studied fell between the 40th and 80th percentile of the national expenditure range. Scores on the <u>Guide for Self-Appraisal of School Systems</u> had a positive correlation of .58 with current expenditure per weighted pupil. Additional correlations were run on the per cent of business and professional workers (.59) and the general educational level (.56). It was determined that more than half the variation in adaptability scores could be accounted for by expenditure level.

Mort did a comprehensive study of Rhode Island Public Education in 1941. The Mort-Cornell Guide was given to 38 of the 39 districts in the state. A

²³Paul R. Mort, "Cost-Quality Relationships in Education," in <u>Problems and Issues in Public School Finance</u>, ed. by R. L. Johns and E. L. Morphet (New York: National Conference of Professors of Education, Teachers College, Columbia University, 1952), p. 17.

Paul R. Mort and Frances G. Cornell, American Schools in Transition: How Our Schools Adapt Their Practices to Changing Needs (New York: Teachers College, Columbia University, 1941), p. 178.

correlation of .66 was found between quality scores and expenditure. One of the important findings of this study was the realization that a large percentage of the items in the scale were not directly related to costs. Fifty-two of the 58 items dealing with classroom instruction were correlated with expenditure. One of the major conclusions drawn by Mort was:

When expenditure reaches the higher levels, where there is no longer concern with length of term, adequately trained and experienced teachers, and scores of other things that still concern all too many schools, expenditure shows up in bringing the day-to-day work of the school more fully into the life of each boy and girl. 25

West Virginia was studied in 1945 by Strayer using the Mort-Cornell Guide as the instrument to measure quality. West Virginia represented the middle level of national expenditure, although the study involved all levels within the state. The conclusions of the Rhode Island study were confirmed in West Virginia. Positive correlations were found between expenditure and quality as measured by the Guide. Again of the 58 curriculum items, only five showed no relationship with expenditure. Significantly this study again pointed out that there is

²⁵Paul R. Mort, "Cost-Quality Relationships in Education," in <u>Problems and Issues in Public School Finance</u>, ed. by R. L. Johns and E. L. Morphet, p. 24.

something about a school, other than expenditure, that is related to quality. ²⁶

Two older studies (Ferrell, 1936 and Powell, 1933)²⁷ of middle-expenditure states also showed positive relationships between expenditure and their respective measures of quality. Powell's study involved one-room schools in New York. It showed improvement in achievement test scores with increase in expenditure levels.

Studies of Low Expenditure Levels

McLure in a 1947 study of Mississippi between the relationship of expenditure and quality found that quality scores were low for schools sampled. Mississippi ranks in the lowest levels of expenditure by states. Some of the major conclusions of McLure's study were: (1) most of the buildings were poorly designed for educational purposes, (2) there were few supplementary materials, (3) there were few teaching supplies and laboratory equipment, (4) the subjects were poorly taught, and (5) there were few activities for developing good citizenship.

McLure concluded with:

²⁶George D. Strayer, Director, <u>A Report of a Survey</u> of <u>Public Education in the State of West Virginia</u> (Charleston, West Virginia: State of West Virginia, Legislature Interm Committee, 1945).

²⁷Paul R. Mort, "Cost-Quality Relationships in Education," in <u>Problems and Issues in Public School Finance</u>, ed. by R. L. Johns and E. L. Morphet, p. 32.

Perhaps most important of all next to expenditure level, there must be in the minds of the laymen and the educators the picture of what constitutes a good education.²⁸

Mort's study of Maine in 1934, which is a lowexpenditure state, found a positive correlation between expenditure and quality. The areas of evaluation were:

- (1) administrative services, (2) supervisory services,
- (3) services to atypical pupils, (4) course offerings,
- (5) school buildings, (6) instructional staff, (7) class-room procedures, and (8) home-school contacts.²⁹ It is interesting to note that most of these areas are included in the What Do You Think About Your Schools.

Other Studies of Cost-Quality Relationships

Furno conducted a 1956 study of the hypothesis that the amount of expenditure per year was not as important as the sustained level of expenditure over a long period of time. He found that schools that were high-expenditure schools in 1921 tended to be high in 1945. A

²⁸ William P. McLure, Let Us Pay for the Kind of Education We Need: Report of a State and Local Support of Mississippi's Schools (University of Mississippi: Bureau of Education Research, University of Mississippi, 1948), pp. 3-29.

²⁹ Paul R. Mort, Director, The Financing of the Public Schools of Maine (Augusta: Maine School Finance Commission, 1934).

positive correlation was found for school districts as measured in 1945 and 1955 with the <u>Growing Edge</u>. 30

Griffis in a study of 44 Texas school systems in 1955 found a significant rise in program and services with an increase in expenditure levels. His study was done by direct observation of the sampled school systems in relation to 100 modern educational practices. 31

Bothwell (1958) studied the effect of selective increases in small-expense items on the total quality of the school's program. In his sample of 71 districts across the country, Bothwell found that balancing of all items advanced quality education, while overemphasis on any one item can hinder quality.³²

A study of the St. Louis area schools in 1957 by Hirsch applied the following <u>Index</u> to measure quality:

- 1. Number of teachers per 100 pupils in average daily attendance.
- 2. Number of college hours per average teachers.
- 3. Average teacher's salary.
- 4. Percentage of teachers with more than ten years of experience.

³⁰ Orlando F. Furno, "The Projection of School Quality from Expenditure Level" (unpublished Ed.D. project, Teachers College, Columbia University, 1956).

³¹ James R. Griffis, Education Production at Three Cost Levels (Houston, Texas: Gulf School Research Development Association, 1955).

³²Bruce K. Bothwell, <u>Creative Expenditure for Qual-ity Education</u> (New York: Associated Public School Systems, 1958).

- 5. Number of high school units.
- 6. Percentage of high school seniors entering college.33

The schools were rated by a panel of experts on a five-point scale as to their quality. He found his <u>Index</u> consistent with the opinions of educational experts and a positive correlation with expenditure.

Criterion, (ECC) have shown a relationship between costfactors and quality as perceived by educational experts. Berg³⁴ in 1962 found the ECC discriminated positively between high and low financial support districts in a sampling of Michigan school districts. Mueller³⁵ replicated Berg's study on a national sample in 1964 and substantiated the findings of Berg in relationship to costfactors. The results of these studies made possible the construction of norms on a state and national level for the comparison of school quality scores.

Studies in Relationship to Size, Effort, Ability, and Need

Turck (1960) studied Michigan's school districts to see if there was a relationship between need, ability, and

³³Werner Z. Hirsch, Analysis of Rising Costs of Public Education (Washington, D. C.: Joint Economic Committee, 1959), p. 27.

³⁴ Berg, op. cit.

³⁵ Mueller, op. cit.

effort. His conclusions were: (1) there is a relation-ship between need and ability, (2) the greater the membership (need) the more likely the district is to increase effort, (3) there seems to be little relation-ship between effort and ability, and (4) other community characteristics have a great influence on the support levels of education.³⁶

Krietlow (1961) made a twelve-year longitudinal study of the relationship of size to the educational program. His study involved the differences in achievement and learning opportunities as compared to reorganized and non-reorganized districts. He concluded that students with the same intelligence showed higher achievement levels and had better learning opportunities in reorganized districts than students from non-reorganized districts. Along with this finding, Krietlow suggested that the additional cost for this improved education was \$12 per elementary pupil more than the non-reorganized districts were spending. 37

A comprehensive study of 60 Wisconsin school districts by the Midwest Administrative Center using

³⁶Merton Turck, Jr., "A Study of the Relationships Among the Factors of Financial Need, Effort, and Ability in 581 High School Districts in Michigan" (unpublished Ed.D. dissertation, Michigan State University, 1960).

³⁷ Burton W. Krietlow, <u>School District Reorganization...Does It Make a Difference in Your Child's Education?</u> (Madison: Agricultural Experiment Station, University of Wisconsin, 1961).

trained observers to rate schools on the quality of their educational program found the highest number of "excellent" ratings were among: (1) districts with the largest number of pupils, (2) districts with the largest valuation per pupil, and (3) districts that made the greatest tax effort. 38

Vincent (1961) reports several studies indicating the various correlations of ability and effort. A summary of his findings on ability, as measured in property valuation per pupil and disposable personal income, include: (1) the Metropolitan School Study Council of New York between 1940-1945 for all schools had a correlation of .77, (2) a nation-wide study by the Association of Public Schools in 1959-60 showed a correlation of .34, and (3) a sample of Pennsylvania school districts in 1939 showed a correlation of .34.³⁹ Vincent's summary of findings on the factor of effort, which is the amount of locally raised millage for school operation, include a correlation of .35 in 1940-45 for the Metropolitan School Study Council and a correlation of .48 for the same districts between 1950-1955.⁴⁰

³⁸John Guy Fowlkes and George E. Watson, <u>School</u> <u>Finance and Local Planning</u> (Chicago: The Midwest Administrative Center, 1957), pp. 74-85.

³⁹William S. Vincent, "Quality Control: A Rationale for Analysis of a School System," <u>IAR Research Bulletin</u>, Vol. I, No. 2 (January, 1961), p. 7.

^{40&}lt;u>Ibid</u>., p. 7.

Summary

- 1. All the studies reported show a positive relationship between expenditure and quality.
- 2. The relationships between expenditure and quality holds for high, middle, and low expenditure districts.
- 3. The respective measure of quality makes little difference in relationship to expenditure.
- 4. A long-range view of expenditure is more important than the short range for influencing quality of the educational program.
- 5. Community characteristics have an influence, along with expenditure, on the quality of the school district.
- 6. Certain items of expenditure correlate better with quality than total current expenditures.
- 7. The factors of size, ability, effort, and expenditure have been found to be positively related to the quality of an educational program.

Instrumentation

The previous section reviewed the empirical studies which indicate a relationship between cost-factors and quality of the school program. The purpose of this section will be to investigate the instrumentation developed for evaluation of school districts based upon: (1) locally defined values, goals, and objectives, (2) national testing programs and achievement tests, (3) quantitative

assessment of school organization, (4) quality assessment based on expert's observations, (5) assessment based on economic output and adult adjustment, and (6) assessment of community attitudes.

Introduction

The need for program assessment has been well estab-If educational programs are to become better, lished. school districts need a periodic audit to find out what they are doing, and what can be done better. The two methods of determining the quality of a school program are evaluation based on process or product. Assessment by process approaches the question through the identification of factors which describe the educational setting of the school; what is taught, how it is taught, and other factors influencing the educational program. The quality of the process is thus used to describe the quality of the educational program. Assessment by product looks at the end result of the educational program. The use of achievement tests, grades, attitudes, and adjustment inventories give measurable results which can be used to make estimates of the school's quality. Mort summarized the research that has been done on assessment over the past forty years and concluded: (1) 64% of the studies

used process-type methods for evaluation, and (2) 36% used product-type measures for evaluation. 41

Instruments Based Upon Locally Defined Values, Goals, and Objectives

A group of instruments used widely to evaluate the school program assumes that a school program should be assessed in terms of the goals it has established. One of the major instruments for this type of evaluation is the Evaluative Criteria developed by the National Study of Secondary School Evaluation. 42 This instrument con-(1) a guide for the statement of objectives and philosophy prior to the evaluation; (2) an analysis of school and community data; (3) a series of checklists to assess, (a) general principles underlying the program of the school, (b) curriculum development procedures, (c) course offerings, (d) outcomes of the program of studies, (e) special characteristics of the program of study, and (f) a general evaluation of course offerings on a five-point rating scale; and (6) a series of charts to present a picture of the overall evaluation. The total rating for the school is determined by the average score

Paul R. Mort, Walter C. Reussen, and John W. Polly, <u>Public School Finance</u> (New York: McGraw-Hill Book Company, 1960), p. 80.

⁴²National Study of Secondary School Evaluation, <u>Evaluative Criteria</u> (Washington, D. C.: The Study, 1960), pp. 3-4.

for each category. The evaluation is to be completed by educationists and lay citizens with a follow-up evaluation by professional educators.

A similar type of instrument for elementary schools was developed by the Southern Association of Secondary Schools. 43 Evaluating the Elementary School: A Guide for Cooperative Study has five sections: (1) formulation of values and goals, (2) listing of functions, (3) school program, (4) resources, and (5) plans for improvement.

No score for total evaluation is given, as the instrument is designed for improving the school program and a tool for planning curriculum change.

Evaluating agencies, such as the University of Michigan Bureau of School Services and the North Central Association of College and Secondary Schools have developed similar instruments for local and professional evaluation of school districts. The criteria used for the Criteria for Accreditation 44 and Policies, Regulations, and Criteria for the Approval of Secondary Schools 45 fall

⁴³ Southern Association of Secondary Schools, Evaluating the Elementary School: A Guide for Cooperative Study (Atlanta: Commission on Research and Service, the Association, 1951).

The University of Michigan, <u>Criteria for Accreditation</u> (Ann Arbor: Bureau of School Services, The University of Michigan, 1961), pp. 1-25.

⁴⁵ North Central Association of Colleges and Secondary Schools, Policies, Regulations, and Criteria for Approval of Secondary Schools (Chicago: The Association, 1961).

in these general classifications: (1) philosophy and objectives; (2) educational program; (3) organization, support, and control; (4) school staff; (5) library and instructional material and equipment; (6) administrative and supervisory services; (7) school plant; (8) length of school year; (9) graduation requirements, pupil load, and credits; and (10) evaluation, guidance, and testing. Observations and evaluations by educationists are given in summary form and local educators are encouraged to develop purposes and objectives to meet the local situation. No numerical summary is reported for total quality of the district.

Several instruments have been developed to stimulate interest in the school program by citizens. The National Education Association, How Good Are Your Schools? 46 is one example. It asks the study group to evaluate several parts of the school program: (1) the school program, (2) the elementary school program, (3) the junior high school program, (4) the senior high school program, (5) adult education, (6) competency and qualification of teachers, (7) materials for instruction, (8) buildings and equipment, (9) administration, (10) adequacy of finance, (11) board of education, and (12) citizen interest.

⁴⁶ National Education Association, How Good Are Your Schools? (Washington, D. C.: The Association, 1958), pp. 1-31.

Yardsticks for Public Schools by the National School Board Association 47 is another example of an instrument designed for citizens to evaluate their schools. The guide permits citizens to measure tangible factors that measure school quality to generate interest and concern on the part of local citizens. The citizens are asked to make judgments on the following areas: (1) goals of the school, (2) school program, (3) teachers and teaching, (4) buildings and equipment, (5) finances, (6) organization and administration, and (7) citizens action. The results of each area are combined to give a total assessment of the school district.

<u>Instruments Used to</u> <u>Measure Achievement</u>

The measurement of student output in amount of subject-matter learned has been the major concern of several achievement tests. Achievement tests can provide a partial estimate of competency for a school district in relationship to basic skills, appreciation of and interest in knwoledge, our cultural heritage, human relations, and citizenship. The 1964 edition of the Stanford Achievement Test is described by a reviewer in The Sixth Mental Measurement Yearbook as one of the best

⁴⁷ National School Board Association, <u>Yardstick for Public Schools</u> (Evanston, Illinois: The Association, 1959).

achievement tests on the market today. 48 The Stanford consists of five batteries: (1) Primary I, (2) Primary II, (3) Intermediate I, (4) Intermediate II, and (5) Advanced. These batteries range in grade placement from 1.6 years to 9.6 years. The specific skills tested are presented in Table 1.49 Total and sub-total scores can be derived for each battery. These scores can be interpreted as a partial measure of school program quality. Studies by Springer 50 and Pelton 1 using the Stanford as a measure of achievement found a small but positive correlation between quality and achievement test scores.

Two other highly regarded achievement tests that have been used as a partial measure of quality are the Iowa Tests of Basic Skills for elementary pupils and the Iowa Tests of Educational Development for high school students. The first test gives a total score of

⁴⁸ Oscar Kristen Buros, ed., The Sixth Mental Measurement Yearbook (Highland Park, New Jersey: The Gryphon Press, 1966), pp. 123-124.

Truman L. Kelley, et al., Stanford Achievement
Test, Technical Supplement (New York: Harcourt, Brace, and World Inc., 1966), p. 11.

⁵⁰ Springer, op. cit.

⁵¹ Pelton, op. cit.

^{52 &}lt;u>Iowa Tests of Basic Skills</u> (Boston: Houghton-Mifflin Company, 1956).

^{53 &}lt;u>Iowa Tests of Educational Development</u> (Chicago: Science Research Associates, 1955).

TABLE 1.--A comparison of skills by battery for the Standford Achievement Test.

Primary I	Primary II	Intermediate I	Intermediate II	Advanced
Word Reading	Word Meaning	Word Meaning	Word Meaning	
Paragraph Meaning	Paragraph Meaning	Paragraph Meaning	Paragraph Meaning	Paragraph Meaning
Vocabulary	Science & Social Studies Concepts			
Spelling	Spelling	Spelling	Spelling	Spelling
Word Study Skills	Word Study Skills	Word Study Skills		
	Language	Language	Language	Language
	Arithmetic Computation	Arithmetic Computation	Arithmetic Computation	Arithmetic Computation
Arithmetic	Arithmetic Concepts	Arithmetic Concepts	Arithmetic	Arithmetic Concepts
		Arithmetic Applications	Arithmetic Applications	Arithmetic Applications
		Social Studies	Social Studies_	Social Studies
		Science	Science	Science

pupil achievement plus five sub-scores by subject matter areas. The second test provides a total score and scores for nine sub-tests.

The latest movement in achievement testing has been the development of a program of national assessment under the sponsorship of the Carnegie Corporation. The purposes of national assessment were stated by Gardner in 1965.

A well-conceived and well-executed assessment would, hopefully, serve several important purposes. First, it would give the nation as a whole data on the strengths and weaknesses of the American educational system. Thus, it might constitute a much more accurate guide than we currently possess to the allocation of public and private funds -- where they are needed, what they achieve -- and to many other decisions affecting education. Second, assessment results, especially if coupled with auxiliary information on characteristics of various regions, communities, schools etc., would provide data necessary for research on educational problems and processes which cannot be undertaken now. Third, when sampling and testing procedures are adequately developed, international comparisons might be possible. And finally, it is hoped that a national assessment of education would make all groups more vitally interested in the educational system -- not just in where it stands, but also in what its goals should be and how it might be improved. 54

Despite spirited opposition, the national assessment program has progressed. Items are being written and tested. When operational it will give another measure of achievement for American education. It is doubted whether the results will have any more value than the now existing

⁵⁴ John W. Gardner, op. cit., p. 1.

achievement tests, and they certainly will reflect only one aspect of the total school program.

<u>Instruments for Quantitative</u> <u>Assessment of School Programs</u>

Many studies over the past forty years have defined quality in terms of teachers employed, materials, facilities, length of school term, and holding power. One of the earliest instruments of this type was developed by Ayres. His <u>Index</u> of ten items consisted of five expenditure items and five quantitative measures. The five quantitative measures were:

- 1. Per cent of school population attending school.
- 2. Average days attended by each child of school age.
- 3. Average number of days the schools were kept open.
- 4. Per cent that high school attendance was of total attendance.
- 5. Per cent that boys were of girls in high school. The correlation between expenditure and non-expenditure items was .78.55

A 1959 study by Hirsch used a six-item <u>Index</u> to measure educational quality. These items were:

- 1. The number of teachers per 1000 pupils in average daily attendance.
- 2. The number of college hours of education of the average teacher.

⁵⁵ Ayres, op. cit.

- 3. The average teacher's salary.
- 4. The per cent of teachers with more than ten years of experience.
- 5. The number of high school credit units offered.
- 6. The per cent of high school seniors entering college. 56

A simple guide offered by the Educational Policies Commission in 1959 suggested a good education system should provide a total per pupil expenditure that was twelve per cent of a qualified beginning teacher's salary. ⁵⁷ In view of educational expenditure today, few schools reach this ideal state.

Each year <u>School Management</u> devotes a whole issue to measuring school quality in terms of expenditure. The "Cost of Education Index"⁵⁸ was designed to present regional comparisons of educational costs for local administrators. The stated goals of the <u>Index</u> are: (1) to compare district costs within regions, (2) to compare specific items of expenditure, (3) to discover areas of over-extensions or under-extensions, (4) to help construct a balanced educational program, (5) to evaluate tax effort, (6) to compare expenditure, spending patterns, and

⁵⁶Hirsch, op. cit.

⁵⁷ Educational Policies Commission, An Essay on Quality in Public Education (Washington, D. C.: National Education Association, 1959), pp. 24-25.

^{58&}quot;The Cost of Education Index," <u>School Management</u>, Vol. 9, No. 1 (January, 1965), pp. 100-151.

staffing ratios with like districts across the nation, and (7) to provide a measurement of what a quality program will cost. ⁵⁹ The basic assumption of this <u>Index</u> is that high cost districts are quality districts.

Another approach to the quantitative aspect of the school program was the instrumentation developed by Love (1968). The model developed a way of looking at program costs in a systematic way for detailed cost information concerning the school's educational program. It is possible from this analysis to compute costs per pupil, per subject, per grade, per building, or per any item used in the school's program. These findings permit judgments of need in light of actual costs for specific programs. Although no assumptions are made in relationship to program quality, a quantitative measure can be objectively assessed and judgments made as to their worth for the school district.

Instruments Based on Assessment by Educational Experts

The measurement of school program in relation to the assessment of educationists has had a long development. The observation of sound educational practices

⁵⁹<u>Ibid</u>., p. 102.

⁶⁰Bryon K. Love, "The Development of a Suggested Instructional Programs Cost Model for K-12 Districts of the State of Michigan" (unpublished Ed.D. dissertation, Michigan State University, 1968).

gives important clues to the over-all quality of the school's educational program. Much of the work done in this area of evaluation was undertaken by Mort and his associates at Columbia University.

One of the first instruments developed by Mort and Cornell was the <u>Guide for the Self-Appraisal of School Systems</u> 61 in 1937. This <u>Guide</u> had as its basic assumption of quality the speed that acceptable education ideas are adopted by a school district. Fifty-eight of the items dealt with classroom instruction, 86 items dealt with educational leadership, and 39 items assessed facilities and business management. The speed with which a district had progressed on the items gave a score of "Adaptability." The <u>Guide</u> made no final evaluation of the educational product or the underlying values to arrive at the eudcational program.

The development and testing of several instruments led to the writing of the <u>Growing Edge</u>. This instrument was also designed to measure the "Adaptability" of a school system in relation to specific school practices. The instrument was composed of two batteries, high school and elementary, which can be locally administered or given by an outside group. The secondary battery was made up of 85 items and the elementary had 64 items. The items

⁶¹ Paul R. Mort and Frances G. Cornell, <u>Guide for Self-Appraisal of School Systems</u> (New York: Bureau of Publications, Teachers College, Columbia University, 1937).

ask the evaluator to rate specific school practices in four areas:

- 1. The teaching of skills in a real or realistic fashion, and the teaching of a wide range of skills.
- 2. The teaching of areas of knowledge realistically.
- 3. The discovery and development of special aptitudes of individuals through test and tryout.
- 4. The development of gross behavior patterns 62 like citizenship, character, and thinking.

Reliability and validity coefficients were found to be high for both batteries.

The identification of characteristics that are related to a quality program was the purpose of a study by Rudman⁶³ in 1961. This study identified ninety factors which curriculum specialists and other educationists claimed influenced the quality of the educational program. Kraft⁶⁴ replicating this study on a national level asked professors of education, professors in areas other than education, and school board members what their perceptions of the ninety factors were in relation to a quality program. Kraft summarized as follows: (1) there appears

Paul R. Mort, William S. Vincent, and Clarence A. Newell, <u>The Growing Edge</u> (New York: Metropolitan School Study Council, Teachers College, Columbia University, 1946).

⁶³Rudman, "The Determination and Measurement of Factors which Directly or Indirectly Affect the Quality of an Educational Program," p. 5.

⁶⁴ Kraft, op. cit.

to be a relationship between the group the individual was a member of and his perception of the factors,

(2) there is agreement in each group as to the importance and relevance of the factors concerned with teaching and teaching methods, (3) there is agreement between groups in attributing less value to the extra-classroom category of factors, and (4) there is no relationship between the geographic region of residence and his perception of characteristics in five of the seven categories used in this study.

From this research Rudman constructed the Educational Characteristics Criterion (ECC). Further studies were carried on by Berg, Mueller, Springer, and Pelton to norm--on cost-factors--and evaluate the instrument on a state and national basis. The results of these studies were cited in the section on cost-quality research.

Instruments Based on Economic Output and Adult Adjustment

Thorndike constructed an instrument to study the effect of education and other factors on social life. His instrument, known as the <u>G. Index</u>, compared the social conditions of 1930 with the social conditions of 1900. The <u>G. Index</u> was made up of five health items, seven education items, two recreation items, eight economic and social items, five "creature comfort" items, and nine miscellaneous items taken as evidence of good

conditions. A positive relationship was found with measures of education quality, but Thorndike cautioned as follows:

On the whole, the facts which I have reported probably attach less causal efficacy to schooling, home life, and special forms of training than the general opinion of educators has attached to them. They certainly do not support the promises of educational evangelists that, if all the children for a generation or two had enough education of the right sort, they would be healthy, wealthy, and wise, living in peace and amity, free from vulgarity and meaness, busy with noble thoughts and deeds. Of

Bagley studied the correlation of quality with percentage of illiteracy, circulation of widely read magazines, persons included in Who's Who, per capita income, and percentage of white soldiers who received high grades on the Army Alpha over a thirty-year period and concluded that education was the main cause in differences of behavior and economic output. 66

Instruments Used to Measure Community Attitudes

There has been a good deal of interest in the polling of public opinion as related to school program.

Mort, Cornell, and Hinton developed the What Should Our

Schools Do? instrument. The instrument was designed to

measure the degree of "conservatism" vs. "liberalism" of

⁶⁵ Edward L. Thorndike, Education as Cause and As Symptom (New York: The MacMillian Company, 1939), p. 67.

⁶⁶W. C. Bagley, <u>Determinism in Education</u> (Baltimore: Warnick and York, 1925), Chapter 4 and 5.

a school district. The respondent answered the items by underlining either "agree" or "disagree." An example of an item reads, "A school should not expect children to reach the same standard of achievement." Mort and Cornell found a significant correlation between this instrument and expenditure in the Pennsylvania study.

Fisk developed the What Can Good Schools Do?. 68

This instrument sought to develop a rationale to explain known relationships between aspects of public understanding and school adaptability. The instrument included three elements: (1) what people knew about changing practice, (2) what people understood about why practices should change, and (3) what people of a community would accept as improvement. The respondent could answer the items by a "yes," "no," "do not understand," or "do not agree."

The Metropolitan School Study Council developed an instrument called <u>A Poll of Opinion</u> which was designed to measure knowledge of educational issues. This instrument included 27 questions. There were three forms which contained three issues on each form, and the respondent was asked to check the three best supporting sentiments for

⁶⁷ Paul R. Mort, Frances G. Cornell, and Norman H. Hinton, What Should Our Schools Do? (New York: Bureau of Publications, Teachers College, Columbia University, 1938).

⁶⁸ Robert S. Fisk, Public Understanding of What Good Schools Can Do (New York: Bureau of Publications, Teachers College, Columbia University, 1944).

his opinion on each issue. Research by the Council indicated that the poll did little more than the Mort-Cornell-Hinton Poll. 69

A Study of Public Opinion About Schools developed by Walling in 1953 again asked opinions on school situations. The respondents were asked to indicate "pleasure" or "displeasure" at hypothetical situations, such as "Many classes where you can hear a pin drop."

The initial development of the present instrument called What Do You Think About Your Schools was done by Hand (1948). Hand's instrument was described in What People Think About Their Schools. The four batteries given were: (1) elementary pupils, 32 items; (2) secondary students, 44 items; (3) parents, 49 items; and (4) teachers, 72 items. The responses to items were indicated by choosing from two to five alternatives to suggest degree of favorable response. The major differences between Hand's and Rudman's instrument are: (1) a general up-dating of all items, (2) a consistent pattern of responses, (3) measurable results in terms of

⁶⁹ Metropolitan School Study Council, A Poll of Opinion About Schools 9New York: The Council, 1955).

⁷⁰ Donald Walling, "Polls of Opinion to Measure Community Understanding of the Power of Education" (unpublished Ed.D. dissertation, Teachers, College, Columbia University, 1951).

⁷¹Harold C. Hand, What People Think About Their Schools (New York: World Book Company, 1948), pp. 153-220.

established norms, (4) the use of sampling to limit the number of respondents, and (5) the addition of a fifth group, patrons, to get a total look at community attitudes. Hand suggested that the results of his instrument were reliable enough for school program evaluation and development. Rudman's instrument makes no such implication. The results of the What Do You Think About Your Schools must be combined with a total look at school program before evaluative statements can be made.

Summary

- 1. A number of instruments have been developed to measure locally defined goals, values, and objectives.

 These instruments offer a good description of the school program in light of existing conditions. Their major weakness is the determination of education results.

 They only assume that meeting stated objectives produces a quality program. Another weakness is the lack of measurable results. No objective norms are available and therefore scores are of limited value.
- 2. Instruments based on achievement tests have been another popular method of measuring the quality of a school program. The major assumption is that high achievement scores indicate high school quality.

 Although achievement tests do give objective measures of content, it can be questioned whether this knowledge can be credited solely to the schools. Many factors

influence learning other than schools, such as intelligence, socio-economic background, cultural experience, and emotional stability.

- 3. Measures designed to assess the quality of the quantitative aspects of the school--teaching staff, salaries, facilities, etc.--assume that more and better educational results can be obtained with better inputs. These instruments have value, but there is no measurable evidence for the educational results obtained.
- 4. Instruments based on the opinion of educationists is another way of assessing school program. It is
 limited to what goes on in the schools, and does not
 reflect output or how results are obtained. The complex
 environmental factors that influence the quality of a
 school program cannot be assessed accurately by a limited
 view of the total school situation.
- 5. The measurement of output in terms of later life success is also hampered by the many factors that influence success. The schools alone cannot take sole blame or credit for the development of individuals. The schools are only one agency that have an influence on the development of mankind.
- 6. The development of opinion polls has added another dimension to school program evaluation which could not be covered in any other manner.

7. The need for a comprehensive model for educational audits is apparent. No one instrument can assess total school quality, but rather a many faceted approach is necessary. Research up to this point has been fragmented to specific areas of school program. A unified model is needed to give a complete picture of school quality. More research is needed on instruments to increase their validity and reliability to measure outcomes of school program. In the end it is not instruments, observations, or judgments that will assess education, but the mark education leaves on the history of mankind.

CHAPTER III

INSTRUMENTATION AND METHODOLOGY OF THE STUDY

Introduction

What Do You Think About Your Schools, the specific batbery items, and the comparison of battery items between
elementary pupils, secondary students, parents, patrons,
and faculty. The classification of school districts by
the financial characteristics of size, effort, ability,
and expenditure will be dealt with, as well as, the
selection of the number of school districts and number
of respondents in each school district for the sample.
The method of distribution and collection of the instrument in the school districts will be explained. The
final section of this chapter will deal with the statistical treatment of the data, the sorting procedure, and
the specific statistics used to test the research hypotheses.

Instrumentation

The What Do You Think About Your Schools was developed by Rudman over a ten-year period to analyze

community attitudes toward their school's educational The original items were developed by Harold Hand in connection with the Illinois Secondary Schools Curriculum Study in the late 1940's. These original items were further refined and expanded by Rudman through use in many community curriculum studies in the State of Michigan. These studies indicated scores on the batteries were in general agreement with the perceptions of educational experts. 1,2,3,4 The need became apparent for an objective measure of scores obtained from the five batteries. In past studies, observers were only able to make general statements in terms of the percentage of favorable response for each specific item. development of norms for each item based on pertinent financial characteristics would give the observers a solid base for measuring the attitudes of the respondents in relationship to similar school districts in the State

Planning for Educational Renewal, Hamtramck, Michigan (East Lansing, Michigan: Educational Service Series Number 19, Educational Publication Services, College of Education, Michigan State University, 1966).

²Montrose in Transition (East Lansing, Michigan: Educational Service Series Number 21, Educational Publication Service, College of Education, Michigan State University, 1967).

³ South Haven Citizens Study Their Schools (South Haven, Michigan: South Haven Board of Education, 1967). (Mimeographed.)

An Analysis of Educational Facilities: Present and Future Needs (Goodrich, Michigan: Goodrich Board of Education, 1967). (Mimeographed.)

of Michigan. The major focus of this study is to establish such norms based on financial characteristics, and to apply statistical tests to the specific items for purposes of determining the reliability and descrimination of the instrument.

The instrument is made up of five batteries which measure the attitudes of elementary pupils, secondary students, parents, patrons, and faculty concerning six categories of the school program: (1) satisfaction with school, (2) school program, (3) essential services desired, (4) school organization and size, (5) school plant, and (6) community relations. The items for each battery are worded slightly differently to meet the reading level and comprehension of the group sampled. The respondents are only asked items for which they are qualified to have an opinion. An analysis of the reading level for the elementary battery indicated a 4.6 grade level of reading difficulty. The secondary form was written at a 7.1 level as computed by the Dale-Chall readability formula. 5 Table 2 summarizes the actual number of items for each category in the batteries.

The number of actual items for each respondent group varies from thirty-one to sixty, but there are nineteen items which are responded to by all groups.

⁵Edgar Dale and Jeanne S. Chall, "A Formula for Predicting Readability," <u>Educational Research Bulletin</u>, 28:35-54 (February, 1948).

TABLE 2.--Number of items within each battery by category.

			Categories	S			
Battery	Satisfaction with School	School Program	Essential Services	Organi- zation	Plant	Community Relations	Total
Elementary Pupils	10	11	13	П	1	П	37
Secondary Students	13	11	18	Н	П	Н	45
Parents	10	11	18	ı	Н	12	53
Patrons	7	5	18	I	ı	7	31
Faculty	17	11	18	2	Т	11	09

Table 3 shows the complete tabulation of items by battery and category. The number indicated within each battery and category is the item number on the specific battery. This is provided to show which item numbers correspond across batteries. Complete batteries can be found in Appendix A.

The instrument is a pencil-and-paper questionnaire which can be completed by the individual in less than thirty minutes. The respondents indicate their attitudes on a four-point value scale. The scores are dichotomized to indicate percentage of favorable and unfavorable response for each item.

Financial Data

The financial data, which included size, expenditure per pupil, operation millage, and state equalized valuation per pupil were obtained from the Michigan Education Association. These four cost factors were chosen for two reasons: (1) empirical studies have shown a relationship between these cost factors and quality, and (2) the Educational Characteristics Criterion was normed on these four cost factors. The curriculum assessment model developed by Rudman would have one more

⁶ Michigan Education Association, Rankings of Michigan High School Districts by Selected Financial Data for 1966-67 (East Lansing, Michigan: Michigan Education Association, 1967).

TABLE 3.--Item numbers for each battery by category.

Parent Fatron	13 13 13 13 11 12 13 13 13 13 14 15 16 17 18 18 18 18 18 18 18 18 18 18 18 18 18	14 116 117 118 119 22 22 23 24 6	Z8 7
Faculty Fa	1 2 3 5 17 6 4 4 11,12,10,**	22 23 23 25 25 25 25 25 25 25 25 25 25 25 25 25	29
Secondary Student	1 2 3 3 13 6 4 7,8,9,10*	1124 122 123 123 123 124	25.5
Elementary Pupil	100 100 47 & 00	11 12 14 15 15 16 17 18 19	22 2
	Satisfaction with Schools Like School Student Feels Part of Group Teacher Knows Child Teacher Gives Help Value of Studies Proud of School Teacher Interest in Child Schools Compare Favorably Satisfaction with Schools Interest in Schools	Effectiveness of Studies Fart in Extra-Curricular Activities Amount of Work to Keep Up Amount of Homework Enough Extra-Curricular Activities School Creates Interest in Extra-Curricular Activities Variety of Subjects Subject Wanted Not Taught Help from Librarian Money Needed for School Usefulness of Subjects	Essential Services Desired Audio-Visual Physical Education

50000000000000000000000000000000000000	1.1	;	13130911188762
2 E E E E E B B B B B B B B B B B B B B	2 1	56	79 79 70 70 70 70 70 70 70 70 70 70 70 70 70
4 W 9 W 8 W 8 W 8 W 8 W 8 W 8 W 8 W 8 W 8	6 ti	84	0.000 0.000 0.000 0.000 0.000 0.000
21000010010	F 1 1	1/ 17	1
K	35	. 36	
Hot Lunch Physical and Dental Examination Job Placement Guidance Speech Correction Mental Health Clinic Remedial Reading Special Education Adult Education Agriculture Program Social Activities Summer School Driver Training	School Organization and Size School Overcrowded Supervisory Assistance	School Flant Adequate Equipment and Facilities Community Relations	

*These items are averaged to give one score for "Schools Compare Favorably."

component added to the assessment of values within the community based on a similar criterion.

The total number of high school districts in Michigan during the 1966-67 school year was 533. Of this number, complete financial data was available for 508. The City of Detroit was excluded from the sample to provide a better statistical distribution of school districts for the state. The inclusion of Detroit in the sample would have made a normal distribution impossible because of its large size and state equalized valuation. The 507 school districts in the sample accounted for 94% of the total state membership if the City of Detroit was excluded.

The 507 Michigan school districts used for this study were ranked on each of the four cost factors. This ranking was then quartiled to provide the necessary stratification of districts on the four cost factors. Table 4 shows the quartile distribution of Michigan school districts by size. The smallest district had only 98 pupils, while the largest enrolled 46,076.

Table 5 indicates the amount of taxable real and personal property per pupil from the poorest district with \$1,754 for each pupil to the richest district with \$55,347 for each pupil.

TABLE 4.--Classification of 507 Michigan school districts according to size, as of the school year, 1966-67.

Quartile	No. of Districts	Size
4	18	14,300 - 46,076
3	47	5,710 - 14.200
2	108	2,550 - 5,670
1	334	98 - 2,505

TABLE 5.--Classification of 507 Michigan school districts according to state equalized valuation per pupil, as of the school year 1966-67.

Quartile	No. of Districts	SEV per Pupil (dollars)
4	63	17,779 - 55,347
3	104	12,992 - 17,762
2	138	9,951 - 12,990
1	202	1,754 - 9,943

Table 6 shows the dollar amount spent on each child by the school district. The least spent was \$324 per pupil, and the greatest amount was \$955 per pupil.

Table 7 shows the amount of effort the community expended for education as expressed in millage for operation. The minimum was 8.00 mills and the maximum was 33.50 mills.

TABLE 6.--Classification of 507 Michigan school districts according to expenditure per pupil, as of the school year 1966-67.

Quartile	No. of Districts	Expenditure (Dollars)
4	92	525 - 955
3	114	466 - 523
2	125	426 - 465
1	176	324 - 425

TABLE 7.--Classification of 507 Michigan school districts according to mills levied for operation, as of the school year 1966-67.

Quartile	No. of Districts	Mills Levied (Effort)
4	85	19.85 - 33.50
3	114	16.22 - 19.84
2	135	13.70 - 16.21
1	173	8.00 - 13.69

Classification of Districts on Cost Factors

The school districts in the sample were assigned a quartile rank on each of the four cost factors from a possible distribution of 4-4-4-4 to 1-1-1-1. This made it possible to pick the school districts that received the same quartile rank on each of the four cost

CHAPTER III

INSTRUMENTATION AND METHODOLOGY OF THE STUDY

Introduction

This chapter will discuss the development of the What Do You Think About Your Schools, the specific batbery items, and the comparison of battery items between elementary pupils, secondary students, parents, patrons, and faculty. The classification of school districts by the financial characteristics of size, effort, ability, and expenditure will be dealt with, as well as, the selection of the number of school districts and number of respondents in each school district for the sample. The method of distribution and collection of the instrument in the school districts will be explained. The final section of this chapter will deal with the statistical treatment of the data, the sorting procedure, and the specific statistics used to test the research hypotheses.

Instrumentation

The What Do You Think About Your Schools was developed by Rudman over a ten-year period to analyze

community attitudes toward their school's educational The original items were developed by Harold Hand in connection with the Illinois Secondary Schools Curriculum Study in the late 1940's. These original items were further refined and expanded by Rudman through use in many community curriculum studies in the State of Michigan. These studies indicated scores on the batteries were in general agreement with the perceptions of educational experts. 1,2,3,4 The need became apparent for an objective measure of scores obtained from the five batteries. In past studies, observers were only able to make general statements in terms of the percentage of favorable response for each specific item. development of norms for each item based on pertinent financial characteristics would give the observers a solid base for measuring the attitudes of the respondents in relationship to similar school districts in the State

Planning for Educational Renewal, Hamtramck, Michigan (East Lansing, Michigan: Educational Service Series Number 19, Educational Publication Services, College of Education, Michigan State University, 1966).

²Montrose in Transition (East Lansing, Michigan: Educational Service Series Number 21, Educational Publication Service, College of Education, Michigan State University, 1967).

³ South Haven Citizens Study Their Schools (South Haven, Michigan: South Haven Board of Education, 1967). (Mimeographed.)

An Analysis of Educational Facilities: Present and Future Needs (Goodrich, Michigan: Goodrich Board of Education, 1967). (Mimeographed.)

of Michigan. The major focus of this study is to establish such norms based on financial characteristics, and to apply statistical tests to the specific items for purposes of determining the reliability and descrimination of the instrument.

The instrument is made up of five batteries which measure the attitudes of elementary pupils, secondary students, parents, patrons, and faculty concerning six categories of the school program: (1) satisfaction with school, (2) school program, (3) essential services desired, (4) school organization and size, (5) school plant, and (6) community relations. The items for each battery are worded slightly differently to meet the reading level and comprehension of the group sampled. The respondents are only asked items for which they are qualified to have an opinion. An analysis of the reading level for the elementary battery indicated a 4.6 grade level of reading difficulty. The secondary form was written at a 7.1 level as computed by the Dale-Chall readability formula. 5 Table 2 summarizes the actual number of items for each category in the batteries.

The number of actual items for each respondent group varies from thirty-one to sixty, but there are nineteen items which are responded to by all groups.

⁵Edgar Dale and Jeanne S. Chall, "A Formula for Predicting Readability," <u>Educational Research Bulletin</u>, 28:35-54 (February, 1948).

TABLE 2. -- Number of items within each battery by category.

			Categories	Ø			
Battery	Satisfaction with School	School Program	Essential Services	Organi- zation	Plant	Community Relations	Total
Elementary Pupils	10	11	13	-1	г	П	37
Secondary	13	11	18	٦	Н	Н	45
Parents	10	11	18	ı	Н	12	53
Patrons	7	2	18	1	ı	7	31
Faculty	17	11	18	2	Н	11	09

Table 3 shows the complete tabulation of items by battery and category. The number indicated within each battery and category is the item number on the specific battery. This is provided to show which item numbers correspond across batteries. Complete batteries can be found in Appendix A.

The instrument is a pencil-and-paper questionnaire which can be completed by the individual in less than thirty minutes. The respondents indicate their attitudes on a four-point value scale. The scores are dichotomized to indicate percentage of favorable and unfavorable response for each item.

Financial Data

The financial data, which included size, expenditure per pupil, operation millage, and state equalized valuation per pupil were obtained from the Michigan Education Association. These four cost factors were chosen for two reasons: (1) empirical studies have shown a relationship between these cost factors and quality, and (2) the Educational Characteristics Criterion was normed on these four cost factors. The curriculum assessment model developed by Rudman would have one more

⁶ Michigan Education Association, Rankings of Michigan High School Districts by Selected Financial Data for 1966-67 (East Lansing, Michigan: Michigan Education Association, 1967).

TABLE 3.--Item numbers for each battery by category.

	Elementary Pupil	Secondary Student	Faculty	Farent	Fatron
Satisfaction with Schools Like School Student Feels Part of Group Teacher Knows Child Teacher Gives Help Value of Studies Proud of School Teacher Interest in Child Schools Compare Favorably Satisfaction with Schools Interest in Schools' Future	HWWNO946 &&	1 2 3 3 13 6 4 7,8,9,10* 11	1 2 3 3 5 17 6 4 4 7,8,9,10,	* * 10000000 110	
Effectiveness of Studies Fart in Extra-Curricular Activities Amount of Work to Keep Up Amount of Homework Enough Extra-Curricular Activities School Creates Interest in Extra-Curricular Activities Variety of Subjects Subject Wanted Not Taught Help from Librarian Money Needed for School Usefulness of Subjects	11 13 13 14 10 10 10 20 21	11 12 12 12 12 12 12 12 12 12 12 12 12 1	2265 222 2265 2265 226 226 236 236 236 236 236 236 236 236	7 1 1 1 1 1 1 2 3 3 4 3 5 5 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7	11111 111100
Essential Services Desired Audio-Visual Physical Education Field Trips Music Art	2 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	25 26 28 28 28	20 32 33 33 34 35 36 36 36 36 36 36 36 36 36 36 36 36 36	28 331 351	28 8 7 110 111

E3010087651200	1 1	}	0000 00 m 00
7 F F F F F M M M M M M M M M M M M M M	25	56	0 4 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2
4 C C C C C C C C C C C C C C C C C C C	6 tr	8 7	00000 0000000 00000 00000 00000 00000 00000 00000 00000 00000 00000 00000 00000 00000 00000 000000
210084224 FFFMMMMMMMMMMM	43	† †	4 5
257 231 331 331 331 433 145 155 155 155 155 155 155 155 155 155	35	36	3.7
al Exnuic	0 54	Adequate Equipment and Facilities Community Relations	School-Community Relations Faculty Status in Community School as Informant Minutes Fublished Farent-Teacher Relations Parent-Teacher Conferences FTA Effectiveness Taxes and Services Rendered Increase Taxes Know Child's Teacher or Parent School-Community Information Interest of Parents in School Activities

*These items are averaged to give one score for "Schools Compare Favorably."

component added to the assessment of values within the community based on a similar criterion.

The total number of high school districts in Michigan during the 1966-67 school year was 533. Of this number, complete financial data was available for 508. The City of Detroit was excluded from the sample to provide a better statistical distribution of school districts for the state. The inclusion of Detroit in the sample would have made a normal distribution impossible because of its large size and state equalized valuation. The 507 school districts in the sample accounted for 94% of the total state membership if the City of Detroit was excluded.

The 507 Michigan school districts used for this study were ranked on each of the four cost factors. This ranking was then quartiled to provide the necessary stratification of districts on the four cost factors. Table 4 shows the quartile distribution of Michigan school districts by size. The smallest district had only 98 pupils, while the largest enrolled 46,076.

Table 5 indicates the amount of taxable real and personal property per pupil from the poorest district with \$1,754 for each pupil to the richest district with \$55,347 for each pupil.

TABLE 4.--Classification of 507 Michigan school districts according to size, as of the school year, 1966-67.

Quartile	No. of Districts	Size
4	18	14,300 - 46,076
3	47	5,710 - 14.200
2	108	2,550 - 5,670
1	334	98 - 2,505

TABLE 5.--Classification of 507 Michigan school districts according to state equalized valuation per pupil, as of the school year 1966-67.

Quartile	No. of Districts	SEV per Pupil (dollars)
4	63	17,779 - 55,347
3	104	12,992 - 17,762
2	138	9,951 - 12,990
1	202	1,754 - 9,943

Table 6 shows the dollar amount spent on each child by the school district. The least spent was \$324 per pupil, and the greatest amount was \$955 per pupil.

Table 7 shows the amount of effort the community expended for education as expressed in millage for operation. The minimum was 8.00 mills and the maximum was 33.50 mills.

TABLE 6.--Classification of 507 Michigan school districts according to expenditure per pupil, as of the school year 1966-67.

Quartile	No. of Districts	Expenditure (Dollars)
4	92	525 - 955
3	114	466 - 523
2	125	426 - 465
1	176	324 - 425

TABLE 7.--Classification of 507 Michigan school districts according to mills levied for operation, as of the school year 1966-67.

Quartile	No. of Districts	Mills Levied (Effort)
4	85	19.85 - 33.50
3	114	16.22 - 19.84
2	135	13.70 - 16.21
1	173	8.00 - 13.69

Classification of Districts on Cost Factors

The school districts in the sample were assigned a quartile rank on each of the four cost factors from a possible distribution of 4-4-4-4 to 1-1-1-1. This made it possible to pick the school districts that received the same quartile rank on each of the four cost

factors. Table 8 summarizes this distribution and points out the number of districts that received the same rank on size, ability, expenditure, and effort.

TABLE 8.--School districts with the same quartile rank on all four cost factors.

	Quar	tile Rank		No. of District
Size	Ability	Expenditure	Effort	No. of Districts
4	4	4	4	2
3	3	3	3	0
2	2	2	2	2
1	1	1	1	0

It is obvious from this distribution that additional flexibility would be required in order to pick a suitable number of school districts for the sample. This was accomplished by adding the total quartile ranks and providing a range within these ranks for the selection of sample school districts. Thus if a district had cost quartile ranks of 4-4-4-4 it range would be 16. If the district ranks were 4-4-4-3 its range would be 15, and the like. This allowed a district to be included in the sample if the ranks were the same on three of the four cost factors. Table 9 shows all the possible quartile ranges if the school district drops or rises one rank.

TABLE 9.--Quartile range and number of school districts represented by the four cost factors.

Quartile Range (16-15) Quartile No. of Dist.	ige (16-15) . of Dist.	Quartile E Quartile	Range (13-11) No. of Dist.	Quartile Quartile	Range (9-7) No. of Dist.	Quartile Quartile	Range (5-4) No. of Dist.
$\eta - \eta - \eta - \eta$	5	3-3-3-3	0	2-2-5-5	2	1-1-1-1	0
4-4-4-3	€.	4-3-3-3	ኒና \	2-1-5-	œ.	1-2-1-1	য
4-4-3-4	(*)	3-4-3-3	0	2-2-1-1	Ø	1-1-2-1	0
4-3-4-4	m	3-3-4-3	0	2-2-2-1	C	1-1-1-2	OJ.
3-4-4-4	0	3-3-3-4	0	3-2-2-2	7	2-1-1-1	58
		2-3-3-3	0	2-3-5-2	0		
		3-2-3-3	O	2-2-3-2	1		
		3-3-2-3	J	2-2-5-3	~ 1		
		3-3-3-2	0				
Total	10		10		<u>22</u>		34

It also shows the number of districts available for the sample within that quartile range. This added degree of flexibility provided seventy-two possible school districts from which to draw the sample school districts.

Selection of the Sample

The selection of the sample depended on selecting an appropriate number of school districts on the four cost factors and an appropriate number of respondents within each school district. The total number of districts available for the sample was seventy-two. A minimum of two districts would be required for each quartile to make statistical comparisons. The fourth quartile included ten possible districts. It was decided to pick two districts for the sample because of the large number of respondents within these two districts. larger number of districts were selected, the number of respondents for second and first quartile districts would be beyond the workable limits of the researcher. districts to be selected represented twenty per cent of the total available, therefore approximately twenty per cent of each quartile was selected for the sample. 10 shows the total number of districts rated on the four cost-factors and the number deemed necessary for this study.

TABLE 10.--Quartile ranking and the number of school districts selected for this study.

	Q ₄	Q ₃	Q ₂	Q ₁
Total N.	10	6	22	34
Sample N.	2	2	4	6

A table of random numbers was used to select the districts in each quartile to represent the twenty per cent deemed necessary for this study. The fourteen school districts selected by random numbers were contacted by mail explaining the purposes of the study and asking their participation. Twelve districts gave their immediate consent, but two districts declined to participate in the study. Alternate school districts were drawn from the representative quartiles to obtain the required fourteen school districts.

In order to pick the number of respondents necessary for each selected school district the universe of potential respondents was determined with the total number of students in the district as the base. These population figures were rounded to the nearest tenth. The extreme variation in number of pupils from the fourth quartile to the first quartile required that a different percentage be applied to each quartile to derive a comparable number of respondents. The percentages used

were: one per cent for the fourth quartile, two per cent for the third quartile, five per cent for the second quartile, and ten per cent for the first quartile. The application of these percentages to each quartile provided for the distribution of 3,729 instruments in the State of Michigan.

Table 11 is designed to show the number of respondents drawn from each school district sampled. The table shows the coded number for each district, the quartile a district represents, the potential number of respondents, the specific number of batteries distributed, and the percentage used for each quartile to derive the required number of respondents.

$\frac{ \mbox{Distribution and Collection}}{\mbox{of the Instrument}}$

General Procedure

Each participating district was visited by the researcher on a mutually arranged date. The researcher visited with the superintendent and building principals to explain the purposes of the study. On recommendation from the school administrators, schools were selected to represent the broadest socio-economic levels of the community. The principals selected sixth and twelfth grade teachers with whom the researcher could discuss the purposes of the study. The teachers were informed

TABLE 11. -- Total school district population and number deemed necessary for this study.

		Elem. Pu	pils	Secon. S	tudents	Par	Parents	Fatro	0110	Fact	Faculty	
Dist.	Quar- tile	Total En- # i rollment Samp	# in Sample	Total En- # in rollment Sample	# in Sample	Sample Sample	Sub- Sample	Sample	Sub- mple Cample	Total	# in Sample	% Csed to Berive Sample
r-t	7	004,4	44	007,4	य य	00+ ' a	-1 -2	ਹੁਰੁਝ ੰ ਤ	-1 -1	-1 	r.	ref
Ci	±31	9,500	(D)	COS 6	12) (0)	5,600	i tu	5, £00	• . • i .	O 30 35	€5 + 1	r=1
	Total	13,000	0. E	13,000	0.00	15,030	ind ind	૦૦૦, દ્વ	6. T	् क् च	G.	
m	65	06° .	а) Ц/	068°2	(1); (1)	2,330	30 37	058°Z	a: uh	र्फ च टा	9	Ĉij
-3*	61	3,200	<u> </u>		4 W	0001	≟ J '•	3,230	5	<u>ာ</u> င က	· 2)	Q
	£ . C 4,	0:01)	ପ ୯. ମ	ာ (၁၈)	τ. . · r t		C) As et	€€0 ' 9	C: (1	11/3 23 13/3	; r 1	
L	(C)	008°0	i.c ed	() ()	ta's ret ret	0000	17:	058 ° 8	siv me e f	() [-	-31	T.C.
ų)	C1	300	 e 1	000	ar e-i	000	÷ T	300	et et	in O	€d	ſΩ
r-	Ci	80 20	30	\$3 \$2 \$4	(2) (6)	009	30	6009	30	i D	. 1	27
QI,	СЛ	003°	5°.	00000	2. 1 2. 1 4.74	60 mg.	ii C.	្ ស្ត្រី ព	ម្នា ក	<u>(, , ,)</u>	() ()	t, v
	64 45 67 64	(6.43	a's (d.) (°a	to the same of the	14 4 3 u	0000		⊖ (** (**	(4) (4) (7)	0 11 77	.m) (5,	
c.	Н	43° 4° 4°-	O c	1 1 7	Ţ.		÷.	C 0 0	0.5	C t~	t -	O ed
() F1	e 4	5) 4	.~; 1. √	φ ⁺ () (μ [*])		<u> </u>	55		0 6	0.7		ć) eđ
r t	г	030	O G	67. 873 691	ä,	200	20	0 0 8	02	O	64	10
Cu e-f	e=4	202	() () -	ن ده ۲۰۰	Ç.	700	7.5	700	-1	ଦୁଦ	Ü) T
(* 7 p. 1	e4	750	un tra	: : U^5 (~-	1.75 1.45	F-0	ur t∽	750	32	70	r-	रूपे टच
-7	rt	700	() t	000	0.5	700	© ~	700	0	n) O	uλ	₫) =1
	Total	3,750	375	3,759	6.1 10.1 10.1	3,750	375	3,750	375	310	E	
Jrand	nd Total	28,540	912	042,85	816	28,540	615	28,540	912	2,655	31	

about the purposes, distribution, and collection of the instrument. Cooperation from teachers was good in all districts. All efforts were made to complete the study with as little disruption to class activities as possible.

Elementary Pupils

The sixth grade was chosen to represent the elementary pupils because they are concluding their elementary education. The instrument was distributed and collected by the researcher or the classroom teacher in one thirty-minute period. Ten per cent of the number distributed was identified with code numbers to be re-tested within three weeks for a check on instrument reliability. The classroom teacher administered this second testing at her convenience.

Secondary Students

The twelfth grade was chosen to represent the secondary students because of their level of maturity and long association with the secondary schools of the district. The instrument was distributed by either the researcher or the classroom teacher in one thirty-minute period. Ten per cent of the number distributed was identified with code numbers to be re-tested within three weeks for a check on instrument reliability. The class-room teacher administered this second testing at her convenience.

Parents

Parents were contacted through the pupils sampled in the sixth grade. The sixth graders were asked to take the instrument home, have a parent fill it out, and return it to their respective classrooms. The previously coded pupils, for re-testing, were instructed to take a second instrument home to be completed within three weeks as a check on instrument reliability.

Patrons

Patrons were contacted through the pupils sampled in the sixth grade. The sixth graders were asked to contact a neighbor with no children in the public schools. Patrons, after completion of the instrument, returned it to the sixth grader who took it to school for collection. The previously coded pupils, for re-testing, were instructed to take a second instrument to the same patron within a three-week period as a check on instrument reliability.

Faculty

The teachers in charge of the sampled sixth and twelfth grade classes were asked to complete the instrument along with those selected by the building principals to derive the required sample number. Ten per cent of the teachers were requested to repeat the instrument

within a three-week period as a check on instrument reliability.

A month after the first visit, the researcher returned to the schools to collect the completed instruments. A check was made to see that the appropriate number had been collected and that instructions had been followed.

Treatment of the Data

The researcher sorted the instruments into six classifications: (1) elementary pupils, (2) secondary students, (3) parents, (4) patrons, (5) faculty, and (6) first and second testing of all groups for reliability. The instruments for reliability were coded to indicate first and second testing. The instruments were then sorted by school district to be punched on IBM cards. A print-out of the tabulations was requested to facilitate checking required sample numbers and the computations necessary for statistical tests.

Statistical Methods

The data recorded are clearly of a nominal nature (favorable-unfavorable) requiring the application of non-parametric procedures for the tests of significance. The significance testing answers the question of independence for the various groups sampled. The specific statistical tests used are:

- l. The Chi Square (X^2) is used to test the significance of difference between the proportion of favorable response by item for fourth, third, second, and first support quartile districts within each battery.
- 2. The Chi Square (X²) is used to test the significance of difference between the proportion of favorable response by item of elementary pupils, secondary students, parents, patrons, and faculty within fourth, third, second, and first support quartile districts.
- 3. The estimate of item reliability is based on the consistency of the first testing to the second testing for a selected sample using a Pearson product-moment correlation method.
- 4. State norms are computed for the total sample of school districts by the application of the standard error of a percentage at the 95% confidence level.

- 1. The Chi Square (X^2) is used to test the significance of difference between the proportion of favorable response by item for fourth, third, second, and first support quartile districts within each battery.
- 2. The Chi Square (X²) is used to test the significance of difference between the proportion of favorable response by item of elementary pupils, secondary students, parents, patrons, and faculty within fourth, third, second, and first support quartile districts.
- 3. The estimate of item reliability is based on the consistency of the first testing to the second testing for a selected sample using a Pearson product-moment correlation method.
- 4. State norms are computed for the total sample of school districts by the application of the standard error of a percentage at the 95% confidence level.

CHAPTER IV

ANALYSIS OF THE HYPOTHESES

Introduction

This chapter analyses the two research hypotheses of this study. Each section presents the results of the statistical treatment in summary form, the rejection or acceptance of the hypotheses, and the interpretation of the various tests used. Section one is the analysis of the What Do You Think About Your Schools ability to discriminate by battery between fourth, third, second, and first support quartile districts. Section two is the analysis of the instrument's ability to discriminate between elementary pupils, secondary students, parents, patrons, and faculty within each cost-quartile.

The hypotheses are stated in the null form for each statistical test. The .05 level of significance is used to define the probability level. If the probability level is smaller the .05 then the data is considered controdictory to the hypotheses and a decision to reject the null hypotheses is made. Rejection of the null hypotheses means the research hypotheses are accepted.

Hypothesis I

The first general null hypothesis and five operational null hypotheses are as follows:

The What Do You Think About Your Schools instrument will not discriminate between the attitudes of elementary pupils, secondary students, parents, patrons, and faculty on the basis of financial support quartiles of Michigan school districts (K-12). Support quartile is defined in terms of size, effort, ability, and expenditure.

HIa: The instrument will not show ability to discriminate between fourth, third, second, and first financial support quartile districts on each item score according to elementary pupil's responses.

HIb: The instrument will not show ability to discriminate between fourth, third, second, and first financial support quartile districts on each item score according to secondary student's responses.

HIc: The instrument will not show ability to discriminate between fourth, third, second, and first financial support quartile districts on each item score according to parent's responses.

HId: The instrument will not show ability to discriminate between fourth, third, second, and first financial support quartile districts on each item score according to patron's responses.

HIe: The instrument will not show ability to discriminate between fourth, third, second, and first financial support quartile districts on each item score according to faculty's responses.

Statistical Tests and Treatment

The Chi Square (X^2) distribution was used to test the significance of the observed differences between the proportion of favorable response by quartile for each battery. The limits within which the hypotheses will be accepted or rejected is based on the 0.05 level of

significance (P > .05). The " χ^2 " values which cut-off five per cent of the " χ^2 " distribution provide the relative measure of difference between the proportion of favorable response. The " χ^2 " will be numerically large when (1) the null hypothesis is not true, or (2) the null hypothesis is true but the difference between the proportion can be accounted for by the errors expected on the basis of the assumptions underlying the use of this experimental design. The null hypothesis will be accepted if the " χ^2 " value does not reach the .05 level of significance. The region of rejection for the " χ^2 " varies according to the "degree of freedom" (d.f.) for the specific hypothesis tested.

Results of the Statistical Treatment

In order to determine if the <u>What Do You Think About</u>

<u>Your Schools</u> instrument would discriminate between finan
cial support quartiles the proportion of favorable

response for each item was compared by battery. The first

null hypothesis is:

HIa: The instrument will not show ability to discriminate between fourth, third, second, and first financial support quartile districts on each item score according to elementary pupil's responses.

Table 12 presents the items that are significant at P > .05 for elementary pupils. Of the 37 items responded to by elementary pupils only 17 were significant at the P > .05 level. The total number significant (17) does

.05 by battery across quartile. TABLE 12. -- Total number of items significant at P >

Item	Elementary Pupils	Secondary Students	Parents	Patrons	Faculty
Satisfaction with Schools					
Like School	+	* +	NS	;	NS
Student Feels Part of Group	NS	* +	NS	;	NS
Teacher Knows Child	+	* +	NS	;	NS
Teacher Gives Help	+	NS	NS	;	NS
Value of Studies	* +	* +	NS	;	NS
Proud of School	NS	* +	NS	NS	NS
Teacher Interest in Child	NS	* +	NS	}	NS
Schools Compare Favorably	+	;	NS	NS	!
curricaly sements Curriculum	l I	!	1	!	NS
Teaching Staff	1	i	ļ	!	NS
Building	i	1	1	!	NS
Equipment	1	!	!	!	NS
Secondary Schools					
Curriculum	!	+	!	!	+
Teaching Staff	;	*+	!	!	NS
Building	1	NS	I	!	+
Equipment	;	+	ļ	!	NS

lA "+" indicates significance, a "NS" indicates non-significance, a "+*" indicates significance with low support schools scoring higher than high support schools, and a "--" means the item was not asked in that battery.

TABLE 12. -- Continued

ols uture		Students			Faculty
School's Future	NS	* +	SN	NS	NS
į	+	* +	NS	NS	NS
Effectiveness of Studies NS	NS	NS	+	;	NS
Part in Extra-Curricular Activities	NS	NS	+	;	NS
Amount of Work to Keep Up	NS	NS	+	1	NS
Amount of Homework	+	NS	NS	;	NS
Enough Extra-Curricular Activities	NS	+	* +	;	NS
School Creates Interest in Extra-Curricular Activities	NS	* +	NS	į	+
Variety of Subjects +	+	+	+	;	+
Subject Wanted not Taught NS	NS	+	+	;	NS
Help from Librarian	+	* +	NS	;	NS
Money Needed for School +*	*+	NS	* +	NS	NS
Usefulness of Subjects	NS	NS	NS	NS	NS
Essential Services Desired					
Audio-Visual +	+	NS	NS	NS	NS

TABLE 12. -- Continued

Item	Elementary Pupils	Secondary Students	Parents	Patrons	Faculty
Physical Education	SN	* +	NS	NS	NS
Field Trips	+	NS	NS	NS	NS
Music	NS	+	+	NS	NS
Art	+	+	+	+	NS
Hot Lunch	*	* +	*	*	* +
Physical and Dental Examination	* +	NS	*	NS	NS
Job Placement	! !	NS	NS	NS	NS
Guidance	}	* +	*	NS	NS
Speech Correction	NS	NS	NS	NS	NS
Mental Health Clinic	NS	NS	NS	NS	NS
Remedial Reading	NS	+	NS	NS	NS
Special Education	NS	NS	* +	+	NS
Adult Education	;	+	NS	NS	NS
Agriculture Program	1	* +	*	*	* +
Social Activities	NS	* +	* +	NS	NS
Summer School	* +	+	NS	+	NS
Driver Training	1	* +	NS	NS	NS

TABLE 12. -- Continued

Item	Elementary Pupils	Secondary Students	Parents	Patrons	Faculty
School Organization					
School Overcrowded	N	+	+	!	NS
Supervisory Assistance	!	1	1	!	+
School Plant					
Adequate Equipment and Facilities	+	+	+	ł	NS
Community Relations					
School-Community Relations	!	!	NS	NS	* +
Faculty Status in Community	;	!	NS	NS	NS
School As Informant	1	1	NS	NS	NS
Minutes Published	1		+	+	NS
Parent-Teacher Relations	1	1	+	I I	+
Parent-Teacher Conferences	}	1	NS	i	NS
PTA Effectiveness	!	1	NS	ł	NS
Taxes and Services Rendered	!	1	NS	NS	NS
Increase Taxes	!	1	+	NS	ł
Know Child's Teacher or Parent	}	1	+	.	NS
School-Community Information	i	!	+	NS	NS
Interest of Parents in School Activities	NS	NS	NS	1	NS

not indicate the true number that support the research hypothesis since " X^2 " is not sensitive to direction. Additional analysis is needed to see if high support districts received the highest scores in terms of favorable response. This was done by observing the scores to see if they corresponded with expected cost-quartile, i.e. high support districts receiving the highest scores. Appendix B shows the total scores obtained, the " X^2 " derived, degrees of freedom, and shows whether the item is significant at P > .05. This analysis revealed that 5 of the 17 significant items are in the opposite direction anticipated by the research hypothesis. In other words, low support districts scored higher than high support districts.

The second null hypothesis is:

HIb: The instrument will not show ability to discriminate between fourth, third, second, and first financial support quartile districts on each item score according to secondary student's responses.

at P > .05 for secondary students. Of the 45 items responded to by secondary students 29 were significant. The 29 significant items include 17 items that are significant but indicate that low support districts scored higher than high support districts, thus these items do not support the research hypothesis. Appendix C shows the total scores obtained, the "X²" derived, degrees of

freedom, and shows whether the item is significant at P > .05 for secondary students.

The third null hypothesis tested is:

HIc: The instrument will not show ability to discriminate between fourth, third, second, and first financial support quartile districts on each item score according to parent's responses.

Table 12 presents the items that are significant at P > .05 for parents. The total number of items for parents is 56. Of the 56 items 22 are significant. The 22 significant items include 8 which indicate that low support districts scored higher than high support districts. Appendix D shows the total scores obtained, the " X^2 " derived, degrees of freedom, and shows whether the items are significant at P > .05 for parents.

The fourth null hypothesis tested is:

HId: The instrument will not show ability to discriminate between fourth, third, second, and first financial support quartile districts on each item score according to patron's responses.

Table 12 indicates the items that are significant at P > .05 for patrons. The total number of items for patrons is 31 of which 6 are significant. The 6 significant items include 2 which are significant but indicate that low support districts scored higher than high support districts. Appendix E shows the total scores by quartile, the "X²" derived, degrees of freedom, and shows whether the items are significant at P > .05 for patrons.

The fifth null hypothesis tested is:

HIe: The instrument will not show ability to discriminate between fourth, third, second, and first financial support quartile districts on each item score according to faculty's responses.

at P > .05 for faculty. The total number of items is 60.

9 of the 60 items are significant for faculty. The 9
significant items include 3 which are significant but
indicate that low support districts scored higher than
high support districts. These 3 items do not support
the research hypothesis that high support districts
should be more favorable in response to the items than
low support districts. Appendix F shows the total scores
obtained, the "X²" derived, degrees of freedom, and shows
whether the items are significant at P > .05 for faculty.

Implications of the Statistical Treatment

The results of the statistical treatment clearly indicates that the What Do You Think About Your Schools instrument does not discriminate by financial support quartile on the majority of items. The research hypothesis that high support districts will score higher than low support districts is significant for 32% of the elementary pupil's items, 27% of secondary student's items, 25% of parent's items, 13% of patron's items, and 10% of faculty's items. These percentages do not warrant the conclusion that the instrument will discriminate between

financial support quartiles. Elementary pupils, who have the lowest level of maturity and least amount of experience with the school situation, have the highest percentage of significant responses to the specific items. The responses of faculty have the lowest percentage of significant items. The evidence indicates that faculty members from all four quartiles have similar attitudes toward the schools. This conclusion could be accounted for by the rather homogeneous socio-economic class and professional training of faculty members. Faculty members seem to be unaware of size and financial differences between school districts.

An analysis of the six categories for the five batteries shows that the following percentages of items were significant: (1) "Satisfaction with School"--17% (2) "School Program"--28%, (3) "Essential Services Desired"--15%, (4) "School Organization and Size"--50%, (5) "School Plant"--75%, and (6) "Community Relations"--22%. The categories with the highest percentage of significant items were (1) "School Organization and Size"--50%, and (2) "School Plant"--75%. These two categories are the most concrete in nature for the five batteries and seem to point out that discrimination between financial support quartile is possible if specific items are used. The category "Essential Services Desired" has the lowest percentage of significant items indicating that

respondents at all levels of financial support view "Essential Services Desired" in much the same fashion.

Summary

- 1. The <u>What Do You Think About Your Schools</u> instrument does not discriminate between fourth, third, second, and first financial support quartile districts for the majority of items (Table 12).
- 2. The highest percentage of significant items (32%) are found in the elementary pupils's battery. The lowest percentage of significant items (10%) are found in the faculty's battery.
- 3. The categories with the highest percentage of items significant are "School Organization and Size" and "School Plant." These two categories have the smallest number of items, but are the most specific for the school situation.
- 4. The results of these significance tests indicate that elementary pupils, secondary students, parents, patrons, and faculty have similar attitudes about their schools regardless of the amount of financial support given to their schools or the size of their district.
- 5. The development of norms based on financial support quartiles would be meaningless, since there is no difference between the batteries on the basis of financial support.

Hypothesis II

The second general null hypothesis and four operational null hypotheses are as follows:

The What Do You Think About Your Schools instrument will not show ability to discriminate between the responses of elementary pupils, secondary students, parents, patrons, and faculty within fourth, third, second, and first financial support quartiles of Michigan school districts.

HIIa: The instrument will not discriminate between responses of elementary pupils, secondary students, parents, patrons, and faculty on each item score for fourth financial support quartile districts.

HIIb: The instrument will not discriminate between responses of elementary pupils, secondary students, parents, patrons, and faculty on each item score for third financial support quartile districts.

HIIc: The instrument will not discriminate between responses of elementary pupils, secondary students, parents, patrons, and faculty on each item score for second financial support quartile districts.

HIId: The instrument will not discriminate between responses of elementary pupils, secondary students, parents, patrons, and faculty on each item score for first financial support quartile districts.

Statistical Tests and Treatment

The Chi Square (X^2) distribution was used to test the significance of the observed difference between the proportion of favorable response by battery within each quartile. The limits within which the hypothesis will be accepted or rejected are based on the 0.05 level of significance (P > .05). The " X^2 " will be numerically large when (1) the null hypothesis is not true, or (2) the null hypothesis is true but the difference between

the batteries can be accounted for by errors expected on the basis of the assumptions underlying the use of this experimental design. The null hypothesis will be accepted if the value of "X²" does not reach the .05 level of significance. The region of rejection for "X²" varies depending upon the "degrees of freedom" (d.f.) for the specific hypothesis tested.

Results of the Statistical Treatment

In order to determine if the What Do You Think About Your Schools instrument would discriminate between elementary pupils, secondary students, parents, patrons, and faculty within a given financial support quartile, the proportion of favorable response was compared for each item. The first null hypothesis is:

HIIa: The instrument will not discriminate between responses of elementary pupils, secondary students, parents, patrons, and faculty on each item score for fourth financial support quartile districts.

Table 13 presents the items that are significant at P > .05 for fourth quartile districts. Of the 54 items responded to by elementary pupils, secondary students, parents, patrons, and faculty for fourth quartile districts, 33 were significant at P > .05. This can be interpreted that 33 items of the responding batteries' percentage of favorable response were significantly different from each other for fourth quartile school districts. Conversely, 21 items were not significant at

TABLE 13.--The items for each quartile across batteries significant at P > $.05.^{1}$

Item	Q ₄	Q ₃	Q ₂	Q ₁
Satisfaction with Schools				
Like School	+	+	+	+
Student Feels Part of Group	+	+	+	+
Teacher Knows Child	+	+	+	+
Teacher Gives Help	_	+	+	
Value of Studies	+	+	+	+
Proud of School	+	+	+	_
Teacher Interest in Child	+	+	+	+
Schools Compare Favorably	+	+	+	+
Satisfaction with Schools	+	+	+	+
Interest in Schools Future	+	+	+	+
	•	•	•	·
School Program				
Effectiveness of Studies Part in Extra-Curricular	+	+	+	+
Activities	+	+	+	+
Amount of Work to Keep Up	+	+	+	+
Amount of Homework	_	+	+	+
Enough Extra-Curricular				
Activities	+	+	+	+
School Creates Interest in				
Extra-Curricular Activities	+	+	+	+
Variety of Subjects	_	+	+	+
Subject Wanted not Taught	+	+	+	+
Help from Librarian	+	+		_
Money needed for School	+	+	+	+
Usefulness of Subjects	+	+	+	+
oserumess or subjects	т	T	Ŧ	T
Essential Services Desired				
Audio-Visual	-	_	-	+
Physical Education	+	+	_	_
Field Trips	+	_	+	+
Music	+	+	+	+
Art	+	_	+	+
Hot Lunch	+	_	+	+
Physical and Dental				
Examination	+	+	+	+
	•			

 $^{^{1}}A$ "+" is significant at P > .05 and a "-" is not significant at P > .05.

TABLE 13.--Continued

Item	Q ₄	^Q 3	Q ₂	Q ₁
Job Placement	_	_	+	_
Guidance	-	_	_	+
Speech Correction	+	+	+	+
Mental Health Clinic	+	+	+	+
Remedial Reading	+	+	+	+
Special Education	+	+	+	+
Adult Education	-	-	-	+
Agriculture Program	-	-	+	-
Social Activities	_		-	+
Summer School	+	+	+	+
Driver Training	_	-	+	+
School Organization and Size				
School Overcrowded	+	+	+	+
Supervisory Assistance	_	<u>-</u>	_	<u>-</u>
School Plant Adequate Equipment and				
Facilities	+	+	+	+
Community Relations				
School-Community Relations	_	_	+	+
Faculty Status in Community	-	-	-	+
School as Informant	-	-	+	+
Minutes Published	-	-	-	-
Parent-Teacher Relations	-	-	-	+
Parent-Teacher Conferences	-	-	-	-
PTA Effectiveness	_	-	+	+
Taxes and Services Rendered	-	-	-	-
Increase Taxes	-	-	-	-
Know Child's Teacher or				•
Parent	-	-	-	+
School-Community Information	+	-	-	-
Interest of Parents in School Activities	+	+	+	+

P > .05. This means on 21 items the responding batteries' percentage of favorable response were statistically the same. Further interpretation would indicate that for the 21 non-significant items the score of one battery will be approximately the same as the other batteries for that specific item. Additional statistical information can be obtained in Appendix G which shows the number of respondents for each battery, the percentage of favorable response, the "X²" obtained, the degrees of freedom, and whether the item is significant at P > .05.

The second null hypothesis tested is:

HIIb: The instrument will not discriminate between responses of elementary pupils, secondary students, parents, patrons, and faculty on each item score for third financial support quartile districts.

Table 13 indicates the items that are significant at P > .05 for third quartile school districts. The 54 items include 32 items significant at P > .05 and 22 not significant at P > .05. The 32 significant items indicate that elementary pupils, secondary students, parents, patrons, and faculty responded to these items at different levels of favorable response. 22 of the items were responded to nearly the same, therefore the different batteries scores were similar. Appendix H presents the number for each respondent group, the percnetage of favorable response, the " X^2 " obtained, the degrees of freedom, and whether the item is significant at P > .05.

The third null hypothesis tested is:

HIIc: The instrument will not discriminate between responses of elementary pupils, secondary students, parents, patrons, and faculty on each item score for second financial support quartile districts.

Table 13 presents the items that are significant at P > .05 for second quartile school districts. 39 of the 54 items are significant at P > .05. 15 items are not significantly different at P > .05, and therefore indicate that these items are responded to with similar scores for elementary pupils, secondary students, parents, patrons, and faculty. Appendix I shows the number of respondents by battery, the percentage of favorable response, the " X^2 " derived, the degrees of freedom, and the significance of the item at P > .05.

The fourth null hypothesis tested is:

HIId: The instrument will not discriminate between responses of elementary pupils, secondary students, parents, patrons, and faculty on each item score for first financial support quartile districts.

Table 13 points out the items that are significant at P > .05 for first quartile school districts. 42 of the 54 items are significant at P > .05. Only 12 items are not significant indicating a similar score for all respondent groups. Appendix J presents the number of respondents for each battery, the percnetage of favorable response, the " X^2 " derived, the degrees of freedom, and whether the item is significant at P > .05.

Implications of the Statistical Treatment

ment's ability to differentiate between responses of elementary pupils, secondary students, parents, patrons, and faculty by quartiles for the majority of items. This indicates that the five batteries measure attitudes about schools from different view-points and the total comparison across batteries is meaningful. An analysis of the four quartiles shows the following percentage of significant items: (1) fourth quartile--61%, (2) third quartile--59%, (3) second quartile--72%, and (4) first quartile--77%. It would seem that the low support districts have the greatest amount of discrimination by battery of the four quartiles tested.

The categories have the following percentage of significant items: (1) "Satisfaction with Schools"-- 93%, (2) "School Program"--91%, (3) "Essential Services Desired"--65%, (4) "School Organization and Size"--50%, (5) "School Plant"--100%, and (6) "Community Relations"-- 29%. The only category with a small number of significant items is "Community Relations." An analysis of the category "Community Relations" indicates some of the possible reasons for the low number of significant items: (1) the category is responded to by adults (parents, patrons, and faculty) except for one item; (2) high financial support districts (Q_4 and Q_3) failed to discriminate

as well as low financial support districts (Q_2 and Q_1); and (3) the number of total respondents is smaller for parents, patrons, and faculty than for elementary pupils and secondary students, thus affecting the magnitude of the " X^2 " necessary for an item to be significant at P > .05.

The category "School Organization and Size" has only two items. The item "Supervisory Assistance" is responded to by faculty only, therefore no significance test can be administered. These two categories, because of the number of items, produce a high percentage of significant items.

Summary

- 1. The <u>What Do You Think About Your Schools</u> instrument does discriminate between elementary pupils, secondary students, parents, patrons, and faculty by quartiles.
- 2. Low support school districts discriminate on more items than high support school districts.
- 3. Of the six categories only "Community Relations" has a majority of non-significant items. This would indicate that adults view the items in a similar fashion.
- 4. The continued use of elementary pupils, secondary students, parents, patrons, and faculty to measure school district attitudes seems justified. Each group supplies its own particular piece of information about the school assessment project.

CHAPTER V

RELIABILITY COEFFICIENTS AND NORMS FOR THE WHAT DO YOU THINK ABOUT YOUR SCHOOLS BATTERIES

Introduction

This chapter will present the findings for reliability and norms for the What Do You Think About Your Schools batteries. Section one considers the research question dealing with the establishment of reliability coefficients for each battery by item. The statistical tests used to determine reliability will be presented along with complete tables of each reliability coefficient by battery and item. Section two presents the norms constructed from the sample school districts. The research question proposed will be followed with the results found for the school districts sampled. The statistical tests used to determine these norms are described.

Reliability Coefficients

This section presents results to the research question concerning the reliability of each battery of the instrument, What Do You Think About Your Schools. The method of determining reliability was a test-retest of the instrument to an identified sample of the total sample.

This would indicate the variation of attitudes over time or the failure of the instrument to measure the same responses over a period of time.

The number of respondents identified for the reliability study was approximately ten per cent of the total number of questionnaires distributed in the districts.

This ten per cent was identified with code numbers for completion of the same instrument after three weeks.

Table 14 shows the number of total respondents and the number identified for the reliability study.

TABLE 14.--Number of respondents for the reliability study by battery and item.

Battery	Total Sample N.	Sub-sample for Reliability N.
Elementary Pupils	882	65
Secondary Students	869	73
Parents	739	56
Patrons	480	58
Faculty	87	33

A Pearson's product-moment coefficient was computed for each item by battery. This coefficient indicates the consistency of the instrument from test to retest. The higher the coefficient the better the item. Table 15

TABLE 15.--The reliability coefficients of the instrument, What Do You Think About Your Schools by battery for each item.

Patron		1	111
Parent	4 4 6 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7	η L.	
Faculty	* * * * * * * * * * * * * *	.80	.71
Secondary Student	20 80 80 80 80 80 80 80 80 80 80 80 80 80	. 68	64
Elementary Pupil		.50	. 49 . 70
Factor	Satisfaction with Schools Like School Student Feels Part of Group Teacher Knows Child Teacher Gives Help Value of Studies Proud of School Teacher Interest in Child Schools Compare Favorably Satisfaction with Schools Interest in Schools' Future	Effectiveness of Studies	tivities nt of Work to Keep nt of Homework

(1) curriculum, * and ** are derived by averaging correlations for: (2) teaching staff, (3) buildings, and (4) equipment.

TABLE 15.--Continued

Factor	Elementary Pupils	Secondary Student	Faculty	Parent	Patron
Enough Extra-Curricular Activities School Creates Interest in Extra-Curricular Activities Variety of Subjects Subject Wanted not Taught Help from Librarian Money Needed for School Usefulness of Subjects	.61 .76 .76 .89 .77	0	9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9		
Essential Services Desired					
Audio-Visual Physical Education Field Trips Music Art Hot Lunch Physical and Dental Examination Job Placement Guidance Speech Correction Mental Health Clinic Remedial Reading Special Education	7	6077777 70077777 7007486 1807088	004040 004000 00400 00400 00400 00400 00400 00400 00400 00400 004000 00400 00400 00400 00400 00400 00400 00400 00400 004000 00400 00400 00400 00400 00400 00400 00400 00400 00400 00400 00400 00400 00400 00400 00400 00400 00400 00400		7.7.7.47. 97.99.90 9.0.0.47. 07.99.90 9.0.0.47. 07.90 9.0.0.47. 07. 07. 07. 07. 07. 07. 07. 07. 07. 0

TABLE 15.--Continued

Factor	Elementary Pupil	Secondary	Faculty	Parent	Patron
Adult Education Agriculture Program Social Activities Summer School Driver Training			. 62 . 89 . 77 . 60	4.0.0.4 4.0.0.4	.39 .70 .75 .63
104	.67	.87	.70	.71	11
School Plant Adequate Equipment and Facilities Community Relations	99.	.71	.72	.72	1
School-Community Relations Faculty Status in Community School as Informant Minutes Published Parent-Teacher Relations Parent-Teacher Conferences PTA Effectiveness	111111	1111111		7.0.0.0.0.0.0.0.0.0.0.0.0.0.0.0.0.0.0.0	68

TABLE 15.--Continued

Factor	Elementary Pupil	Secondary	Faculty	Parent	Patron
Taxes and Services Rendered Increase Taxes	11		.56	.70	.18
Parent School-Community Information			 80.00	.45	.52
School Activities	.33	.72	.55	92.	;

presents the reliability coefficients for each battery by item.

An analysis of Table 15 shows the items vary from a high correlation of .88 for facultys' response to Teacher Gives Help to a low correlation of -.19 for facultys' response to Parent-Teacher Relations. The high amount of variation for faculty is probably associated with the small number (33) of respondents to the test-retest situation. The battery with the highest correlations is secondary students. The lowest correlation for secondary students is .50 for the item Guidance. The battery with the lowest correlations is patrons with item correlations of .25 for Satisfaction with Schools, .38 for Schools Compare Favorably, and .18 for Taxes and Services Rendered.

The category with the lowest correlations for all batteries is <u>Essential Services Desired</u>. Fourteen of the items on the five batteries are below .50. All the items in categories <u>School Organization and Size</u> and <u>School Plant</u> are .66 or above, exhibiting the best reliability among the six categories.

How high must the reliability coefficient be in order to be useful? Attitude testing has never shown the high reliabilities of more concrete measures, such as achievement tests. Parten¹ (1950) investigated the

¹Mildred Parten, <u>Surveys</u>, <u>Polls</u>, and <u>Samples</u> (New York: Harper and Brothers, 1950), p. 500.

consistency of attitudes from testing to retesting and reports that there was a 24 per cent (reliability coefficient of .50) shift in scores about family data, factual personal data, and subjective personal opinions. Most of the shifts occurred with subjective personal opinions.

A widely quoted source concerning practical reliability coefficients is Kelley (1927). Kelley arrives at the following as minimum correlations for several purposes:

- 1. To evaluate level of group accomplishment--.50.
- 2. To evaluate differences in level of group accomplishment in two or more performances--.90.
- 3. To evaluate level of individual accomplish-ment--.95.
- 4. To evaluate differences in level of individual accomplishment in two or more performances--.98.2

It must be noted that these values are arbitrary. How low a reliability one is willing to accept depends upon the practical values which are involved in a particular case. If action is necessary and the only instruments available have a reliability of .50 or lower it is better than nothing and will produce useable results.³

²Truman Kelley, <u>Interpretation of Educational Measurement</u> (New York: World Book Company, 1927).

³Robert L. Thorndike, "Reliability," in <u>Educa</u>tional Measurement, ed. by E. F. Lindquist (Washington: American Council on Education, 1951), p. 609.

Since this study is concerned with evaluation of group accomplishment, in relation to attitudes, a reliability coefficient of .50 or higher would produce useful results. Table 16 lists the battery items that are below .50. Of the total 229 items only 27 are below .50 reliability. The category with the highest number of coefficients below .50 is Essential Services Desired with 13. This would indicate that responses to specific services vary more than any other category of the five batteries. Additional study should be undertaken to revise this section of the five batteries to make the questions clearer.

The Spearman-Brown formula permits estimates of total battery reliability from an individual item. The formula for this estimate of battery reliability is:

Battery reliability =
$$\frac{(N. \text{ of items}) \text{ (reliability of item)}}{1 + (N. \text{ of items}) \text{ (reliability of item)}}$$

Applying this formula to selected item reliability coefficients indicates that the five batteries range in total estimated reliability from .84 for patrons to .95 for secondary students. Table 17 indicates the estimates of battery reliability computed from the lowest positive item by battery as the bases of the estimate. These estimates of battery reliability indicate high reliability for the five batteries of the What Do You Think About Your Schools

TABLE 16.--Item reliabilities below .50 by battery.

Item	Elementary Pupils	Secondary Students	Faculty	Parents	Patrons
Satisfaction with Schools					
Like School Teacher Gives Help	.34		.37	. 48	
Schools Compare Favorably Satisfaction with	;	;	!	!	.38
Schools	. 47	1	;	;	.25
School Program					
Part in Extra-curricular activities Variety of Subjects Money Needed for School	6 † · · ·			 47 .34	
Essential Services Desired					
Audio-Visual Physical Education Field Trips Art Job Placement Guidance Speech Correction	.36		1	. 47	

Patrons .39 .18 1 Parents .38 1 1 1 Faculty -.19 | | ! 1 | Secondary Students | 1 | | 1 l 1 Elementary Pupils .33 1 1 1 1 1 1 Remedial Reading Special Education Adult Education Interest of Parents in School Community Relations Taxes and Services Driver Training Item Parent-Teacher Activities Relations Rendered

TABLE 16.--Continued

and therefore justify its use as part of school program assessment.

TABLE 17.--Estimates of battery reliability.

Battery	Item	Item r	Estimated Battery r
Elementary Pupils	Interest of Parents in School Activities	.33	.92
Secondary Students	Guidance	.50	•95
Parents	Speech Correction	.32	.94
Patrons	Taxes and Services Rendered	.18	.84
Faculty*	Remedial Reading	.30	.94

^{*.30} is the lowest positive correlation, although one item, <u>Parent-Teacher Relations</u> is -.19.

Norms for the Five Batteries of the What Do You Think About Your Schools

The results of Chapter IV indicated that there is no difference in degree of favorable response to the items by cost quartile. Therefore norms for the instrument are constructed on the total sample of school districts. This section describes the norms established for the five batteries by item. These norms are constructed by computing the standard error of a percentage (σ p.c.). The scores for the respective batteries are reported in percentage

of favorable response. The question that presents itself is—to what degree can the sample percentage of favorable response be indicative of the "true score" for the uni—verse of potential respondents? Or to put it in terms of an example; is the favorable response to the item "Like School" correct within 3, 5, or 10 per cent? The smaller the standard error, the greater the precision of the estimate.

The formula for calculating the standard error of a percentage is:

$$\sigma \text{ p.c.} = \sqrt{\text{p.c.} \frac{(100-\text{p.c.})}{n}} \left(\frac{n-1}{n-2}\right)^{4}$$

The multiplier $\frac{n-1}{n-2}$ is dropped from the formula because of the large size of the sample. It would not effect the final error term enough to warrant its computation; therefore the formula used is:

σ p.c.
$$\sqrt{\frac{p.c. (100-p.c.)}{n}}$$

To illustrate how this formula will work, the percentage of favorable response for "Like School" is 85.4%.

$$\sigma$$
 p.c. = $\sqrt{\frac{85.4 (100-85.4)}{882}}$ = 1.18

⁴Mildred Parten, op. cit., p. 500.

Thus we may conclude that there are two chances out of three that the estimate of 85.4% from our sample is within \pm 1.18% of the correct value. If we multiply the σ p.c. by 2 we can conclude that the value of the standard error is correct 19 times out of 20 for a probability of .95.

The main assumption of this technique is that the sample estimate represents the population percentage, which is the most probable assumption in random sampling; but the sample estimate can be considerably off without greatly effecting the standard error.

A table can be constructed for the sample to indicate the range of scores by favorable response for each item at the .95 probability level. This will assure users of the instrument that the score derived should be within the stated range 19 times out of 20.

Table 18 presents the standard error range for the sample districts at the .95 probability level rounded to the nearest whole per cent.

Summary

This chapter dealt with the establishment of reliability coefficients for the five batteries by item and the norming of the five batteries on the randomly drawn sample of Michigan school districts. The following conclusions were reached:

1. The batteries are reliable from testing to testing, indicating that attitudes of elementary pupils,

TABLE 18.--Sample districts attitudes about the schools expressed in a range of percentage of favorable response at the .95 probability level.

Factor	Elementary Pupil	Secondary Student	Faculty	Parent	Patron
Satisfaction with Schools					
Like School Student Feels Part of Group Teacher Knows Child Teacher Gives Help Value of Studies Proud of School Teacher Interest in Child Schools Compare Favorably Satisfaction with Schools Interest in Schools' Future	88 88 88 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	71-77 59-65 74-80 89-93 61-67 70-76 71-77	* 000 010 010 010 010 010 010 01	911-95 881-95 881-87 881-87 881-87 881-89 871-89	 90-94 81-87 81-87 93-97
School Program					
Effectiveness of Studies	95-36	80-86	91-95	95-96	!
rare in Exera-curricular Activities	5-6	8-7	4-8	5-8	1
Amount of Work to Keep Up Amount of Homework	59-65 44-50	52-58 53-59	78-84 66-72	78-84 59-67	
Enough Extra-Curricular Activities	52-58	67-73	76-82	85-89	;
School Creates Interest in Extra-Curricular Activities Variety of Subjects Subject Wanted not Taught	57-63 84-88 46-52	65-71 52-58 64-71	69-75 65-71 63-69	73-79 78-84 32-40	

TABLE 18.--Continued

Factor	Elementary Pupil	Secondary Student	Faculty	Parent	Patron
Help from Librarian Money Needed for School Usefulness of Subjects	67-73 61-67 95-97	54-60 89-93 82-86	57-63 66-72 91-95	54-62 72-78 90-94	67-75 81-87
Essential Services Desired					
	6-18 6-18	2-8 6-9	100	110	7-8
⊣ • ~	763-169 63-169 76-180	71-77 71-77	100 100 96–98	880 880 1 - 1 880 1 - 1	82-88 82-88 73-81
7	7-8	7-8	4-8	1 <u>-</u> 8	3-8
flysical and Dental Examination Job Placement	65-71	50 - 56 76-82	66-72 72-78	68-74 63-69	73-81 65-73
Guidance Speech Correction	7-8	3-9	1007	2-9	7-9
~	68-74 78-84	3-7-8	5-8 7-9	0-8	5-8 8-9
्घ =	3-7	6-9	- 1 1 1 1 1 1 1 1 1	5-7	2-8 2-8
cure Prog		7-7	2-4-2	4-7	5-7-0
> 0	53-59 53-59	5-19	6 I 1 8 1 8	1-0	4-0
Driver Training	!	4-9	8–9	9-6	8–9

TABLE 18. -- Continued

Factor	Elementary Pupil	Secondary Student	Faculty	Parent	Patron
School Organization and Size School Overcrowded Supervisory Assistance School Plant	25-31	54-60	72 - 78 71 - 77	39-47	1 1
Adequate Equipment and Facilities	33-39	23-29	34-40	50-58	1
School-Community Relations Faculty Status in Community School as Informant Minutes Published Parent-Teacher Relations Parent-Teacher Conferences PTA Effectiveness Taxes and Services Rendered Increase Taxes Know Child's Teacher or Parent School-Community Information Interest of Parents in School Activities	 91-95	80-86	59-65 461-75 76-75 78-82 78-82 78-82 78-82 78-85 78-85 78-85 86-135 78-135	63-69 75-81 70-68 71-74 76-90 78-82	47-57 63-71 43-53 74-82 73-81 50-60 33-41

* and ** are derived by combining the norms for (1) curriculum, (2) teaching staff, (3) buildings, and (4) equipment.

secondary students, parents, patrons, and faculty are generally the same over a short period of time.

- 2. The category "Essential Services Desired" exhibits the weakest reliability and should be revised to increase the precision of responses to the various services presented. This is especially true for the parents', patrons', and facultys' batteries.
- 3. The reliability coefficients derived indicate that 27 of the 229 items must be interpreted with care. These coefficients indicate that the error present for these items require some caution in analyses. Estimates of battery reliability are high enough to indicate that the over-all pattern of scores are meaningful for school program assessment.
- 4. The norms established are derived from the total sample of Michigan school districts. This permits users of the What Do You Think About Your Schools to compare school district's item scores to the state-wide norms established for the purpose of school program assessment.

CHAPTER VI

SUMMARY, CONCLUSIONS, IMPLICATIONS, AND RECOMMENDATIONS

Summary

Purposes of the Study

The purposes of this study are three-fold: (1) to determine the ability of the What Do You Think About Your Schools to discriminate between the responses of elementary pupils, secondary students, parents, patrons, and faculty within and between cost quartiles; (2) to establish the instrument's reliability; and (3) to establish Michigan norms for the What Do You Think About Your Schools.

Two major hypotheses were formulated to test the discriminating ability of the instrument within the four financial support quartiles and between the five batteries by each financial support quartile. The two major hypotheses are:

1. The What Do You Think About Your Schools instrument will discriminate between the attitudes of elementary pupils, secondary students, parents, patrons, and faculty on the basis of financial support quartiles of Michigan

school districts (K-12). Support quartile is defined in terms of size, effort, ability, and expenditure.

2. The What Do You Think About Your Schools instrument will show ability to discriminate between the responses of elementary pupils, secondary students, parents, patrons, and faculty within fourth, third, second, and first financial support quartiles of Michigan school districts (K-12).

The need to improve the precision of the instrument required that two additional studies be done: (1) a study of item reliabilities, and (2) the establishment of norms for the total sample of school districts. These two studies established the stability of the instrument from test to test and provided a norm population against which users of the instrument could make comparisons.

Sample and Design

The sample was selected on the basis of a stratified random sample of Michigan school districts (K-12).

The sample included two districts in the fourth quartile,
two districts in the third quartile, four districts in
the second quartile, and six districts in the first quartile. A one per cent sample of fourth quartile, a two
per cent sample of third quartile, a five per cent sample
of second quartile, and a ten per cent sample of first
quartile were taken to determine the number of respondents for the district. Useable data were collected from

3,057 respondents representing 882 elementary pupils, 869 secondary students, 739 parents, 480 patrons, and 87 faculty members. Ten per cent of each battery was identified with code numbers for re-testing within three weeks as a check on instrument reliability.

Instrumentation and Data Collection

Data for this study came from two sources: (1) the Michigan Education Association's Ranking of Michigan High School Districts by Selected Financial Data for 1966-67, and (2) the responses to the five batteries of the What Do You Think About Your Schools.

The financial data on size, effort, ability, and expenditure was organized by quartiles and a composite quartile was derived rom the four financial characteristics. This procedure equated the districts within each quartile.

The measurement of attitudes was obtained through responses of elementary pupils, secondary students, parents, patrons, and faculty to the instrument, What Do You Think About Your Schools. This instrument is based on the assumption that attitudes of the school community can be measured with precision. Each battery varies in the number of items asked, but nineteen items are responded to by all groups. The total number of items for each battery is: (1) elementary pupils--37, (2) secondary students--45, (3) parents--53, (4) patrons--31, and

(5) faculty--60. Each battery consists of six categories, each dealing with some aspect of the school program:

(1) satisfaction with schools, (2) school program, (3) essential services desired, (4) school organization and size, (5) school plant, and (6) community relations. The respondent indicates his choice to the item by checking a four-point value scale. The scores are dichotomized to indicate percentage of favorable and unfavorable response for each item.

Each district in the sample was visited by the researcher. The instruments were distributed with the cooperation of administrators and teachers to sixth graders, twelfth graders, parents of sixth graders, patrons who were neighbors of sixth graders, and teacher. An analysis of the returns for each battery included: (1) elementary pupils--96.7%, (2) secondary students--95.3%, (3) parents--81%, (4) patrons--52.6%, and (5) faculty--100%.

Method of Treatment and Analysis

The testing of the two hypotheses required the use of a statistical test for non-parametric data, since the responses were reported in percentage of favorable response. The following tests of significance were used:

1. The Chi Square (X^2) was used to test the significance of difference between the percentage of favorable

response by item for fourth, third, second, and first support quartile districts within each battery.

2. The Chi Square (X²) was used to test the significance of difference between the percentage of favorable response by item for elementary pupils, secondary students, parents, patrons, and faculty within fourth, third, second, and first support quartile districts.

Three other statistical procedures were undertaken to increase the precision of the instrument.

- 1. Item reliability was determined for the consistency of the first testing to the second testing for a ten per cent sample of the total sample using a Pearson product-moment correlation method.
- 2. Estimates of battery reliability were computed using a Spearman-Brown prophecy formula for estimating total battery reliability.
- 3. State norms were computed for the total sample of school districts by the application of the standard error of a percentage at the 95% confidence level.

Scope and Delimitations of the Study

The parameters of this study were delimited by the following factors:

l. The major variables in this study were the individual's attitudes, as measured by the What Do You Think About Your Schools and the cost factors of size,

effort, ability, and expenditure as derived from the Michigan Education Association's, Ranking of Michigan High School Districts by Selected Financial Data for 1966-67.

- 2. The study was limited to a sample of fourth, third, second, and first quartile school districts in the State of Michigan. No results were drawn from individual school districts.
- 3. This study used only selected financial cost factors and did not include all possible permutations of cost analyses.
- 4. The findings of a relationship between school-community attitudes and financial factors were viewed as associational and not causal.
- 5. The study assumed that the individual sampled would respond to the instrument with his true perceptions in regards to the school-community situation.

Major Findings

The major findings of this study were:

1. The What Do You Think About Your Schools instrument does not discriminate between fourth, third, second, and first financial support quartile districts on the majority of items. The percentage of significant items for each battery was: (1) elementary pupils--32%, (2) secondary students--27%, (3) parents--25%, (4) patrons--13%, and (5) faculty--10% (Table 12). The categories with the

highest percentage of significant items were "School Organication and Size" with 50% and "School Plant" with 75%.

These two categories have the smallest number of items, two and one respectively, but are the most specific for the school situation (Table 12).

- 2. The <u>What Do You Think About Your Schools</u> instrument did discriminate between elementary pupils, secondary students, parents, patrons, and faculty within each financial support quartile. An analyses of the four quartiles indicates the following percentage of significant items:

 (1) fourth quartile--61%, (2) third quartile--59%, (3) second quartile--72%, and (4) first quartile--77%. Low support districts had more significant items than high support school districts (Table 13). The categories had the following percentages of significant items: (1) satisfaction with schools--93%, (2) school program--91%, (3) essential services desired--65%, (4) school organization and size--50%, (5) school plant--100%, and (6) community relations--29% (Table 13).
- 3. The <u>What Do You Think About Your Schools</u> instrument produced item reliability coefficients ranging from a high of .88 for faculty response to "Teacher Gives Help" to a low correlation of -.19 for faculty response to "Parent-Teacher Relations" (Table 15). Of the total 229 items for the five batteries, only 27 were below .50 (Table 16). The category that had the highest number of

low reliability coefficients (<.50) was "Essential Services Desired" with 13 (Table 16).

4. The What Do You Think About Your Schools instrument produced useable norms for the State of Michigan at a 95% confidence level. The standard error of a percentage was applied to each item providing a range of percentages to which users of the instrument could make comparisons (Table 18).

Conclusions

This study provides several conclusions regarding the relationship between community attitudes and the factors of size, effort, ability, and expenditure. It also provides conclusions regarding the item reliability of each battery and norms for a sample of Michigan school districts.

The conclusions derived are:

1. There is no difference in the attitudes of the five groups sampled according to financial support quartiles. High support district scores are similar to low support district scores. This would indicate that attitudes at all levels of financial support were not dependent on size, effort, ability, and expenditure. The instrument measures a distinct variable other than financial support, which supplies useful information to a curriculum assessment model.

- 2. The items that did reflect high scores for high quartile districts were those most directly related to financial expenditure, i.e. "School Overcrowded," "Supervisory Assistance," "Adequate Equipment and Facilities," and "Variety of Subjects" (Table 12).
- 3. Items dealing with interpersonal relationships and communications within the districts (category—"Community Relations") showed that high quartile districts did not score higher than low support districts. This would seem to indicate that large relatively wealthy districts are not doing a better job of relating to their constituents and providing information than are small poorer districts (Table 12).
- 4. Norms for this instrument based upon financial support quartiles would be meaningless, since no significant differences were observed in the responses of the five populations used. Norms were constructed, instead, upon the total sample which combined the scores for all four financial support quartiles.
- 5. There was a difference between the scores of elementary pupils, secondary students, parents, patrons, and faculty on the majority of items within each cost quartile. This meant that each group did have different attitudes toward their schools and that each group contributes useful information about each item. The continued use of all five batteries contributes useful

information for a total assessment of a school district's educational program (Table 13).

- 6. The only category in which the respondents did not differ was "Community Relations." This category is responded to by parents, patrons, and faculty on eleven of the twelve items. It would seem that adults' attitudes in regards to "Community Relations" were similar.
- The batteries were reliable from testing to testing, indicating that attitudes of elementary pupils, secondary students, parents, patrons, and faculty were generally consistent over a short period of time. The category "Essential Services Desired" exhibited the weakest reliability (Table 15). This was especially true for faculty, parents, and patrons. It seems that adults' attitudes vary more than do students. The estimated reliability for each battery was: (1) elementary pupils, .92; (2) secondary students, .95; (3) parents, .94; (4) patrons, .84; and (5) faculty, .94 (Table 17). These are respectable reliability coefficients for an instrument which measures attitudes. The analyses of item reliability indicate that future scores for 27 of the 229 items below R.50 must be interpreted with care. The error present in these items require some caution in interpretation. One of the reasons for low reliability coefficients for faculty can be attributed to the small sample. With only 33 respondents there is a wide variation

in the theoretical "true reliability" and the "observed reliability" of the faculty battery. The remaining four batteries had a larger sample upon which the reliability coefficient was derived, and therefore the "true reliability" is closer to the "observed reliability."

8. The norms established for the instrument were based on the total sample of 14 school districts in the State of Michigan. The total number of respondents was large, therefore the range of scores at a 95% confidence level were small. The largest variation around the sample score was plus or minus five percentage points (Table 18). These norms are useful for comparing a given district's scores with the norms of the sample population.

Implications

This study has several implications for school program assessment:

1. Attitudes of elementary pupils, secondary students, parents, patrons, and faculty are not related to financial support quartiles. Attitudes are dependent on factors other than size, effort, ability, and expenditure. This finding strengthens the curriculum assessment model used in this study, since attitudes give additional information concerning the school program not obtained from analyses of size, effort, ability, or expenditure.

- 2. Each battery measures attitudes from a different view-point. Elementary pupils, secondary students,
 parents, patrons, and faculty score differently for the
 majority of items, which indicates that no one segment of
 the community can be presumed to speak for another community group. This finding adds another important dimension to the curriculum assessment model used in this
 study, i.e. that all members of the school community have
 different attitudes toward the schools and each battery
 adds its segment of information to the total assessment
 model.
- 3. Item scores are not as reliable as total scores. The interpretation of individual items must be made with caution. An individual using this instrument can place considerable confidence in a total interpretation of its results, but specific items must be interpreted with care.
- 4. The norms for the What Do You Think About Your Schools are established on 3,057 cases. This is not a large sample for the development of norms, but the districts represent several important variables: (1) dize, (2) effort, (3) ability, (4) expenditure, (5) geographic distribution of the State of Michigan, (6) socio-economic distribution, and (7) population density. A user of the instrument can be fairly sure that all types of public school districts are included in the norming sample.

Recommendations

Further investigations are needed to improve the curriculum assessment model presented in this study and to improve the precision of attitude measurement for elementary pupils, secondary students, parents, patrons, and faculty.

- You Think About Your Schools be continued in the assessment of school programs. The findings of this study indicate that the instrument gives additional information about a school district that cannot be obtained in any other manner. The reliability coefficients for the five batteries are high enough to place confidence in the scores obtained and show the relative stability of attitudes over a short period of time.
- 2. It is recommended that further study be undertaken for the improvement of the norms developed. The criteria for selection of school districts should be size, socio-economic distribution, geographic dispersion, and type of population center. The present study seems to include these variables, but a study designed to control these specific variables might prove more useful than a random sample based on size, effort, ability, and expenditure.
- 3. It is recommended that this study be replicated with a national sample of school districts. The results

of a national study would increase the application of results to a much larger population.

- 4. It is recommended that a better method of selecting parents and patrons be established to provide a better representation of these groups. The method used in this study, contact by sixth grade pupils, leaves some questions about the true responses from the total population of parents and patrons.
- 5. It is recommended that further study of instrument reliability be done. This study should sample a larger number of respondents for each battery to decrease the amount of variation present within a small sample. This would give more precise limits for the scores obtained from each respondent group.
- 6. It is recommended that the category "Essential Services Desired" be reviewed. The scores obtained do not represent adequate reliability and show non-significant differences between batteries. The researcher feels that the possible responses to the items in the category "Essential Services Desired" do not sample the true attitudes from the various groups. Rather than "Very Essential," "Essential," "Not Essential," and "Waste of Time;" a better classification might be "Very Essential," "Essential," "Essential," "Needed," and "Not Needed." This would tend to spread the scores over a greater range.

- 7. It is recommended that the possible responses to items "Amount of Work to Keep Up," "Amount of Home-work," and "Money Needed for School" be revised to include two favorable responses instead of the present one to be consistent with the rest of the items. The present possible responses limit the choice of the respondents and increase computational difficulties.
- 8. It is recommended that further study be conducted to find the relationship between the What Do You Think About Your Schools and a measure of school quality, such as the Educational Characteristics Criterion. Are the two instruments measuring the same thing? If so, modifications would be necessary in the curriculum assessment model developed by Rudman.
- 9. It is recommended that further study be done transcribing the nominal data to interval data. This would permit a more sophisticated analyses of the differences between batteries and the financial support quartiles. The use of non-parametric statistics does not permit as sensitive an analyses of the data as a parametric technique. Much of the information is lost when combining scores into favorable and unfavorable responses.

BIBLIOGRAPHY

BIBLIOGRAPHY

- American Association of School Administrators and National School Boards Association. <u>Judging Schools With</u> Wisdom. Washington: The Association, 1959.
- An Analysis of Educational Facilities: Present and

 Future Needs. Goodrich, Michigan: Goodrich Board
 of Education, 1967. (Mimeographed.)
- Ayres, Leonard P. An Index Number for State School
 Systems. New York: Russell Sage Foundation, 1920.
- Bagley, W. C. <u>Determinism in Education</u>. Baltimore: Warnick and York, 1925.
- Berg, Arthur D. "The Determination of the Discrimination and Reliability Indices of the Educational Characteristics Criterion With Implications Concerning Educational Cost-Quality Relationships." Unpublished Ph.D. dissertation, Michigan State University, 1962.
- Bothwell, Bruce K. <u>Creative Expenditure for Quality Education</u>. New York: Associated Public School System, 1958.
- Buros, Oscar Kristen, ed. <u>The Sixth Mental Measurement</u> <u>Yearbook</u>. Highland Park, New Jersey: The Gryphon Press, 1966.
- Dewey, John. The School and Society. Chicago: The University of Chicago Press, 1900.
- Dale, Edgar and Jeanne Chall. "A Formula for Predicting Readability," Educational Research Bulletin. Vol. 27 (February, 1948), 37-54.
- Educational Policies Commission. An Essay on Quality in Public Education. Washington: National Education Association, 1959.
- Ferrell, Doctor Thomas. Relation Between Current Expenditure and Certain Measures of Educational Efficiency in Kentucky County and Graded School Systems.

 Richmond, Kentucky: Eastern State Teachers College, 1937.

- Fisk, Robert S. <u>Public Understanding of What Good Schools</u>
 Can Do. New York: Bureau of Publications, Teachers
 College, Columbia University, 1944.
- Fowlkes, John Guy, and Watson, George E. School Finance and Local Planning. Chicago: The Midwest Administrative Center, 1957.
- Furno, Orlando F. "The Projection of School Quality from Expenditure Level." Unpublished Ed.D. project, Teachers College, Columbia University, 1956.
- Gardner, John W. "A National Assessment of Educational Progress." Unpublished report, Carnegie Corporation, April 23, 1965.
- Grace, A. G., and Moe, G. A. State Aid and School Costs:

 Report of the Regent's Inquiry. New York: McGrawHill Book Company, 1938.
- Griffis, James T. Educational Production at Three Cost Levels. Houston: Gulf School Research Development Association, 1955.
- Hand, Harold C. What People Think About Their Schools. New York: World Book Company, 1948.
- The Camel's Nose," Phi Delta Kappan. Vol. 47 (September, 1965), 9-12.
- Hirsch, Werner Z. Analysis of Rising Costs of Public Education. Washington: Joint Economic Committee, 1959.
- Iowa Tests of Basic Skills. Boston: Houghton Mifflin
 Company, 1956.
- <u>Iowa Tests of Educational Development</u>. Chicago: Science Research Associates, 1958.
- Kelley, Truman, et al. Stanford Achievement Test, Technical Supplement. New York: Harcourt, Brace, and World Incorporated, 1966.
- _____. Interpretation of Educational Measurements. New York: World Book Company, 1927.
- Kraft, Leonard E. "The Perceptions Held by Professors of Education, Professors in Areas Other than Education and School Board Members on Ninety Factors Which May or May Not Affect Quality of an Educational Program." Unpublished Ed.D. dissertation, Michigan State University, 1962.

- Kreitlow, Burton W. School District Reorganization . . .

 Does it Make A Difference in Your Child's Education?

 Madison: Agriculture Experiment Station, University of Wisconsin, 1961.
- Lazarus, Ralph. We Can Have Better Schools. New York: Committee for Economic Development, 1959.
- Love, Bryon K. "The Development of a Suggested Instructional Programs Cost Model for K-12 Districts of the State of Michigan." Unpublished Ed.D. dissertation, Michigan State University, 1968.
- McLure, William P. Let Us Pay For The Kind of Education

 We Need: Report of a Study of State and Local

 Support of Mississippi's Schools. Oxford: University of Mississippi, Bureau of Educational Research,

 1948.
- McNeil, John D. <u>Curriculum Administration</u>. New York: Macmillian Company, 1965.
- Metropolitan School Study Council. A Poll of Opinion About School. New York: The Council, 1955.
- Michigan Education Association. Rankings of Michigan High School Districts by Selected Financial Data for 1966-67. East Lansing, Michigan: Michigan Education Association, 1967.
- Montrose in Transition. East Lansing, Michigan: Educational Service Series No. 21, Educational Publication Services, Michigan State University, 1967.
- Mort, Paul R. (director). The Financing of the Public Schools of Maine. Augusta: Maine School Finance Commission, 1934.
- in the State of New Jersey, Vol. II: Report of the Governor's Survey Commission. Trenton: The Commission, 1933.
- ______. "Cost-Quality Relationships in Education,"

 Problems and Issues in Public School Finance.

 Edited by R. L. Johns and E. L. Morphet. New York:

 National Conference of Professors of Education,

 Teachers College, Columbia University, 1952.

- Mort, Paul R.; Burke, Arvid J.; and Fisk, Robert S.

 A Guide for the Analysis and Description of Public School Services. New York: Institute of Educational Research, Teachers College, Columbia University, 1942.
- , and Cornell, Francis G. American Schools in Transition: How Our Schools Adapt Their Practices to Changing Needs. New York: Teachers College, Columbia University, 1941.
- _____, and Cornell, Francis G. <u>Guide for Self-Appraisal</u>
 of School Systems. New York: Bureau of Publications, Teachers College, Columbia University, 1937.
- ; Cornell, Francis G.; and Hinton, Norman. What
 Should Our Schools Do?: A Poll of Public Opinion
 on the School Program. New York: Bureau of Publications, Teachers College, Columbia University,
 1938.
- _____, and Furno, Orlando F. Theory and Synthesis of a Sequential Simplex. New York: Institute of Administrative Research, Teachers College, Columbia University, 1960.
- ; Reusser, Walter C.; and Polley, John W. <u>Public</u>
 School Finance. New York: McGraw-Hill Book Company, 1960.
- ; Vincent, William S.; and Newell, Clarence A. The Growing Edge. New York: Metropolitan School Study Council, Teachers College, Columbia University, 1946.
- Mueller, Van Dyck. "A Study of the Relationships Between Teachers-Administrators Perceptions of Educational Quality as Measured by the Educational Characteristics Criterion, (ECC) and Selected Cost Factors."

 Unpublished Ed.D. dissertation, Michigan State University, 1964.
- National Education Association. How Good Are Your Schools? Washington: The Association, 1958.
- National School Boards Association. <u>Yardsticks for</u>
 Public Schools. Evanston, Illinois: The Association, 1959.
- National Study of Secondary School Evaluation. <u>Evaluative</u> Criteria. Washington: The Study, 1960.

- Norton, John K. The Ability of States to Support Education. Washington: National Education Association, 1926.
- Parten, Mildred. Surveys, Polls, and Samples. New York: Harper and Brothers, 1950.
- Pelton, Maurice D. "A National Analysis of Educational Quality as Measured by the Educational Characteristics Criterion (ECC), Achievement, and Selected Cost Factors." Unpublished Ed.D. dissertation, Michigan State University, 1966.
- Planning for Educational Renewal, Hamtramck. Educational Service Series No. 19. East Lansing, Michigan: Educational Publication Services, Michigan State University, 1966.
- Rickover, Hyman G. "Education for all Chilcren," Hearings Before the Committee on Appropriations, House of Representatives Eighty-Seventh Congress. Washington: The Committee, 1962.
- Ross, Donald, ed. <u>Administration for Adaptability</u>. New York: Institute of Administrative Research, Teachers College, Columbia University, 1958.
- Rudman, Herbert C. "The Curriculum," Unpublished report,
 Michigan State University, East Lansing (February,
 1968).
- _____, and Hecker, Stanley E. "The Determination and Measurement of Factors Which Directly or Indirectly Affect Quality of an Educational Program." Unpublished proposal, Michigan State University, 1961.
- School Management. "The Cost of Education Index." Vol. 9, No. 1 (January, 1965), 100-151.
- Smith, B. Othanel; Stanley, William O.; and Shores, J. Harlan. Fundamentals of Curriculum Development. New York: The World Book Company, 1950.
- South Haven Citizens Study Their Schools. South Haven, Michigan: South Haven Board of Education, 1967. (Mimeographed.)
- Southern Association of Secondary Schools. Evaluating the Elementary School: A Guide for Cooperative Study. Atlanta: Commission on Research and Service, The Association, 1951.

- Springer, Owen. "A Study of the Relationship Between the Educational Characteristics Criterion (ECC), the Stanford Achievement Test, and Selected Cost Factors." Unpublished Ed.D. Dissertation, Michigan State University, 1964.
- Stake, Robert S. "Testing in the Evaluation of Curriculum Development," Review of Educational Research. Vol. 38, No. 1 (February, 1968), 77-84.
- Stoddard, George. "The Issues That Divide Us," School and Society. Vol. 86 (May 24, 1958), 236-237.
- Strayer, George D. (director). A Report of a Survey of Public Education in the State of West Virginia. Charleston, West Virginia: Legislative Interim Committee, State of West Virginia, 1945.
- Thorndike, Edward L. Education as Cause and as Sympton. New York: The Macmillian Company, 1939.
- by E. F. Lindquist. Washington: American Council on Education, 1951.
- Turck, Merton J., Jr. "A Study of the Relationship Among the Factors of Financial Need, Effort, and Ability in 581 High School Districts in Michigan." Unpublished Ed.D. dissertation, Michigan State University, 1962.
- U. S. Department of Health, Education and Welfare. Education for Freedom and World Understanding. Bulletin OE-10016. Washington: U. S. Government Printing Office, 1962.
- University of Michigan. <u>Criteria for Accreditation</u>. Ann Arbor: Bureau of School Services, University of Michigan, 1961.
- Vincent, William S. Emerging Patterns of Public School Practice. New York: Bureau of Publications, Teachers College, Columbia University, 1945.
- . "Quality Control: A Rationale for Analysis of a School System," <u>I.A.R. Research Bulletin</u>. Vol. 1, No. 2 (January, 1961), 1-7.
- policy for Public Education in New York State-Public Tests of School Quality. New York: New York Educational Conference Board, 1960.

- Walling, W. Donald. A Study of Public Opinion About Schools. New York: Metropolitan School Study Council, Teachers College, Columbia University, 1952.
- Wollatt, Lorne, H. The Cost-Quality Relationships on the Growing Edge. New York: Teachers College, Columbia University, 1949.

APPENDICES

APPENDIX A

WHAT DO YOU THINK ABOUT YOUR SCHOOLS

Elementary Pupils

Secondary Students

Parents

Patrons

Faculty

WHAT DO YOU THINK ABOUT YOUR SCHOOLS?

(c) Herbert C. Rudman, 1967

This questionnaire represents one of several methods being used to analyze your school situation. Faculty members, parents of school children, other adults, high school students, and grade school students are being asked to complete similar questionnaires designed to measure the reaction of the community to its schools.

YOUR MALE ANAMERS ON THIS FORM.

Please check the statement that comes closest to answering each question. After completing this form, please return it to a proper collection point.

Thank you so much for taking the time to complete this questionnaire. We need to know how you feel about your schools so that they can be made even better than they are today.

1.	School District	
2.	Check one:	
	Elementary Student	1
	Secondary Student	2.
	Parent of Student Enrolled in Public School	3
	Patron (No Chil- dren Enrolled in Public School)	4
	Faculty Member	5

Very well	1.	How well do you like school?	
Quite well Very little Not at all 2. Do you feel that you are "one of the group" in your school? Yes Usually Sometimes No 3. How well do you think that your teacher knows you? Very well Somewhat Little Not at all 4. Do you feel that your teacher is interested in you as a person? Yes Somewhat Little Not at all 4. Do you feel that your teacher is willing to help you when you have a problem? Yes Sometimes Little No 4. Do you feel that your teacher is willing to help you when you have a problem? Yes Sometimes Little No 4. Do you feel that your school? Yery Some Little No 4. Do you feel that your school? Very Some Little No 2		Very well	1.
Very little 3.		-	2.
Not at all 4. 2. Do you feel that you are "one of the group" in your school? Yes 1			3.
2. Do you feel that you are "one of the group" in your school? Yes			4.
Usually Sometimes 3. Sometimes 3. Sometimes 3. Sometimes 3. Somewhat 2. Somewhat 3. Somewhat 2. Somewhat 3. Somewh	2.	Do you feel that you are "on	ne of the group" in your school?
Usually Sometimes 3. Sometimes 3. Sometimes 3. Sometimes 3. Somewhat 2. Somewhat 3. Somewhat 2. Somewhat 3. Somewh		Yes	1.
Sometimes 3. 4. 3. 3. 4. 3. 3. 3. 3. 4. 3. 3. 3. 4. 3. 3. 3. 3. 4. 3. 3. 3. 3. 3. 3. 3. 3. 3. 3. 3. 3. 3.		Usually	2.
No 4 3. How well do you think that your teacher knows you? Very well 1 Somewhat 2 Little 3 Not at all 4 4. Do you feel that your teacher is interested in you as a person? Yes 1 Somewhat 2 Little 3 Little 3 Not at all 4 5. Do you feel that your teacher is willing to help you when you have a problem? Yes 1 Sometimes 2 Little 3 No 4 6. How proud are you of your school? Very 1 Some 2 Little 3 Little 3 None 4 7. Do you feel that your school is as good as some other schools you know about? Very good 1 Good 2 Fair 3 Very good 1 Good 2 Fair 3 7. Do you feel that your school is as good as some other schools you know about?			3.
Very well so you think that your teacher knows you? Very well 1 Somewhat 2 Little 3 Not at all 4 4. Do you feel that your teacher is interested in you as a person? Yes 1 Somewhat 2 Little 3 Little 3 Not at all 4 5. Do you feel that your teacher is willing to help you when you have a problem? Yes 1 Sometimes 2 Little 3 Little 3 No 4 6. How proud are you of your school? Very Some 2 Little 3 Little 3 None 4 7. Do you feel that your school is as good as some other schools you know about? Very good 1 Good 2 Fair 3 Very good 1 Good 2 Fair 3 7 Very good 1 Good 2 Fair 3 Yes Some 1 Some 2 Some 2 Some 2 Some 2 Some 3 Some 3 Some 4 Some 4 Some 4 Some 5 Some 6 Some 6 Some 7 Some 8 Some 9 Some 9.			
Very well 1.			
Somewhat Little Not at all 4. Do you feel that your teacher is interested in you as a person? Yes Somewhat Little Not at all 5. Do you feel that your teacher is willing to help you when you have a problem? Yes Little Sometimes Little No 4. 6. How proud are you of your school? Very Some Little None 7. Do you feel that your school is as good as some other schools you know about? Very good Fair 1. Some other schools 2. Little Some Little	3.	How well do you think that y	our teacher knows you?
Somewhat 2. Little 3		Very well	1.
Little Not at all 4. Do you feel that your teacher is interested in you as a person? Yes Somewhat Little Not at all 5. Do you feel that your teacher is willing to help you when you have a problem? Yes Little Sometimes Little No 4. 6. How proud are you of your school? Very Some Little None 7. Do you feel that your school is as good as some other schools you know about? Very good Fair 1. 3. 4. 7. Do you feel that your school is as good as some other schools you know about?			2.
Not at all 4. Do you feel that your teacher is interested in you as a person? Yes Somewhat Little Not at all 5. Do you feel that your teacher is willing to help you when you have a problem? Yes Sometimes Little No 4. 6. How proud are you of your school? Very Some Little None 7. Do you feel that your school is as good as some other schools you know about? Very good Good Fair 1. 3. 4. 7. Do you feel that your school is as good as some other schools you know about?		Little	3.
Yes Somewhat Little Not at all 5. Do you feel that your teacher is willing to help you when you have a problem? Yes Sometimes Little No 4. 6. How proud are you of your school? Very Some Little None 7. Do you feel that your school is as good as some other schools you know about? Very good Good Fair 1. Some other schools Some o		Not at all	4.
Yes Somewhat Little Not at all 5. Do you feel that your teacher is willing to help you when you have a problem? Yes Sometimes Little No 4. 6. How proud are you of your school? Very Some Little None 7. Do you feel that your school is as good as some other schools you know about? Very good Good Fair 1. Some other schools Some o			
Somewhat Little Not at all Not at all 4. 5. Do you feel that your teacher is willing to help you when you have a problem? Yes Sometimes Little No 4. 6. How proud are you of your school? Very Some Little No 4. 7. Do you feel that your school is as good as some other schools you know about? Very good Fair 1. 3. 4. 7. Do you feel that your school is as good as some other schools you know about?	4.	Do you feel that your teache	r is interested in you as a person?
Somewhat Little Not at all Not at all Little Not at all Little Sometimes Little No Little No Little No Little No Little No Little Some Little None Lit		Yes	1
Not at all Not at all 4. 5. Do you feel that your teacher is willing to help you when you have a problem? Yes Sometimes 2. Little No 4. 6. How proud are you of your school? Very Some Little None 1. Some 2. Little 3. Little 3. Little 3. Little 3. Some 4. 7. Do you feel that your school is as good as some other schools you know about? Very good Good Fair 3. 3. 3. 3. 3. 3. 3. 3. 3. 3		Somewhat	2.
Not at all 4. 5. Do you feel that your teacher is willing to help you when you have a problem? Yes Sometimes Little No 4. 6. How proud are you of your school? Very Some Little None 1. 2. 1. 3. Little 3. Little 3. Little 3. Little 3. Little None 4. 7. Do you feel that your school is as good as some other schools you know about? Very good Good Fair 1. 3. 3. 3. 3.		Little	3.
have a problem? Yes Sometimes Little No 4. 6. How proud are you of your school? Very Some Little None 1. 2. 2. 3. 4. 7. Do you feel that your school is as good as some other schools you know about? Very good Good Fair 3. 3. 3.		Not at all	4.
have a problem? Yes Sometimes Little No 4. 6. How proud are you of your school? Very Some Little None 1. 2. 2. 3. 4. 7. Do you feel that your school is as good as some other schools you know about? Very good Good Fair 3. 3. 3.	5	Do you fool that your torobo	m is redlight to holo you when you
Sometimes Little No 4. 6. How proud are you of your school? Very Some Little None 7. Do you feel that your school is as good as some other schools you know about? Very good Good Fair 3. 1. 2. 3. 4. 7. 1. 5. 6. How proud are you of your school? 1. 5. 6. How proud are you of your school? 1. 6. How proud are you of your school? 1. 5. 6. How proud are you of your school? 1. 6. How proud are you of your school? 1. 6. How proud are you of your school? 1. 5. 6. How proud are you of your school? 1. 6. How proud are you of your school? 1. 6. How proud are you of your school? 1. 6. How proud are you of your school? 1. 6. How proud are you of your school? 1. 6. How proud are you of your school? 1. 6. How proud are you of your school? 1. 6. How proud are you of your school? 1. 6. How proud are you of your school? 1. 6. How proud are you of your school? 1. 6. How proud are you of your school? 1. 6. How proud are you of your school? 2. 5. 6. How proud are you of your school? 2. 6. How proud are you of your school? 2. 6. How proud are you of your school? 2. 6. How proud are you of your school? 3. 6. How proud are you of your school? 2. 6. How proud are you of your school? 3. 6. How proud are you of your school? 2. 6. How proud are you of your school? 2. 6. How proud are you of your school? 2. 6. How proud are you of your school? 2. 6. How proud are you of your school? 2. 6. How proud are you of your school? 2. 6. How proud are you of your school? 8. 8. 8. 8. 8. 8. 8. 8. 8. 8	J.		I IS WITTING to help you when you
Sometimes Little No 4. 6. How proud are you of your school? Very Some Little None 7. Do you feel that your school is as good as some other schools you know about? Very good Good Fair 3. 1. 2. 3. 4. 7. 1. 5. 6. How proud are you of your school? 1. 5. 6. How proud are you of your school? 1. 6. How proud are you of your school? 1. 5. 6. How proud are you of your school? 1. 6. How proud are you of your school? 1. 6. How proud are you of your school? 1. 5. 6. How proud are you of your school? 1. 6. How proud are you of your school? 1. 6. How proud are you of your school? 1. 6. How proud are you of your school? 1. 6. How proud are you of your school? 1. 6. How proud are you of your school? 1. 6. How proud are you of your school? 1. 6. How proud are you of your school? 1. 6. How proud are you of your school? 1. 6. How proud are you of your school? 1. 6. How proud are you of your school? 1. 6. How proud are you of your school? 2. 5. 6. How proud are you of your school? 2. 6. How proud are you of your school? 2. 6. How proud are you of your school? 2. 6. How proud are you of your school? 3. 6. How proud are you of your school? 2. 6. How proud are you of your school? 3. 6. How proud are you of your school? 2. 6. How proud are you of your school? 2. 6. How proud are you of your school? 2. 6. How proud are you of your school? 2. 6. How proud are you of your school? 2. 6. How proud are you of your school? 2. 6. How proud are you of your school? 8. 8. 8. 8. 8. 8. 8. 8. 8. 8		Yes	1.
Little No 4. 6. How proud are you of your school? Very Some Little None 7. Do you feel that your school is as good as some other schools you know about? Very good Good Fair 3. 1. 2. 3. 4. 7. Do you feel that your school is as good as some other schools you know about?			2
No 4 6. How proud are you of your school? Very 1 Some 2 Little 3 None 4 7. Do you feel that your school is as good as some other schools you know about? Very good 1 Good 2 Fair 3			3.
Very Some Little None Very school? 1 Some Little None 7. Do you feel that your school is as good as some other schools you know about? Very good Good Fair 3 Some 2 1 3 3			4.
Very Some 2 Little 3 None 4 7. Do you feel that your school is as good as some other schools you know about? Very good 1 Good 2 Fair 3			
Some Little None 7. Do you feel that your school is as good as some other schools you know about? Very good Good Fair 3. 1. 2. 3. 3. 3. 3. 3. 3. 3. 3. 3	6.	How proud are you of your sch	001?
Some Little None 7. Do you feel that your school is as good as some other schools you know about? Very good Good Fair 3. 1. 2. 3. 3. 3. 3. 3. 3. 3. 3. 3		Very	1.
Little 3 None 4 7. Do you feel that your school is as good as some other schools you know about? Very good 1 Good 2 Fair 3			2.
None 4 7. Do you feel that your school is as good as some other schools you know about? Very good Good Fair 3			3.
7. Do you feel that your school is as good as some other schools you know about? Very good Good Fair 3.			4.
you know about? Very good Good Fair 3.		• • • • • • • • • • • • • • • • • • • •	· · · · · · · · · · · · · · · · · · ·
you know about? Very good Good Fair 3.	7.	Do you feel that your school	is as good as some other schools
Good 2. Fair 3.			•
Good 2. Fair 3.		Very good	1.
Fair 3		• -	2.
			3.
			4.

8.	How satisfied are you with y	your school?
	Very satisfied	1.
	Satisfied	2
	Dissatisfied	3
	Very dissatisfied	4
9.	How interested are you in you	our school?
	Very interested	1
	Interested	2
	Not interested	3
	Not at all interested	4
10.	How much of what you are stube valuable to you?	dying in school do you think will
	Practically everything	1
	Most	2
	Half	3
	Very little	4
11.	How much do you think you ar	re learning from your studies?
	A great deal	1
	Something	2
	A little	3
	Very little	4
12.	Is there a chance for you to parties, plays, games, and o	eattend as many of the school clubs as you would like to?
	Al ways	1
	Most of the time	2
	Once in a while	3
	Never	4
13.	How much work do you have to studies?	o do to "keep up" in your school
	Too much	1
	About right	2
	Not very much	3
	None at all	4.

14.	How much homework do you have	re?
	None at all	1
	Little	2
	About right	3
	More than I can do	4
15.	Does your school offer as ma school activities as you wou	ny plays, clubs and other after- ild like to see offered?
	All that are needed	1
	Most that are needed	2
	Few that are needed	3
	None that are needed	4.
16.	Does your school try to get and other after-school activ	you interested in plays, clubs, ities?
	A great effort is made	1
	An effort is made	2
	Little effort is made	3
	No effort is made	4
17.	Are you satisfied with the vin school?	eristy of subjects that are taught
	Very satisfied	1
	Satisfied	2
	Unsatisfied	3
	Very unsatisfied	4
18.	Are there subjects that you offered in your school?	would like to take that are not
	Many	1
	Some	2
	Few	3
	There are enough now	4
19.	If you have a school library do your school work?	, how helpful is it to you as you
	Very helpful	1
	Helpful	2
	Not very helpful	3
	No libraryin my school	4.

	_		_	_	
20.	Do you need to spend of plays, clubs, or parti				
	Too much money is	needed 1.			
	About right	2.	***************************************		
	I could pay more	3.	******		
	There are no char	rges 4.			
21.	How useful do you this school?	nk $arepsilon$ re the s	ubjects yo	u ere now	teking in
	Very useful	1.	eth-militain-subjects		
	Useful				
	Not useful	3.			
	No value at all	4 .	-		
	Check the activities a must have if it is to			you:	school Waste
		Must Have	Need	Need_	
22.	Films, film strips, T.V., records, tape recorders	1	2	3	4
23.	Physical education	1	2		4
24.	Field trips	1	2	3	4
25.	Music	1	2	3	4
26.	Art	1	2	3	4.
27.	Hot lunch	1	2	3	4
28.	Physical and dental exams	1	2	3,	4
29.	Spaech correction	1	2	3	4
30.	Mental health clinic	1	2	3	4
31.	Remedial reading	1	2	3	4
32.	Special education	1	2	3	4

1. ___ 2. __ 3. __ 4. __

1. ___ 2. ___ 4. ___

33. Social activities

34. Summer school

35.	Do you think that your s	school is overcrowded?
	Very crowded	1.
	Crowded	2
	Enough room	3
	Wasted space	4
36.	Does your school have as ment as you would like i	much playground and gymnasium equip- t to have?
	All we need	1
	Enough	2.
	Could use more	3.
	Lacking	4.
37.	How much interest do you related activities?	r parents show in your school work and
	A great deal	1
	Some	2
	Very little	3
	None	4

APPENDICES

APPENDIX A

WHAT DO YOU THINK ABOUT YOUR SCHOOLS

Elementary Pupils

Secondary Students

Parents

Patrons

Faculty

(c) Herbert C. Rudman, 1967

This questionnaire represents one of several methods being used to analyze your school situation. Faculty members, parents of school children, other adults, high school students, and grade school students are being asked to complete similar questionnaires designed to measure the reaction of the community to its schools.

YOUR MAIE ANAMERE ON THIS FORM.

Please check the statement that comes closest to answering each question. After completing this form, please return it to a proper collection point.

1.	School District	
2.	Check one:	
	Elementary Student	1
	Secondary Student	2
	Parent of Student Enrolled in Public School	3
	Patron (No Chil- dren Enrolled in Public School)	4.
	Faculty Hember	5

1.	How well do you like s	chool?
	Very well	1.
	Quite well	2.
	Very little	3.
	Not at all	4.
2.	Do you feel that you a	re "one of the group" in your school?
	Yes	1
	Usually	2.
	Sometimes	3.
	No	4
3.	How well do you think	that your teacher knows you?
	Very well	1
	Somewhat	2.
	Little	3
	Not at all	4
4.	Do you feel that your	teacher is interested in you as a person?
	Yes	1
	Somewhat	1.
		2
	Little Not at all	3 4
		•
5.	Do you feel that your have a problem?	teacher is willing to help you when you
	Yes	1
	Sometimes	2
	Little	3.
	No	4.
6.	How proud are you of yo	ur school?
	Very	1.
	Some	2.
	Little	3.
	None	4.
7.	De you fool that your	school is as good as some other schools
'•	you know about?	school is as good as some other schools
	Very good	1.
	Good	2.
	Fair	3.
	Poor	4.

8.	How satisfied are you with y	our school?
	Very satisfied	1
	Satisfied	2
	Dissatisfied	3
	Very dissatisfied	4.
9.	How interested are you in you	ur school?
	Very interested	1
	Interested	2
	Not interested	3
	Not at all interested	4.
10.	How much of what you are stube valuable to you?	dying in school do you think will
	Practically everything	1
	Most	2
	Half	3
	Very little	4.
11.	How much do you think you ar	e learning from your studies?
	A great deal	1
	Something	2
	A little	3
	Very little	4
12.	Is there a chance for you to parties, plays, games, and c	attend as many of the school lubs as you would like to?
	Always	1
	Most of the time	2
	Once in a while	3
	Never	4
13.	How much work do you have to studies?	do to "keep up" in your school
	Too much	1
	About right	2
	Not very much	3
	None at all	4.

14.	How much homework do you have	7 e?
	None at all	1
	Little	2.
	About right	3
	More than I can do	4.
15.	Does your school offer as ma school activities as you wou	ny plays, clubs and other after- ild like to see offered?
	All that are needed	1
	Most that are needed	2
	Few that are needed	3
	None that are needed	4.
16.	Does your school try to get and other after-school activ	you interested in plays, clubs, ities?
	A great effort is made	1
	An effort is made	2
	Little effort is made	3
	No effort is made	4
17.	Are you satisfied with the vin school?	eristy of subjects that are taught
	Very satisfied	1
	Satisfied	2
	Unsatisfied	3
	Very unsatisfied	4
18.	Are there subjects that you offered in your school?	would like to take that are not
	Many	1
	Some	2
	Few	3
	There are enough now	4.
19.	If you have a school library do your school work?	, how helpful is it to you as you
	Very helpful	1
	Helpful	2
	Not very helpful	3
	No libraryin my school	

2. ___ 4. ___

20.	Do you need to spend of plays, clubs, or parti	much money in Les that you	n order to r school m	take part ay sponsor	in the
	Too much money is	s needed l.			
	About right	2.			
	I could pay more	3.			
	There are no char	rges 4.			
21.	How useful do you this school?	nk $arepsilon$ re the su	ıbjects yo	u ere now	taking in
	Very useful	1.			
	Useful	2.			
	Not useful	3.			
	No value at all	4.			
	Check the activities a must have if it is to		chool for	you:	Waste
		Must have	Need	Necd	OI IIMC
22.	Films, film strips, T.V., records, tape recorders	1.	2	3	4
23.	Physical education	1	2	3	4
24.	Field trips	1	2	3	4
25.	Music	1	2	3	4
26.	Art	1	2	3	4
27.	Hot lunch	1.	2	3	4
28.	Physical and dental exams	1	2	3,	4
29.	Speech correction	1	2	3	4
30.	Mental health clinic	1	2	3	4
31.	Remedial reading	1	2	3	4
32.	Special education	1	2	3	4
33.	Social activities	1	2	3	4

34. Summer school

35.	Do you think that your s	school is overcrowded?
	Very crowded Crowded	1 2
	Enough room	3
	Wasted space	4
36.	Does your school have as ment as you would like i	much playground and gymnasium equip-
	All we need	1
	Enough	2
	Could use more	3
	Lacking	4
37.	How much interest do you related activities?	r parents show in your school work and
	A great deal	1
	Some	2
	Very little	3
	None	4

(c) Herbert C. Rudman, 1967

This questionnaire represents one of several methods being used to analyze your school situation. Faculty members, parents of school children, other adults, high school students, and grade school students are being asked to complete similar questionnaires designed to measure the reaction of the community to its schools.

YOUR ANSWERS WILL BE COMPLETELY ANONYMOUS. DO NOT WRITE YOUR NAME ANYWHERE ON THIS FORM.

Please check the statement that comes closest to answering each question. After completing this form, please return it to a proper collection point.

1.	School District	
2.	Check one:	
	Elementary Student	1
	Secondary Student	2
	Parent of Student Enrolled in Public School	3
	Patron (No Chil- dren Enrolled in Public School)	4
	Faculty Member	5

1.	How well do you like sch	0001?
	Very well	1
	Quite well	2
	Very little	3
	Not at all	4.
2.	Do you feel that you are	"one of the group" in your school?
	Yes	1
	Usually	2
	Sometimes	3
	No	4
3.	Generally, how well do y	ou think your teachers know you?
	Very well	1
	Somewhat	2
	Little	3
	Not at all	4
4.	Generally, do you feel t you as a person?	hat your teachers are interested in
	Yes	1
	Somewhat	2
	Little	3
	Not at all	4.
5.	Generally, do you feel t you when you have a prob	hat your teachers are willing to help lem?
	Yes	1
	Sometimes	2
	Little	3
	No	4
6.	How proud are you of you	r school?
	Very	1
	Some	2
	Little	3
	None	4

Do you feel that your school compares favorably with other schools that you know about in:

		Very <u>Favorably</u>	Favorably	Slightly Favorably	Unfavorably
7.	Curriculum	1	2	3	4
8.	Teaching staff	1	2	3	4
9.	Building	1	2	3	4
10.	Equipment	1	2	3	4
11.	How satisfied	are you with	n your school	L ?	
	Very sat	isfied	1		
	Satisfie	d	2		
	Dissatis	fied	3		
	Very dis	satisfied	4		
12.	How intereste	d are you in	the future of	of your school	01?
	Very int	erested	1		
	Interest	ed	2		
	Disinter	ested	3		
	Very dis	interested	4		
13.	How much of w be valuable t		studying in s	school do you	ı think will
	Practica	11y everythin	ng 1		
	Most		2		
	Half		3		
	Very lit	tle	4		
14.	How much do y	ou think you	are learning	g from your	studies?
	A great	deal	1		
	Somethin	3	2		
	A little		3		
	Very lit	tle	4		

15.	Is there a chance for you to parties, plays, games and cl	attend as many of the school ubs as you would like to?
	Always	1
	Most of the time	2
	Once in a while	3
	Never	4
16.	How much work do you have to studies?	do to "keep up" in your school
	Too much	1
	About right	2
	Not very much	3
	None at all	4
17.	How much homework do you hav	ve?
	None at all	1
	Little	2
	About right	3
	More than I can do	4
18.	Does your school offer as mayou would like to see offere	ny extra-curricular activities as
	All that are needed	1
	Most that are needed	2
	Few that are needed	3
	None that are needed	4
19.	Does the school create enougactivities?	h interest in extra-curricular
	A great effort is made	1
	An effort is made	2
	Little effort is made	
	No effort is made	4
20.	Does your school offer a wid to take?	le enough choice of courses for you
	Wide choice	1
	Enough choice	2
	Little choice	3
	No choice	4

-.

en.

21.	Are there subjects the offered by your school		ould like t	to take that a	re not
	Many		l		
	Some		2		
	Few		3		
	They now have en	ough 4	4		
22.	How much help do you school library?	get from	the librar	ian when you	go to the
	All I need		l		
	Most of the help				
	Some of the help	I need 3	3		
	I never use the library		4		
23.	How do you feel about curricular activities and clubs?				
	Too much money		l		
	About right		2		
	Very little		3		
	We don't have to	pay 4	4		
24.	How much of what you to you?	are study	ying do you	think will b	e of use
	Most		1		
	About half		2		
	Less than half	•	3		
	Very little	ı	4		
	Check the activities to a good school curr		ices that y	ou feel are <u>e</u>	ssential
		Very Es- sential	<u>Essenti</u>	Not Es- al <u>sential</u>	
25.	Audio-visual aids	1.	2	3	4
26.	Physical education	1	2	3	4
27.	Field trips	1	2	3	4
	Music	1.	2.	3	4.
_∪.	1.70TC	- •			• •

		Very Es- sential	<u>Essential</u>	Not Es- sential	Waste of Time
29.	Art	1	2	3	4
30.	Hot lunch	1	2	3	4
31.	Physical and dental exams	1	2	3	4
32.	Job placement	1	2	3	4
33.	Guidance	1	2	3	4
34.	Speech correction	1	2	3	4
35.	Mental health services	1.	2	3	4
36.	Remedial reading	1	2	3	4
37.	Special education	1	2	3	4
38.	Adult education	1	2	3	4
39.	Agriculture pro- grams	1	2	3	4
40.	Social activities	1.	2	3	4
41.	Summer school	1.	2	3	4
42.	Driver training	1	2	3	4
43.	Is your school overcr	cowded?			
	Very crowded Crowded Enough room Excess space	2. 3.			
44.	Does your school have classroom equipment t				ry, and
	All we need Adequate Could use more Lacking	2. 3.			

45.	How much	interest	do you	r parents	show	in	your	school	work	and
		activities		_			•			

A great deal 1. _____

Somewhat 2. ____

Very little 3. ____

None 4. ____

(c) Herbert C. Rudman, 1967

This questionnaire represents one of several methods being used to analyze your school situation. Faculty members, parents of school children, other adults, high school students, and grade school students are being asked to complete similar questionnaires designed to measure the reaction of the community to its schools.

YOUR ANSWERS WILL BE COMPLETELY ANONYMOUS. DO NOT WRITE YOUR NAME ANYWHERE ON THIS FORM.

Please check the statement that comes closest to answering each question. After completing this form, please return it to a proper collection point.

1.	School District	
2.	Check one:	
	Elementary Student	1
	Secondary Student	2
	Parent of Student Enrolled in Public School	3
	Patron (No Chil- dren Enrolled in Public School)	4
	Faculty Member	5

	· ·		
		•	·

	NOTE: Please indicate attending.	the grade your child or children are
1.	Elementary (Kindergarte	en to 6th grade)
2.	Secondary (7th to 12th	grade)
3.	Both Elementary and Sec	condary
<i>t</i> .	The coll do one which	
4.	•	our children like school?
	Very well	1
	Quite well	2
	Very little	3
	Not at all	4
5.	Do you feel that your one of the group"?	child is accepted by his classmates as
	Yes	1
	Usually	2
	Sometimes	3
	No	4.
6.	Do you feel that your o	child's teachers really know your child?
	Very well	1
	Somewhat	2
	Little	3
	Not at all	4
7.	To what extent do teach child?	ers show a personal interest in your
	Much	1
	Somewhat	2
	Little	3
	Not at all	4
8.	To what extent are teachas a problem?	thers willing to help your child when he
	Much	1
	Somewhat	2
	Little	3
	Not at all	4

9.	How much pride do you have i	n your district's system of schools?
	Very much	1
	Some	2
	Little	3
	None	4
10.	Do you feel that the schools with other schools that you	<pre>in your district compare favorably know about?</pre>
	Very favorably	1
	Favorably	2
	Slightly favorably	3
	Unfavorably	4
11.	In general how satisfied are attends?	you with the school your child
	Very satisfied	1
	Satisfied	2
	Dissatisfied	3
	Very dissatisfied	4
12.	How interested are you in the district?	e future of the schools in your
	Very interested	1
	Interested	2
	Disinterested	3
	Very disinterested	4
13.	How much of what your child to him?	is learning in school is of value
	Practically everything	1
	Most	2
	Half	3
	Very little	4
14.	How much is your child getti	ng from his or her studies?
	A great deal	1
	Something	2
	A little	3
	Very little	4

17,		s and clubs as he or she would like
	Always	1.
	Most of the time	2
	Occasionally	3
	Never	4
16.		mount of work assigned to your to "keep up" with their classwork?
	Too much	1
	About right	2
	Not very much work	3
	Too little	4
17.	How much homework does your school?	child get assigned to him by the
	None at all	1.
	Little	2
	Appropriate for class work	3
	A great deal	4
18.	Does the school offer as manactivities as you would like	ny plays, games, proms, and other eto see offered?
	All that are necessary	1
	Most that are necessary	2
	Few that are necessary	3
	None that are necessary	4
19.	Does the school do enough to school activities?	get your child interested in after-
	All that is necessary	1.
	Most that is necessary	2
	Little that is necessary	3
	None that is necessary	4

20.	Does the school offer your courses for him to take?	child a wide enough variety of
	Wide variety Enough variety Little variety No variety	1 2 3 4
21.	Are there courses that you ware not now being offered to	would like your child to take that them?
	Many Some Few They have enough now	1 2 3 4
22.	How much help does your chil or she goes to the school li	ld get from the librarian when he brary?
	All they need Most of the help they need Some of the help they need They have no librarian	1 2 3 4.
23.	How do you feel about the mo	oney your children spend for such textbook fees, ball games, and
	Too much money About right Children could pay more They pay no money	1 2 3 4
24.	How much of what your child use to him after he leaves s	is studying in school will be of school?
	Most About half Less than half Very little	1 2 3 4

25.	Do you feel that the	school your	r child atter	nds is over	ccrowded?
	Very crowded	1.			
	Crowded	2.			
	Enough room	3.			
	Excess space	4.			
26.	Does your child's sch and laboratory equip	nool have alment that it	ll of the pla t needs to do	ayground, o o an adequa	classroom ate job?
	All they need	1.			
	Adequate	2.	-		
	Could use more	3.			
	Lacking	4.			
27.	Do you feel that the schools and the commu	re is a good unity?	i relationshi	ip between	the
	Very good	1.			
	Good	2.			
	Could be improve	ed 3.			
	Poor	4.			
	Check the activities to a good school curr		es that you	feel are <u>e</u>	ssential
		Very Es- sential	<u>Essential</u>	Not Es- sential	Waste of Time
28.	Audio-visual aids	1	2	3	4
29.	Physical education	1.	2	3	4
30.	Field trips	1	2	3	4
31.	Music	1	2	3	4
32.	Art	1	2	3	4
33.	Hot lunch program	1	2	3	4
34.	Physical and dental exams	1	2	3	4
35.	Job placement	1	2	3	4
36.	Guidance	1.	2	3	4
37.	Speech correction	1	2	3	4
38.	Mental health	1	2	3	/ 1

		Very Es- sential	<u>Essential</u>	Not Es- sential	
39.	Remedial reading	1	2	3	4
40.	Adult education	1.	2	3	4
41.	Special education	1.	2	3	4
42.	Agriculture programs	1	2	3	4
43.	Social activities	1	2	3	4
44.	Summer school	1	2	3	4
45.	Driver training	1.	2	3	4
46.	What is your estimate community?	of the so	cial status o	of teachers	in your
	Very good	1.			
	Good	2.			
	Could be improve	d 3.	************		
	Very low status	4.			
47.	Do you feel that the about the school and			nunity adeq	uately
	Excellent	1.			
	Good	2.			
	Fair	3.			
	Poor	4.			
48.	Do you feel that the of their meetings in			lish the m	inutes
	Always	1.			
	Sometimes	2.			
	Occasionally	3.	ter teritoria de dell'esse		
	Never	4.			
49.	To what degree do you satisfactory?	feel paren	nt-teacher re	lationship	s are
	Very good	1.	-		
	Good	2.			
	Poor	3.			
	Very poor	4.			

50.	To what degree do you feel desirable?	l parent-teacher conferences are
	Very desirable	1
	Desirable	2
	Some help	3
	Waste of time	4.
51.	To what degree do you feel	PTA's are effective?
	Very effective	1
	Effective	2
	Slightly effective	3
	Waste of time	4
52.	Do you feel that the school the level of services rend	ol tax rate compares favorably with dered by the schools?
	Very favorable	1
	Favorable	2
	Unfavorable	3
	Way out of line	4
53.	Would you be willing to pational program in your com	ny more taxes for an improved educa-
	A good deal more	1
	Slightly more	2
	No increase	3
	Want a cut	4
54.	Do you feel that there is parents, administration, a	adequate communication between and school board?
	Excellent	1
	Good	2
	Fair	3
	Poor	4
55.	Do you know your child's t	teacher as well as you would like to?
	Very well	1
	Well	2
	Little	3
	Not at all	4

56.	Do you	feel	that	you	show	much	interest	in	the	school	work	and
	related											

A great deal	1
Somewhat	2
Very little	3
None	4.

(c) Herbert C. Rudman, 1967

This questionnaire represents one of several methods being used to analyze your school situation. Faculty members, parents of school children, other adults, high school students, and grade school students are being asked to complete similar questionnaires designed to measure the reaction of the community to its schools.

YOUR ANSWERS WILL BE COMPLETELY ANONYMOUS. DO NOT WRITE YOUR NAME ANYWHERE ON THIS FORM.

Please check the statement that comes closest to answering each question. After completing this form, please return it to a proper collection point.

1.	School District	
2.	Check one:	
	Elementary Student	1
	Secondary Student	2
	Parent of Student Enrolled in Public School	3
	Patron (No Chil- dren Enrolled in Public School)	4
	Faculty Member	5

1.	How much pride do you have schools?	in your district's system of
	Very much	1
	Some	2.
	Little	3
	None	4
2.	Do you feel that the school ably with other schools the	ls in your district compare favor- at you know about?
	Very favorably	1
	Favorably	2
	Slightly favorably	3
	Unfavorably	4
3.	In general, how satisfied district?	are you with the schools in your
	Very satisfied	1
	Satisfied	2
	Dissatisfi ed	3
	Very dissatisfied	4
4.	How interested are you in your district?	the future of the public schools in
	Very interested	1
	Interested	2
	Disinterested	3
	Very disinterested	4
5.	Should children pay for su clubs, proms, laboratory f	ch school activities as ball games, ees and textbooks?
	All of the cost	1.
	Some of the cost	2
	Token payment	3
	No charge	4

6.	As far as you can tell from your contact with the students
	that have been educated in your school district, how much of
	what these students learn in school is of use to them after
	they leave school?

Most	1.
About half	2
Less than half	3
Very little	4

Check the activities and services that you feel are $\underline{\text{essential}}$ to a good school curriculum:

		Very Es- sential	<u>Essantial</u>	Not Es- sentiel	Waste of Time
		Sential	Lasencial	Benciel	OI IIIIG
7.	Audio-visual aids	1.	2	3	4
8.	Physical education	1	2	3	4
9.	Field trips	1	2	3	4
10.	Music	1	2	3	4
11.	Art	1	2	3	4
12.	Hot lunch programs	1	2	3	4
13.	Physical and dental				
•	exams	1	2	3	4
14.	Job placement	1	2	3	4
15.	Guidance	1	2	3	4
16.	Speech correction	1	2	3	4
17.	Mental health clinic	1	2	3	4
13.	Remedial reading	1	2	3	4
19.	Special education	1	2	3	4
20.	Adult education	1	2	3	4
21.	Agriculture pro- grams	1	2	3	4
22.	Social activities	1	2	3	4
23.	Summer school	1	2	3	4
24.	Driver training	1	2	3	4

25.	Do you feel that there is schools and the community?	a good relationship between the
	Very good	1.
	Good	2
	Could be improved	3
	Poor	4.
0.0		
20,	your community?	he social status of teachers in
	Very 300d	1
	Good	2
	Could be improved	3
	Very low	4.
27.	Do you feel that the schoo about the school and the s	ls inform the community adequately chool program?
	Excellent	1
	Good	2
	Fair	3
	Poor	4.
28.	Do you feel that the schoo of their meetings in the 1	l board should publish the minutes ocal papers?
	Always	1
	Sometimes	2
	Occasionally	3
	Never	4.
29.	Do you feel that the schoo the level of services rend	1 tax rate compares favorably with ered by the school?
	Very favorably	1
	Favorably	2
	Unfavorably	3
	Way out of line	4
30.	Would you be willing to pa programs in your community	y more taxes for improved educational?
	A good deal more	1
	Slightly more	2
	No increase	3
	Want a cut	4

31.	Do you fee!	l that	there is	adequate	communi	catio	on betw	een the
	community,	the scl	hool admi	inistratio	n, and	the s	school	board?

Excellent	1.
Good	2
Fair	3
Poor	Δ.

(c) Herbert C. Rudman, 1967

This questionnaire represents one of several methods being used to analyze your school situation. Faculty members, parents of school children, other adults, high school students, and grade school students are being asked to complete similar questionnaires designed to measure the reaction of the community to its schools.

YOUR ANSWERS WILL BE COMPLETELY ANONYMOUS. DO NOT WRITE YOUR NAME ANYWHERE ON THIS FORM.

Please check the statement that comes closest to answering each question. After completing this form, please return it to a proper collection point.

1.	School District	
2.	Check one:	
	Elementary Student	1
	Secondary Student	2
	Parent of Student Enrolled in Public School	3
	Patron (No Chil- dren Enrolled in Public School)	4
	Faculty Member	5.

1.	How well do you think	that your students like school?
	Very well	1
	Quite well	2
	Very little	3
	Not at all	4
2.	Do you feel that, in a each other as "one of	general, the students in your class accept the group"?
	Yes	1
	Usually	2
	Sometimes	3
	No	4
3.	In general, how well o	do you think you know your students?
	Very well	1
	Somewhat	2
	Little	3
	Not at all	4
4.	In general, to what exeach of your students?	ktent do you show a personal interest in
	Much	1
	Somewhat	2
	Little	3
	Not at all	4
5.	In general, to what exthey have problems?	ktent are you willing to help students when
	Much	1
	Somewhat	2
	Little	3
	Not at all	4
6.	How much pride do you	have in your school system?
	Very much	1
	Some	2
	Little	3
	None	4

.

The State of the S

Do y	ou feel	that	the	eleme	entary	school:	s in	your	district	com-
pare	favora	bly w	ith	other	school	that	you	know	about?	

		Very <u>Favorably</u>	<u>Favorably</u>	Slightly <u>Favorably</u>	<u>Unfavorably</u>
7.	Curriculum	1	2	3	4
8.	Teaching staff	1	2	3	4
9.	Building	1	2	3	4
10.	Equipment	1	2	3	4
	D			•	

Do you feel that the secondary schools in your district compare favorably with other schools that you know about?

		Very <u>Favorably</u>	Favorably	Slightly Favorably	Unfavorably
11.	Curriculum	1	2	3	4
12.	Teaching staff	1	2	3	4
13.	Building	1	2	3	4
14.	Equipment	1	2	3	4

15. In general, how satisfied are you with the schools in your district?

Very satisfied	1
Satisfied	2
Dissatisfied	3
Very dissatisfied	4.

16. How interested are you in the future of the schools in your district?

Very interested	1
Interested	2
Disinterested	3
Very disinterested	4.

17.	How much of what your studer value to them?	nts are studying in school is of
	Practically everything	1
	Most	2
	Half	3
	Very little	4
18.	How much do you think your studies?	students are getting from their
	A great deal	1
	Something	2
	A little	3
	Very little	4
19.		ly available for all students to parties, plays, games and clubs
	Always	1
	Most of the time	2
	Occasionally	3
	Never	4
20.		nount of work assigned to your stu- keep up" with their class work?
	Too much work	1
	About right	2
	Not very much work	3
	Too little	4
21.	How much homework do you ass	ign to your students?
	None at all	1
	Little	2
	Appropriate for class work	3
	A great deal	4

22.	you would like to see offered	d?
	All that are necessary Most that are necessary	
	·	
	Few that are necessary	
	None that are necessary	4.
23.	Does the school create enough them to partake in extra-cur	h interest in students to stimulate ricular activities?
	All that is necessary	1
	Most that is necessary	2
	Little that is necessary	3
	None that is necessary	4
24.	Does your school offer a great it offers its students?	at enough variety in the courses
	Wide variety	1
	Enough variety	2
	Little variety	3
	No variety	4
25.	Are there courses which shou are not now being offered to	ld be offered your students that them?
	Many	1
	Some	2
	Few	3
	They now have enough	4
26.	How much help do your studen go to the school library?	ts get from the librarian when they
	All they need	1
	Most of the help they need	2
	Some of the help they need	3
	We have no library	4.
		· · · · · · · · · · · · · · · · · · ·

21.	extra-curricular acti books, gym, laborator	vities and	fees that a	ve to spen re require	d for
	Too much money	1.			
	About right	2.			
	Student's could more		New York Company		
	We don't levy th charges	ese 4.			
28.	How much of what your them after they leave	students a	are studying	will be o	f use to
	Most	1.			
	About half	2.			
	Less than half	3.			
	Very little	4.			
•	Check the activities to a good school curr		es that you i	feel are <u>e</u>	ssential
		Very Es- sential	Essential	Not Es- sential	
29.	Audio-visual aids	1	2	3	4
30.	Physical education	1	2	3	4
31.	Field trips	1.	2	3	4
32.	Music	1	2	3	4
33.	Art	1	2	3	4
34.	Hot lunch program	1	2	3	4
35.	Physical and dental exams	1	2	3	4
36.	Job placement	1	2	3	4
37.	Guidance	1	2	3	4
38	Speech correction	1	2	3	4
	Mental health services	1	2	3	4

1.

2. ____

2. ____ 3. ___

40. Remedial reading

41. Special education

		Very Es- sential	<u>Essential</u>	Not Es- sential	
42.	Adult education	1	2	3	4
43.	Agriculture pro-				
	grams	1	2	3	4
44.	Social activities	1	2	3	4
45.	Driver training	1	2	3	4
46.	Summer school	1	2	3	4
47.	Do you feel that the	schools in	your distric	t are over	crowded?
	Very crowded	1.			
	Crowded				
	Enough room	3.	-		
	Excess space	4.	Virginia de Managaria de Caración de Carac		
48.	Does your school have tory equipment it nee				
	All we need	1.			
	Adequate	2.			
	Could use more	3.	******		
	Lacking	4.			
49.	Do you feel that you	are getting	g enough supe	ervisory as	ssistance?
	Very much	1.	-		
	Adequate	2.	-		
	Little	3.	*****************************		
	None	4.			
50.	Do you feel that ther schools and the commu		d relationshi	p between	the
	Very good	1.	-		
	Good				
	Could be improve	d 3.			
	Poor	4.			

		• •
51.	What is your estimate of t community?	he social status of teachers in your
	Very good	1
	Good	2
	Could be improved	3
	Very low status	4
52.	Do you feel that the schoo about the school and the s	ls inform the community adequately chool program?
	Excellent	1
	Good	2
	Fair	3
	Poor	4
53.	Do you feel that the schoo their meetings in the loca	l board should publish the minutes of l papers?
	Always	1
	Sometimes	2
	Occasionally	3.
	Never	4
54.	To what degree do you feel satisfactory?	parent-teacher relationships are
	Very good	1
	Good	2
	Poor	3
	Very poor	4
55.	To what degree do you feel desirable?	parent-teacher conferences are
	Very desirable	1
	Desirable	2
	Some help	3
	Waste of time	4
56.	To what degree do you feel	PTA's are effective?
	Very effective	1
	Effective	2
	Slightly effective	3

Waste of time

; ;

- .-

•••

•

<i>J</i> / •	the level of services re	endered?
	Very favorably	1
	Favorably	2
	Unfavorably	3
	Way out of line	4
58.	Do you know your student to know them?	s' parents as well as you would like
	Very well	1.
	Well	2
	Little	3
	Not at all	4.
59.	Do you feel that there is parents, administration,	s adequate communication between and school board?
	Excellent	1.
	Good	2
	Fair	3
	Poor	4
60.	Do you feel parents show related activities of the	w much interest in the school work and neir children?
	A great deal	1
	Somewhat	2
	Very little	3
	None	4

APPENDIX B

Elementary Pupils Results by Quartile with Number of Respondents, Total Percentage of Favorable Response, "X2", Degrees of Freedom, and Significance at P > .05

APPENDIX B.--Summary appendix for comparison of elementary pupils by quartile.

Item	Q ₄	Q ₃	92	Q ₁	x ²	d.f.	P > .05
Satisfaction with Schools							
Like School Student Feels Part of Group Teacher Knows Child Teacher Gives Help Value of Studies Proud of School Teacher Interest in Child Schools Compare Favorably Satisfaction with Schools Interest in Schools	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	88688899999999999999999999999999999999	88688899999888 88689999999 7.2808999999	8477888778984447498444749844748884447	8.499 25.209 85.214 15.369 15.369 10.344	ммммммммм	+ + + + + +
School Program							
Effectiveness of Studies Part in Extra-Curricular	9 1	ж •	9 (5	7 - 1	m (ı
>	65. 59.79	63.1 64.3 51.4	66.3 42.4	603 400.0 60.0	0.744 3.049 11.744	๛๛๛	ı ı +
Activities School Creates Interest in	8.94	51.4	56.9	57.4	5.272	m	ı
Extra-curricular Activities Variety of Subjects Subject Wanted Not Taught	52.8 86.5 50.4	56.4 48.8 2.8	63.8 888.8 48.9	60.4 81.4 49.9	5.033 13.875 0.171	mmm	1 + 1

APPENDIX B. -- Continued

Item	Θħ	ه ع	92	4 1	x ²	d.f.	P > .05
Help from Librarian Money Needed for School Usefulness of Subjects	81.1 56.7 98.4	68.5 52.8 95.5	66.3 65.0 97.5	69.4 70.8 95.4	9.433 16.187 3.805	mmm	* + + 1
Essential Services Desired							
Audio-Visual Physical Education Field Trips Music Art Hot Lunch Physical and Dental Examination Speech Correction Mental Health Clinic Remedial Reading Special Education	7.00 7.00 7.00 7.00 7.00 7.00 7.00 7.00	88888 8007 77687 6007 6007 7009 6007	888883 87680 87680 771.70 771.70	188790 076889 1987679 1987679 19878	12.681 15.139 12.886 12.587 75.587 75.587 11.133 6.398 6.398		+ + + +
$\mathbf{A} \mathbf{\Omega}$. 60	96.	9.	828	mm	* +
School Organization and Size School Overcrowded	32.3		24.4	•	3.202	m	ı

APPENDIX B. -- Continued

Item	θη η ₀		92	Q ₁	x ²	d.f.	d.f. P > .05
School Plant							
Adequate Equipment and Facilities	58.1	39.5	58.1 39.5 39.1	25.7	25.7 44.441	κ	+
Community Relations							
Interest of Parents in School Activities	0.96	92.7	96.0 92.7 95.6 94.5		1.761	m	ı

The symbol "+" indicates significance, the symbol "+*" indicates significance, but with low support districts scoring higher than high support districts, and the symbol "_" indicates non-significance at P > .05. $^{1}\mathrm{The}$ number of respondents are: and (4) $\mathbb{Q}_{1}\text{--}366$.

(1) Q_{μ} --125, (2) Q_{3} --112, (3) Q_{2} --275,

APPENDIX C

Secondary Students Results by Quartile with Number of Respondents, Total Percentage of Favorable Response, "X2", Degrees of Freedom, and Significance at P > .05

APPENDIX C.--Summary appendix for comparison of secondary students by quartile.

P > .05		*** ***	* **++++	+ + +
d.f.		๓๓๓๓๓๓๓	๓๓๓๓๓๓	ന നനന ന
x ²		11.210 21.246 30.826 3.898 17.045 20.281	19.915 11.757 3.830 9.694 19.146 22.327	7.057 6.483 3.720 3.720 24.030
Q ₁		772 934.75 93.44 79.46 79.46	865 83.0 83.0 81.4 81.6	85.8 57.8 64.3 86.3 86.8
92		76.7 76.7 91.2 70.2 72.3	61.7 68.0 56.8 74.1	84.8 68.7 54.2 72.8 74.8
4 ₃		65373 6773 6775 6775 6775 6775 6775 6775	400000 400000 4000000 0000000	76.8 64.2 48.2 48.6 72.2 56.9
o ₄		61. 89. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1.	84.8 77.5 76.8 70.5 60.2	79.0 47.4 57.9 87.7
Item	Satisfaction with Schools	art hild elp s t in	compare ravorable talum ng Staff ng nent tion with School tin Schools' Fut	School Program Effectiveness of Studies Part in Extra-Curricular Activities Amount of Work to Keep Up Amount of Homework Enough Extra-Curricular Activities School Creates Interest in Extra-Curricular Activities

APPENDIX C.--Continued

Item	φ t	4 ₃	92	Q ₁	x ²	d.f.	P > .05
Variety of Subjects Subject Wanted Not Taught Help from Librarian Money Needed for School Usefulness of Subjects	78.1 50.4 47.8 90.2	62.4 71.4 48.7 92.0 82.1	50.0 67.2 60.9 94.7 85.2	50.7 69.7 60.9 89.7	32.394 16.170 10.905 5.252 3.417	നനനന	* +++11
Essential Services Desired							
Audio-Visual Physical Education	mmc	2.7	6.	± 0. ⊾	5.91	mmn	* +
Music Art Hot Lunch	71.781.4	888.0.83.0	70.5	76.1	7.179 10.948 21.914	റനനന	* + + +
	⇒ m -	⇒ w 0	⇒0°	40.	5.11 3.83 1.58	mmm	* ! +
Speech Correction Mental Health Clinic Remedial Reading	 	000	٠. د	087	20.8	๛๛๙	1 1 1
Special Education Adult Education			, , ~ ~	. 0 .	 	า๛๛	- +
Agriculture Program Social Activities Summer School Driver Training	43.2 82.1 94.7 93.8	63.1 82.1 96.1	59.8 90.8 94.2	88.5 91.7 98.9	115.116 9.153 25.985 12.998	๛๛๛๛	* * * + + + +

APPENDIX C .-- Continued

Item	Q4	93	92	Q ₁	x ²	d.f.	P > .05
School Organization and Size							
School Overcrowded	78.1	9.69	68.8	40.0	87.800	Ω	+
School Plant							
Adequate Equipment and Facilities	39.5	17.0	25.0	25.3	15.728	က	+
Community Relations							
Interest of Parents in School Activities	4.78	87.5	78.9	84.4	6.680	٣	ı
The number of respondents are: (4) $Q_1 - 380$.	are:	(1) Q ₄ -	113,	(2) 63	(1) Q_{μ} 113, (2) Q_{3} 112, (3) Q_{2} 262, and	9256	2, and

The symbol "+" indicates significance, the symbol "+*" indicates significance, but with low support districts scoring higher than high support districts, and the symbol "-" indicates non-significance at P > .05.

APPENDIX D

Parents Results by Quartile with Number of Respondents, Total Percentage of Favorable Response, "X2", Degrees of Freedom, and Significance at P > .05

APPENDIX D.--Summary appendix for comparison of parents by Quartile.

Item	Q _t	4 ₃	92	Q ₁	x ²	d.f.	P > .05
Satisfaction with Schools							
Like School Student Feels Part of Group Teacher Knows Child Teacher Gives Help Value of Studies Proud of School Teacher Interest in Child Schools Compare Favorably Satisfaction with Schools Interest in Schools' Future	000 000 000 000 000 000 000 000 000 00	100888999999999999999999999999999999999	99988999999999999999999999999999999999	0.088988899 0.08871.60999	11740 443703 11740 443700 10000000000000000000000000000000000	тттттттт	11111111
School Frogram							
Effectiveness of Studies	100.0	92.2	7.76	7.46	10.354	m	+
1 Q (94.3 71.6 54.1	77.7 86.8 71.2	79.8 82.5 63.4	74.2 83.2 65.1	16.836 8.729 6.447	mmm	++1
Enougn Extra-curricular Activities	93.0	80.2	87.0	0.46	20.068	m	*

APPENDIX D. -- Continued

Item	Q ₄	ه ع	8 8	ϕ_1	x ²	d.f.	P > .05
School Creates Interest in Extra-Curricular Activities Variety of Subjects Subject Wanted Not Taught Help from Librarian Money Needed for School Usefulness of Subjects	76.2 888.8 833.7 90.7	955.9 955.9 955.9 955.9	77 88.4 20 57.3 96.7	82.7 80.2 41.5 94.9 94.3	7.471 8.083 10.993 6.392 10.068 3.663	๓๓๓๓๓๓	1++1+1
Essential Services Desired							
Audio-Visual Physical Education Field Trips Music Art Hot Lunch	99999999999999999999999999999999999999	91.2 96.7 91.6 90.7	91.1 94.5 83.0 85.7 87.9	89 89 89 86.6 93.5	0.285 2.166 2.729 7.909 12.837 46.039	๓๓๓๓๓๓	111+++
Examination Job Placement Guidance Speech Correction Mental Health Clinic Remedial Reading Special Education Adult Education Agriculture Program	00000000000000000000000000000000000000	74 689 689 689 689 689 689 689 689 689 689	99 89 90 80 70 70 70 70 70 70 70 70 70 70 70 70 70	9975. 89975. 877. 83.33. 83.33.	15.591 10.623 1.764 3.066 1.167 68.423	ттттттт	* * * +!!!!!+!+

APPENDIX D. -- Continued

Item	Θų	ه ₃	92	Q ₁	x ²	d.f.	P > .05
Social Activities Summer School Driver Training	77.4 82.6 94.6	89.7 79.5 91.8	89.1 77.9 92.1	89.5 75.3 97.3	10.113 2.386 7.557	നനന	* + 1 1
School Organization and Size School Overcrowded	52.3	52.9	39.6	41.6	8.688	m	+
School Plant Adequate Equipment and Facilities	77.3	50.4	54.7	50.2	21.496	m	+
Community Relations							
School-Community Relations Faculty Status in Community School as Informant Minutes Published Parent-Teacher Relations Parent-Teacher Conferences PTA Effectiveness Taxes and Services Rendered Increase Taxes	00000000000000000000000000000000000000	68788888888888888888888888888888888888	82.7 88.7 88.9 788.9 51.6 57.7	68.7 67.7 84.7 55.1 55.1 55.1	7.115 22.011 25.0011 8.045 6.653 1.919 12.756	๓๓๓๓๓๓๓๓๓	1 1 1 + + 1 1 1 +

APPENDIX D. -- Continued

Item	Ф ₄ Ф ₃		92	Q ₁	x ²	d.f.	d.f. P > .05
Know Child's Teacher or Parent	4.49	48.7	41.0	51.0	14.055	3	+
	53.9	30.3	43.9	37.7	13.972	m	+
incerest of Farencs in School Activities	97.8	98.4	8.96	98.3	1.640	М	1
The number of respondents are: (1) \mathbb{Q}_{4} 91, (2) \mathbb{Q}_{3} 117, (3) \mathbb{Q}_{2} 215, and (4) \mathbb{Q}_{1} 300.	are:	(1) Q ₄ -	91, (;	2) Q ₃]	(3) (3)	2215	and

The symbol "+" indicates significance, the symbol "+*" indicates significance, but with low support districts scoring higher than high support districts, and the symbol "-" indicates non-significance at P > .05.

APPENDIX E

Patrons Results by Quartile with Number of Respondents, Total Percentage of Favorable Response, "X2", Degrees of Freedom, and Significance at P > .05

APPENDIX E.--Summary appendix for comparison of patrons by quartile.

Item	φ	ه ₃	92	Q ₁	x ²	d.f.	P > .05
Satisfaction with Schools							
Proud of School Schools Compare Favorably Satisfaction with Schools Interest in Schools' Future	00000 00000 00000	94.0 86.6 80.3 98.5	92.7 84.7 90.9 94.6	92.7 82.8 83.6 95.7	0.224 6.256 5.336 2.796	നനനന	1111
School Program							
Money Needed for School Usefulness of Subjects	63.5 82.8	73.9	80.9	71.5	6.661	mm	1 1
Essential Services							
Audio-Visual Physical Education Field Trips Music	9899 83.9 98.19	84.1 93.9 75.0 94.0	89.0 91.9 70.0 76.4	91.0 89.5 73.7 86.8	2.486 2.163 3.752 21.701	നനനന	111+
Art Hot Lunch	٠.	7.	1.	7.4	.56	mm	* + +
Examination Job Placement	wc	ώωα	20.0	7.4.	.0.5	mmn	1 1 1
Speech Correction Mental Health Clinic Remedial Reading	988.7 97.0 07.0	00000 00000 00000	94.6 77.1 92.3	94.7 83.0 94.3	3.487 6.518 3.398	า๓๓๓	111

APPENDIX E.--Continued

Item	ηδ	ه ₃	92	o ₁	x ²	d.f.	P > .05
Special Education Adult Education Agriculture Program Social Activities Summer School Driver Training	96.8 75.8 40.7 72.6 78.1	98.3 79.4 71.4 88.7.3 89.4	887.44 883.74 893.74 893.74	88888 74888 74688	9.011 5.114 47.008 4.281 8.472	നനനനന	* + * +
Community Relations							
School-Community Relations Faculty Status in Community School as Informant Minutes Published	51.6 74.6 51.6 76.2	41.8 69.2 46.3 78.5	60.0 77.4 45.5 73.0	53.3 51.1 86.0	5.601 5.723 1.319 9.266	ოოოო	111+
Rendered Increase Taxes	81.3	77.6	 8	79.7	0.514 5.915	mm	1 1
Information	33.9	40.3	34.0	40.0	1.747	ж	ı

(1) Q_{μ} --65, (2) Q_{3} --67, (3) Q_{2} --110, and $^{
m l}$ The number of respondents are: (4) Q₁--233. The symbol "+" indicates significance, the symbol "+*" indicates significance, but with low support districts scoring higher than high support districts, and the symbol "-" indicates non-significance at P > .05.

APPENDIX F

Faculty Results by Quartile with Number of Respondents, Total Percentage of Favorable Response, "X2", Degrees of Freedom, and Significance at P > .05

APPENDIX F.--Summary appendix for comparison of faculty by quartile.

Item	9 [†]	43 3	92	9,1	x ²	d.f.	P > .05
Satisfaction with Schools							
Like School Student Feels Part of Group Teacher Knows Child Teacher Gives Help Value of Studies Proud of School Teacher Interest in Child Schools Compare Favorably	87.5 1000.0 1000.0 1000.0	0.0001	93.1 100.0 100.0 100.0 100.0	93.8 93.6 100.0 87.7 93.6	1.582 3.698 0.000 2.946 1.874	๓๓๓๓๓๓๓	111111
ools f ol	000 1 000	0040 886	∞00± 000	10H4 V0V		നനനന നന	1111 + 1
Building Equipment Satisfaction with Schools Interest in Schools' Future	933. 936.73 100.0	60.0 77.8 90.0 100.0	6000 6000 60000	56.7 60.0 77.4 93.6	8.253 3.653 1.8055 4054	๓๓๓๓	+ 1 1 1
o t	100.0	100.0	93.1	90.3	2.512	m m	1 1

APPENDIX F.--Continued

Item	Q ₄	93	92	91	x ²	d.f.	P > .05
	75.0 62.5	100.0	82.8 74.1	74.2 66.7	3.846	mm	1 1
Enougn Extra-curricular Activities School Creates Interest in	87.5	81.8	85.7	71.0	2.759	m	I
Extra-Curricular Activities Variety of Subjects	0 %	0	7.		.12	mm	++
Subject Wanted Not Taught Help from Librarian		45.5	72.4	• •	3.58	า๛๛	1 1
Money Needed for School Usefulness of Subjects		0		77.4	• •	mm	1 1
Essential Services Desired							
Audio-Visual Physical Education	100:0	100.0	0.001	100.0	0.000	mmn	1 1 1
d - 1 - 1	. 00		000	000		n m n	1 1
h S	60.	81.		96.	.89	nm	* +
Injercal and Dencal Examination Job Placement	43. 87.	72.	75. 81.	71. 64.	.36	mm	1 1
Guldance Speech Correction	100.0	100.0	0.001	100.0	0.000	mm	1 1
Mental Health Clinic	87.	00	9.	83.	.24	m	ı

APPENDIX F. -- Continued

Item	Qų	ه ₃	92	۵ ₁	x ²	d.f.	P > .05
Remedial Reading Adult Education Agriculture Program Social Activities Summer School Driver Training	89 99 93.53 88 88	100.0 72.7 45.5 100.0 100.0	100.0 71.4 21.4 96.4 84.0	100.0 87.1 83.8 96.6 71.0	4.426 3.213 31.992 0.742 6.803	നനനനന	11+111
School Organization and Size School Overcrowded Supervisory Assistance	68.8 81.3	72.7	69.0 86.2	83.9 51.6	2.207	mm	I +
School Plant Adequate Equipment and Facilities	50.0	45.5	28.6	38.7	2.287	m	ı
School-Community Relations School-Community Relations Faculty Status in Community School as Informant Minutes Published Parent-Teacher Relations Parent-Teacher Conferences	31.3 75.0 37.5 87.5 100.0	54.6 63.6 90.0 90.0	82.8 72.8 55.2 67.9 93.1	58.1 74.2 45.2 87.1 80.7	12.019 0.527 2.408 4.175 10.259	mmmmm	* + +

APPENDIX F.--Continued

Item	φ	ه ₃	92	Q ₁	x ²	d.f.	P > .05
PTA Effectiveness Taxes and Services Rendered Increase Taxes	56.3 81.3	20.0	25.9 85.7	24.1 76.6	6.347	mm	1 1
Know Child's Teacher or Parent School-Community Information	43.8 43.8	36.4	28.6	30.0	1.264	mm	1 1
School Activities	93.8	81.4	6.79	6.79	4.755	3	ı

(1) Q_{4} --16, (2) Q_{3} --11, (3) Q_{2} --29, and The number of respondents are: (4) $Q_1 - 30$. The symbol "+" indicates significance, the symbol "+*" indicates significance, but with low support districts scoring higher than high support districts, and the symbol "-" indicates non-significance at P > .05.

APPENDIX G

Fourth Quartile Results by Battery with Number of Respondents, Total Percentage of Favorable Response, "X2", Degrees of Freedom, and Significance at P > .05

APPENDIX G.--Summary appendix for comparison of fourth quartile by battery. 1

Item	Elemen- tary Pupil	Secon- dary Student	Parent	Patron	Faculty	x ²	d.f.	P>.05
Satisfaction with Schools	17 00	г гу	8 20	ł	87 5	789 95	~	+
an c	•	• • L	- ^				ר רי	- 4
nows		· .:	· .		000	202	า സ	- +
ives	4.	6	8	!	00	3.60	m	I
Value of Studies	<u>.</u>	۲,	م	-	00	9.56	Υ-	+
Proud of School Teacher Interest in Child	0 0 0 0 0	76.3	95.6	93.9	100.0	29.068	4 %	+ +
Schools Compare Favorably	, ,	 	, ,	5	93.	9.63	14	+
Satisfaction with Schools	9	0	٠.	88.9	'n	6.37	7	+
interest in Schools' Future	94.5	60.2	100.0	98.5	100.0	102.984	7	+
School Program								
Effectiveness of Studies	1.96	0.67	100.0	!	100.0	36.333	3	+
Activities	62.9	71.9	94.3	1	100.0	29.901	3	+
	59.7	4.74	71.6	!	75.0	13.879	\sim	+
Amount of Homework Fronch Extra_Chrrfonlar	9	7	-	!	\sim	83	m	ı
	8.94	87.7	93.0	i i	87.5	77.159	3	+

APPENDIX G. -- Continued

	•		
P>.05	+ + + + +	1 + + + + +	+ 1 1 + + + +
d.f.	mmmm≠ ≠	ਰਹਰਰ ਹਰ	a wwa a a a
x ²	22.146 6.308 7.946 30.185 52.886 32.291	7.121 12.590 16.031 45.011 17.629 15.828	14.548 4.324 3.988 20.349 19.642 15.049
Faculty	100.0 93.8 56.3 68.8 81.3	100.0 93.8 81.3 100.0 100.0	43.8 87.5 100.0 100.0 93.8 93.8
Patron	63.5 82.8	99 98 98 99 95 95 95 95 95 95 95 95 95 95 95 95	673.8 97.0 982.3 96.3
Parent	76.2 88.8 33.7 90.7	99999999999999999999999999999999999999	89999999999999999999999999999999999999
Secon- dary Student	69.3 78.1 47.8 90.2	833.6 83.2 711.7 78.4 81.4	44 733.2 842.2 8748.1 85.1 85.1
Elemen- tary Pupil	850 850 986 987 17-4	92.1 94.4 71.4 87.0	56.9 80.8 65.3 78.1
Item	School Creates Interest in Extra-Curricular Activities Variety of Subjects Subjects Wanted Not Taught Help from Librarian Money Needed for School Usefulness of Subjects	Essential Services Desired Audio-Visual Physical Education Field Trips Music Art Hot Lunch Physical and Dental	Examination Job Placement Guidance Speech Correction Mental Health Clinic Remedial Reading Special Education

APPENDIX G.--Continued

Item	Elemen- tary Pupil	Secon- dary Student	Parent	Patron	Faculty	x ²	d.f.	P>.05
Adult Education Agriculture Program Social Activities Summer School Driver Training	88.0 46.8	75.0 43.2 82.1 94.7 93.8	75.0 37.0 77.4 82.6 91.8	75.8 40.7 72.6 78.1 87.5	87.5 12.5 93.8 93.8	1.265 5.766 9.469 81.690 2.236	mm≠≠m	111+1
School Organization and Size School Overcrowded Supervisory Assistance	32.3	78.1	52.3	11	68.8 81.3	51.734	mo	+ 1
School Plant Adequate Equipment and Facilities	58.1	39.5	77.3	}	50.0	29.284	κ	+
Community Relations								
School-Community Relations	;	;	61.1	51.6	31.3	5.298	N	ı
Faculty Status in Community School as Informant Minutes Published		111	86.4 64.8 75.8	74.6 51.6 76.2	37.5 87.5	3.668 5.498 1.092	N N N	1 1 1

APPENDIX G. -- Continued

Item	Elemen- tary Pupil	Secon- dary Student	Parent	Patron	Parent Patron Faculty	x ²	d.f.	d.f. P>.05
Parent-Teacher Relations	;		97.8		100.0	0.362	7	1
rarent-reacher Conferences PTA Effectiveness		11	84.6 44.0		81.3	0.115	ηп	1 1
Taxes and Services Rendered Increase Taxes	1 1	11	89.0	81.3	81.3	2.054 0.911	7 7	1 1
Anow Unita's Teacher or Parent	!	ł	4.49	! !	43.8	2.419	7	ı
Information	}		53.9	ł	43.8	6.138	7	+
School Activities	0.96	4.78	8.76	1	93.8	10.852	m	+

The number of respondents are: (1) elementary pupils--127, (2) secondary students--114, (3) parents--91, (4) patrons--65, and (5) faculty--16.

The symbol "+" indicates significance and the symbol "-" indicates nonsignificance at P>.05.

APPENDIX H

Third Quartile Results by Battery with Number of Respondents, Total Percentage of Favorable Response, "X2", Degrees of Freedom, and Significance at P > .05

APPENDIX H.--Summary appendix for comparison of third quartile by battery. 1

05															
P.	+	+	· +	+	+	+	+	+	+	+		+	+	+ +	+
d.f.	m	~	n (۲	m	m	4	m	7	4	⇉		\sim	m	mm	3
x ²	18.356	2 07	5.26	13.649	6.04	8.73	0.89	3.04	6.95	46.952		20.185	.11	45.183 16.947	24.049
Faculty	100.0	0	000	100.0	90.	00	0	85.	•	100.0		100.0		100.0	81.8
Patron	1	i	;	ł	!	0.46		9	98.5	98.5		;	1	1 1	1
Parent	92.3	ע	/ 7	83.5	ä		5	5	0	100.0		92.2	7	86.8	80.2
Secon- dary Student	73.2	ĸ	, ~	7.88	5	ά,	1	ω	5.	75.0		8.92	4.	48.5 48.6	72.2
Elemen- tary Pupil	85.7	7	٠ ,	97.3	ä	·	·	·	4.	8.48		93.8	'n	64.3 51.4	51.4
Item	ادا ا	Student Feels Part of Group	Teacher Knows Child	ives	•	Proud of School	Н	Schools Compare Favorably	\vdash	Interest in Schools' Future	School Program	Effectiveness of Studies		Amount of Work to Keep Up Amount of Homework	Enough Extra-Curricular Activities

APPENDIX H.--Continued

P>.05	+++++
d.f.	mmmma
x ²	8.511 19.825 19.825 79.320 13.693 25.742 25.742 26.433 26.433 27.272 27.272 28.472 28.472 28.472 28.472
Faculty	81.8 645.5 1000.0 1000.0 1000.0 1000.0 1000.0 1000.0
Patron	99999999999999999999999999999999999999
Parent	9999997 899999 9999999 768999 999999999999999999999999999999999
Secon- dary Student	99888879 888879 877778 800000 10011000
Elemen- tary Pupil	7768 7 8887888 7 8687.84 9997. 7 1001 7 100.04
Item	School Creates Interest in Extra-Curricular Activities Variety of Subjects Subject Wanted Not Taught Help from Librarian Money Needed for School Usefulness of Subjects Essential Services Desired Audio-Visual Physical Education Field Trips Music Art Hot Lunch Physical and Dental Examination Job Placement Guidance Spech Correction Mental Health Clinic Remedial Reading Special Education

APPENDIX H. -- Continued

Item	Elemen- tary Pupil	Secon- dary Student	Parent	Patron	Faculty	x ²	d.f.	P>.05
Adult Education Agriculture Program Social Activities Summer School Driver Training	86.4 52.8	87.3 63.1 90.2 82.1 96.4	74.1 71.3 89.7 79.5 92.1	79.4 71.4 82.3 84.4 89.4	72.7 45.5 100.0 100.0	6.507 4.619 4.542 38.865 3.520	mm≠≠m	111+1
School Organization and Size School Overcrowded Supervisory Assistance	28.4	9.69	52.9		72.7	40.026	m 0	+ 1
School Plant Adequate Equipment and Facilities	39.5	17.0	50.4	}	45.5	29.412	က	+
Community Relations School-Community Relations	}	!	58.5	41.8	54.6	5.004	α	1
Faculty Status in Community School as Informant Minutes Published			83.1 83.3	69.2 46.3 78.5	63.6 63.6 81.8	5.810 5.301 0.711	~~~	1 1 1

APPENDIX H.--Continued

ı												
d.f. P>.05	i	ı	ı		ı	1		ı		ı		+
d.f.	٦	Н	႕		7	-		_		7		\sim
	0.107	0.211	3.322		1.614	2.039	,	0.618		2.070		12.001
Parent Patron Faculty X ²	0.06	90.0	20.0		70.0		,	36.4		40.0		81.8
Patron	;	!	!		9.77	53.7		!		40.3		!
Parent	86.3	86.0	50.0		83.2			48.7		30.3		98.4
Secon- dary Student	ł	!	;		!	;		!		i		87.5
Elemen- tary Pupil	;	;	!		!	!		!		!		92.7
Item	Parent-Teacher Relations	Parent-Teacher Conferences	PTA Effectiveness	Taxes and Services	Rendered	Increase Taxes	Know Child's Teacher	or Parent	School-Community	Information	Interest of Parents in	School Activities

The number of respondents are: (1) elementary pupils--112, (2) secondary students--112, (3) parents--117, (4) patrons--67, (5) faculty--11.

The symbol "+" indicates significance and the symbol "-" indicates nonsignificance at P > .05

APPENDIX I

Second Quartile Results by Battery with Number of Respondents, Total Percentage of Favorable Response, "X2", Degrees of Freedom, and Significance at P > .05

APPENDIX I.--Summary appendix for comparison of second quartile by battery. 1

Item	Elemen- tary Pupil	Secon- dary Student	Parent	Patron	Faculty	x ²	d.f.	P>.05
Satisfaction with Schools								
1001	88.7	7.97	6.46	!	93.1	36.789	3	+
Group	۲,	Ŋ.	Ψ.	!	0.00	4.38	Υ.	+
Knows	83.8	76.7	90.1	!	100.0	42.747	m	+
Teacher Gives Help	9	Ļ.	7	!	00	5.75	m	+
Studi	ς.	。	寸	!	86.	3.18	m	+
School	ت	5		92.7	0	9.64	7	+
\vdash	ņ.	ς.	9		9	5.90	m	+
Schools Compare Favorably	9	ij	φ	.	φ.	3.77	7	+
001	7	4.	0	6.06	9	4.66	→	+
Interest in Schools' Future	87.3	7.97	99.5	9.46	9.96	67.374	7	+
School Program								
Effectiveness of Studies	2.96	84.8	7.76	;	93.1	39.664	3	+
. U	9	φ.	6	;	8	7.74	m	+
Amount of Work to Keep Up	66.3	56.1	82.5	;	85.8	41.222	Υ	+
Amount of Homework	ک	4.	÷	1	4.	6.64	\sim	+
Activities	6.95	72.8	87.0	i	85.7	56.901	m	+

APPENDIX I.--Continued

Item	Elemen- tary Pupil	Secon- dary Student	Parent	Patron	Faculty	×2	d.f.	P>.05
Ext Ext ivi ivi ty cts fro fro Ine	683.8 488.9 666.3	74.8 50.0 67.2 94.7	77 888.4 29.3 57.3 96.7	80.9	77.8 71.4 72.4 64.3 82.1	14.415 134.484 71.949 4.240 107.892 36.622	MMMM44	+++++
tial Subjects								
Audio-Visual Physical Education Field Trips Music Art Hot Lunch	8883.4 669.0 83.4 83.4	86.2 77.0 70.9 71.8	91.1 883.0 87.7 87.9	89.0 91.9 70.0 76.4 75.0	100.0 96.6 100.0 96.6 65.5	10.425 4.897 17.106 31.595 16.278 25.302	11111	1 1 + + + +
ation ement orrec ealth Read Educa	9. 	877799897 8777999999	99 88 3 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	807±000	75.9 81.5 100.0 96.6 88.7 100.0	40.536 10.489 45.458 45.812 15.665 34.204 29.439	オののオオオオ	++ ++++

APPENDIX I. -- Continued

Item	Elemen- tary Pupil	Secon- dary Student	Parent	Patron	Faculty	s 2	d.f.	P>.05
Adult Education Agriculture Program Social Activities Summer School Driver Training	86.7 59.6	73.7 59.8 90.8 78.2	72.6 67.3 89.1 77.9 93.4	65.4 72.4 83.6 92.8	71.4 21.4 96.4 84.0	2.662 27.447 6.596 31.088	mm==m	1+1++
School Organization School Overcrowded Supervisory Assistance	24.4	68.9	39.6		69 86.2 86.2	115.575	m 0	+ 1
Bchool Plant Adequate Equipment and Facilities	39.1	25.0	54.7	1	28.6	44.939	m	+
Community Relations School-Community Relations Faculty Status in Community School as Informant Minutes Published			71.3 82.7 68.4 62.4	60.0 77.4 45.5 73.0	82.8 72.4 67.9	7.218 2.457 16.277 3.655	N N N N	+ +

APPENDIX I.--Continued

¹The number of respondents are: (1) elementary pupils--275, (2) secondary students--262, (3) parents--215, (4) patrons--110, and (5) faculty--29.

The symbol "+" indicates significance and the symbol "-" indicates nonsignificance at P > .05.

APPENDIX J

First Quartile Results by Battery with Number of Respondents, Total Percentage of Favorable Response, "X2", Degrees of Freedom, and Significance at P > .05

APPENDIX J.--Summary appendix for comparison of first quartile by battery.

Item	Elemen- tary Pupil	Secon- dary Student	Parent	Patron	Faculty	x ²	d.f.	P>.05
Satisfaction with Schools								
Like School	82.0	75.5	95.0	;	93.3	49.391	\sim	+
of Group	7	-	4	!	$\overset{\bullet}{\sim}$	44.	m	+
Teacher Knows Child	82.5	84.7	8.06	1	86.7	9.702	m	+
Teacher Gives Help	Ή.	ж •	$_{\infty}$;	0	7.52	Υ	ı
Value of Studies	$\overset{\cdot}{\sim}$	9	6	1	83.	·84	\sim	+
Proud of School	6	დ	÷	92.7	<u>.</u>	6.32	4	1
n Chil	დ	6	6	i	÷	7.68	Υ	+
Schools Compare Favorably	4.	9	9		ä	7.49	4	+
0]	9	9	6	'n	7	4.34	7	+
Interest in Schools' Future	92.1	81.6	99.3	95.7	93.6	77.070	7	+
School Program								
ctiv	92.1	85.8	7.46	!	90.3	17.391	\sim	+
	ς.	5	4	;	0	4.95	\sim	+
Amount of Work to Keep Up	60.3	58.7	83.2	1	74.2	55.819	Υ	+
of Homework	5	∞	5	!	9	8.07	\sim	+
enougn extra-curricular Activities	4.75	4.49	0.46	!	71.0	117.337	3	+

APPENDIX J. -- Continued

Item	Elemen- tary Pupil	Secon- dary Student	Parent	Patron	Faculty	x 2	d.f.	P>.05
School Creates Interest in Extra-Curricular Activities Variety of Subjects Subject Wanted Not Taught Help from Librarian Money Needed for School Usefulness of Subjects	60.4 49.9 70.8 95.4	66.98 69.7-7-8 69.99-7-1	882 80.7 64.9 94.9	71.5	58.1 48.4 71.0 56.7 93.6	41.469 109.799 62.145 6.855 119.753 27.082	mmmm = =	+++++
Essential Services Desired								
Audio-Visual Physical Education Field Trips Music Art Hot Lunch	91.4 887.8 87.5 75.9 7.9	84.0 77.1 76.1 73.1 86.5	89 883.0 860.10 93.66	91.0 89.5 73.7 86.8 77.1	100.0 100.0 83.3 96.8	16.553 6.312 22.631 75.781 24.418 17.021	aaaaa a	+ + + + +
and thion ment prrecesalth Read	0010000		F-7 0 8 -7 F-12	647.1 647.5 94.7 883.0 88.3		39.801 10.980 68.509 13.228 50.602 46.048	⊐ ∓ π σ σ σ σ σ σ σ σ σ σ σ σ σ σ σ σ σ σ	+ + + + + +

APPENDIX J. -- Continued

Item	Elemen- tary Pupil	Secon- dary Student	Parent	Patron	Faculty	X 2	d.f.	P>.05
Adult Education Agriculture Program Social Activities Summer School Driver Training	91.7	77.7 88.5 91.7 72.7 98.9	70.1 83.3 75.3	888.0 83.6 83.6 94.4	87.1 83.9 96.6 71.0	11.297 4.140 14.130 22.693 20.096	mm=== m	+ + + +
School Organization School Overcrowded	29.3	40.0	41.6	1	83.9	41.613	\sim	+
School Plant Adequate Equipment and Facilities Community Relations	25.7	25.3	50.2	}	38.7	59.567	m	+
Communit ions	1	!	68.5	53.3	58.1	13.143	7	+
raculty Status in Community School as Informant Minutes Published			78.0 62.5 79.2	65.4 51.1 86.0	74.2 45.2 87.1	10.470 8.817 4.595	~~~	+ + 1

APPENDIX J.--Continued

d.f. P>.05	+	ı	+		ı			+		ı		+
д Н	Н	٦	٦		7			Ч		7		\sim
x ₂	740.6	0.285	5.645		4.652	0.828		4.813		4.521		65.613
Patron Faculty	70.0	80.7	24.1		7.97	!		30.0		20.0		6.79
Patron	1	;			7.67	51.1		!		40.0		1
Parent	89.1	76.4	47.1		86.1	55.1		51.0		37.7		98.3
Secon- dary Student	}	1	i		!			!		!		84.4
Elemen- tary Pupil	į	;	!		!	 		1		1		94.5
Item	Parent-Teacher Relations	conferences	PTA Effectiveness	Taxes and Services	Rendered	Increase Taxes	Know Child's Teacher	or Parent	School-Community	Information	Interest of Parents in	School Activities

 $^{1}\mathrm{The}$ number of respondents are: (1) elementary pupils--366, (2) secondary students--380, (3) parents--300, (4) patrons--233, and (5) faculty--30.

The symbol "+" indicates significance and the symbol "-" indicates nonsignificance at P $^{>}$.05.

